Transversals, indivisible plexes and partitions of latin squares

Judith Egan

A thesis submitted for the degree of Doctor of Philosophy,
Monash University,
March 2010.

Under the Copyright Act 1968, this thesis must be used only under the normal conditions of scholarly fair dealing. In particular no results or conclusions should be extracted from it, nor should it be copied or closely paraphrased in whole or in part without the written consent of the author. Proper written acknowledgement should be made for any assistance obtained from this thesis.

I certify that I have made all reasonable efforts to secure copyright permissions for third-party content included in this thesis and have not knowingly added copyright content to my work without the owner's permission.

Declaration

I declare that this thesis is my own work and that this thesis contains no material which has been accepted for the award of any other degree or diploma in any university or other institution. To the best of my knowledge this thesis contains no material previously published or written by another person, except where due reference is made in the text of the thesis.

Judith Egan 29 March 2010

Acknowledgements

I thank my supervisor, Dr. Ian Wanless. Ian's encouragement, expertise and steady guidance has been fundamental to this thesis. Ian has been extremely generous with the amount of time that he has dedicated to my supervision, and also in the funding of my regular visits to Melbourne. Secondly, I thank my associate supervisor, Assoc. Prof. Graham Farr, for his engaging topics and tuition, and an ongoing interest in my progress and development. My regular participation at research meetings via the internet is due to Graham's initiative. I thank my third supervisor Prof. Darryn Bryant and Dr. Barbara Maenhaut for their interest and contribution to the results in Section 4.2.2. I also thank Prof. Brendan McKay for his generous invitation to visit the Australian National University and the resultant inclusion in the results in Section 4.5.3.

I thank the following people for their interest, comments, and helpful suggestions on an aspect of this thesis or a related manuscript: Nicolas Cavenagh, Ian Roberts, Daniel Delbourgo, Jeanette McLeod, Kyle Pula, Paulette Lieby, Kerri Morgan and Michael Egan. More broadly, thank you to the many members of the mathematics community, whom I have met at conferences or elsewhere, and who have given comment about my work.

This research was supported by a Monash Graduate Scholarship. My participation at conferences, workshops and courses in Australia, New Zealand and the United Kingdom was made possible by generous funding assistance from an Australian Research Council Grant, Monash University's School of Mathematical Sciences, Monash Graduate School, Australian Mathematics Society, and the Australian Mathematical Sciences Institute.

The computational results in this thesis were made possible by high performance facilities of: VPAC, Monash Sun Grid and Monash Green SPONGE.

My thanks to Kerri and David Morgan and family, for great company, travel assistance, and a home during my visits to Melbourne.

Parents John and Thelma Britton, Dale and Ellen Egan, my friends and extended family have all been thoroughly supportive. Thanks!

My immediate family and theirs: Alan, Kerry, Sean, Caitlin, Patrick, Daniel and Vanessa, Leila and Chad; my husband Michael. Thank you all for your encouragement and involvement in this project.

Abstract

A k-plex in a latin square of order n is a selection of kn entries in which each row, column and symbol occurs exactly k times. A 1-plex is also called a transversal. A k-plex is indivisible if it contains no c-plex for 0 < c < k. Two or more plexes are parallel if no two of them share an entry. This thesis is the first major study giving general results on indivisible k-plexes for k > 1.

For $n \notin \{2,6\}$, the existence of a latin square of order n with a partition into 1-plexes was shown by Bose, Shrikhande and Parker. A natural extension of the result to k-plexes is the following. We prove that if k is a proper divisor of n, then there exists a latin square of order n composed of parallel indivisible k-plexes.

Define $\kappa(n)$ to be the largest integer k such that some latin square of order n contains an indivisible k-plex. A conjecture by Rodney says that every latin square contains a 2-plex, which implies that $\kappa(n) < n$. We show that for all n > 2, there exists a latin square of order n with two parallel indivisible $\lfloor \frac{n}{2} \rfloor$ -plexes. This proves that $\kappa(n) \geqslant \lfloor \frac{n}{2} \rfloor$ for all n > 2. We report on extensive computations of the indivisible partitions of the latin squares of order $n \leqslant 9$. Up to order 8 we count the number of indivisible partitions of every type.

Due to Bose, Shrikhande and Parker, and Finney for n=6, we know that there exists a latin square for each order n>2 which possesses a k-plex for all $0\leqslant k\leqslant n$. Wanless showed that there are latin squares of every even order which contain no odd plexes, where an odd plex is a k-plex such that k is odd. It has been conjectured by Rodney, and by Wanless, that every latin square has a set of $\lfloor \frac{n}{2} \rfloor$ parallel 2-plexes which implies that every latin square of odd order has a k-plex for each $0\leqslant k\leqslant n$. We show that among the latin squares of even order there are many other possibilities concerning the existence of odd k-plexes. We prove that for all even n>2, there exists a latin square of order n which has no k-plex for any odd $k<\lfloor \frac{n}{4}\rfloor$, but does have a k-plex for every other $k\leqslant \frac{1}{2}n$.

A result by Wanless and Webb is that, for all n > 3, there exists a latin square of order n with at least one entry not in any transversal. Such a latin square is called a confirmed bachelor and illustrates a restriction on the transversals in a latin square. In our study of transversals we consider confirmed bachelor latin squares in greater detail and other types of restrictions that might occur. We present a concise

alternative proof of the result by Wanless and Webb. A main result is that for all even $n \ge 10$, except perhaps if n is a power of 2, there exists a latin square of order n that possesses a transversal, but every transversal coincides on a single entry. A theorem by Wanless, which was prompted by our data, shows that there exist arbitrarily large latin squares of odd order in which the proportion of entries not in a transversal is asymptotic to one ninth. We report on computations of parallel transversals in the latin squares of order 9. Thus we prove that the above mentioned conjectures by Rodney and Wanless are true for order 9, and that every latin square of order 9 has at least 3 parallel transversals. Our computations suggest that the constructions which prove our theorems illustrate rare behaviour in large latin squares. The computations also give further evidence in support of a conjecture by Ryser that every latin square of odd order has a transversal.

Many of the theorems in this thesis rely on a simple but powerful lemma. The lemma gives a necessary condition for the existence of k-plexes. To show the existence of k-plexes we use constructive techniques. It remains open to establish a sufficient condition for the existence of k-plexes in latin squares.

Keywords: latin square, transversal, plex, duplex, triplex, quadruplex, indivisible plex, indivisible partition, orthogonal partition, bachelor square, mutually orthogonal latin squares, partial latin square, protoplex, homogeneous partial latin square, latin trade, homogeneous latin trade, rainbow factor, graph factorisation, quasigroup, complete mapping, orthomorphism.

AMS subject classification (2010): Primary 05B15. Secondary 62K99, 05C70.

Contents

| 1 | Prel | iminary | 1 |
|---------|------|----------------------------------------------------|----|
| | 1.1 | Introduction | 1 |
| | 1.2 | Equivalence of latin squares | 3 |
| | 1.3 | Plexes, indivisible plexes and partitions | 4 |
| | 1.4 | Similar concepts in designs and graphs | 6 |
| | 1.5 | Partial transversals | 7 |
| | 1.6 | Sets of permutations | 8 |
| | 1.7 | Orthogonal latin squares and k -partitions | 10 |
| | 1.8 | Bachelor latin squares | 12 |
| | 1.9 | Step type latin squares | 14 |
| | 1.10 | Quasigroups, loops and groups | 14 |
| | 1.11 | Diagonally cyclic latin squares | 17 |
| | 1.12 | Partial latin squares and completability | 18 |
| | 1.13 | Latin trades and homogeneous partial latin squares | 22 |
| | 1.14 | The number of plexes and partitions | 28 |
| | | 1.14.1 Small order studies | 28 |
| | | 1.14.2 General cases | 30 |
| ${f 2}$ | A ke | ey lemma, definitions and latin families | 33 |
| | 2.1 | Introduction | 33 |
| | 2.2 | A necessary condition for plexes | 33 |
| | 2.3 | Notation | 34 |
| | | | |

| viii <i>Contents</i> | |
|----------------------|--|
|----------------------|--|

| | 2.4 | Latin families of even order | 35 |
|---|------|------------------------------------------------------------------------------------------------|----|
| | | 2.4.1 \mathcal{P}_n , \mathcal{Q}_n and \mathcal{R}_n | 35 |
| | | 2.4.2 $\mathcal{V}_{n,j}$ | 37 |
| | | 2.4.3 \mathcal{U}_n | 38 |
| | | 2.4.4 \mathcal{A}_n and \mathcal{B}_n | 39 |
| | 2.5 | Latin families of odd order | 40 |
| | | $2.5.1 \mathcal{H}_n \dots \dots \dots \dots \dots \dots \dots \dots$ | 40 |
| | | $2.5.2 \mathcal{W}_n \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$ | 41 |
| | | $2.5.3 \mathcal{O}_{m,k} \dots \dots \dots \dots \dots \dots \dots \dots \dots $ | 42 |
| | | $2.5.4 \mathcal{D}_{3m} \dots \dots \dots \dots \dots \dots \dots \dots$ | 43 |
| 3 | Lati | in squares with no small odd plexes | 45 |
| | 3.1 | Introduction | 45 |
| | 3.2 | Necessary conditions | 45 |
| | 3.3 | Partitions of \mathcal{P}_n | 47 |
| | 3.4 | Partitions of Q_n | 52 |
| | 3.5 | Concluding remarks | 54 |
| 4 | Indi | ivisible plexes and partitions | 57 |
| | 4.1 | Introduction | 57 |
| | 4.2 | Latin squares of even order | 57 |
| | | 4.2.1 Indivisible partitions of \mathcal{P}_{2h} , \mathcal{Q}_{2h} and \mathcal{R}_{2h} | 58 |
| | | 4.2.2 Embedding plexes into larger latin squares | 64 |
| | 4.3 | Latin squares of odd order | 67 |
| | | 4.3.1 Indivisible partitions of $\mathcal{O}_{m,k}$ | 67 |
| | | 4.3.2 Indivisible partitions of \mathcal{H}_n | 68 |
| | | 4.3.3 Indivisible partitions of \mathbb{Z}_n | 70 |
| | 4.4 | Subsquares and restrictions on odd plexes | 72 |
| | 4.5 | Latin squares of small order | 77 |
| | | 4.5.1 Definitions and explanation of tables | 77 |
| | | | |

| Contents | | | | | |
|--------------|-----------|-----------------------------------------------------|-----|--|--|
| | | 4.5.2 Species of order less than 9 | 78 | | |
| | | 4.5.3 Species of order 9 | 88 | | |
| | 4.6 | Concluding remarks | 92 | | |
| 5 | Lat | in squares with restricted transversals | 95 | | |
| | 5.1 | Introduction | 95 | | |
| | 5.2 | Transversals in \mathcal{D}_{3m} | 97 | | |
| | 5.3 | Transversals in \mathcal{U}_n and \mathcal{B}_n | 99 | | |
| | 5.4 | Latin squares of small order | 103 | | |
| | 5.5 | Concluding remarks | 107 | | |
| \mathbf{A} | Afterword | | | | |
| \mathbf{R} | efere | nces 1 | 11 | | |

x Contents

List of Tables

| 3.1 | Species of order $n \leq 9$ according to plex range | 54 |
|------|----------------------------------------------------------------------------------|-----|
| 4.1 | Plexes and indivisible partitions in species of order $n=4$ | 79 |
| 4.2 | Plexes and indivisible partitions in species of order $n = 5$ | 79 |
| 4.3 | Categories of order $n = 6$ | 79 |
| 4.4 | Plexes and indivisible partitions in species of order $n = 6$ | 80 |
| 4.5 | Plexes and indivisible partitions in groups of order $n = 6$ | 80 |
| 4.6 | Categories of order $n = 7$ | 81 |
| 4.7 | Plexes and indivisible partitions in species of order $n = 7$ | 82 |
| 4.8 | Plexes and indivisible partitions in group of order $n = 7$ | 83 |
| 4.9 | Categories of order $n=8$ for species possessing an indivisible 5-plex | 84 |
| 4.10 | Categories of order $n=8$ for species with no indivisible 5-plex | 85 |
| 4.11 | Plexes and indivisible partitions in species of order $n=8.\ldots$ | 86 |
| 4.12 | Plexes and indivisible partitions in groups of order $n = 8$ | 88 |
| 4.13 | Plexes and indivisible partitions in six species of order $n=9.\ldots$ | 91 |
| 4.14 | $\kappa(n)$ for $n \leqslant 11$ | 92 |
| 5.1 | Species of order $4 \le n \le 9$ according to number of transversal-free entries | 106 |
| 5.2 | Species of order $n \leq 9$ according to λ | 106 |

Chapter 1

Preliminary

1.1 Introduction

A latin square of order n is an $n \times n$ matrix in which n distinct symbols are arranged so that each symbol occurs once in each row, and once in each column. The unbordered Cayley table of a finite group is an example of a latin square. The very popular Sudoku puzzle found in contemporary media, when completed, is an example of a latin square of order 9.

A transversal in a latin square of order n is a selection of n entries in which each row, column and symbol occur once.

A k-plex in a latin square is a selection of entries in which each row, column and symbol occurs exactly k times. A 1-plex is usually called a transversal. In statements of a general nature we may simply refer to a plex or the plural form 'plexes'. We call a k-plex an odd plex or an even plex if k is odd or even, respectively. Various names for specific cases of k-plexes have been used. Statistical literature sometimes refers to a transversal as a directrix and uses the terms duplex, triplex and quadruplex for a 2-plex, 3-plex and 4-plex respectively. Some other names, k-stagger [20] and k-transversal [39], are discussed in [150] where the name k-plex is first adopted in the current context. The name k-transversal is used with a different meaning in [46, p453] and [48, p8]. A k-plex is mentioned, but not named, as one of three generalisations of transversals which are described in [48, p33].

This thesis is about the existence and non-existence of plexes within latin squares and their substructures. The major part of the author's original contribution lies in the following results:

1. Theorem 3.2 showing existence of latin squares with k-plexes for some but not all odd values of k.

- 2. Theorem 4.1 showing existence of latin squares composed of disjoint plexes that contain no smaller plex.
- 3. Theorem 4.2 showing latin squares with relatively large plexes that contain no smaller plex.
- 4. Theorem 5.2 showing latin squares with transversals coinciding on a single entry.
- 5. Extensive computations of plexes and partitions of latin squares of order less than or equal to 9. See Sections 4.5 and 5.4.

Interest in transversals in latin squares began around 1782 with the work of Euler [71]. In the 1930's, the statistical work of Fisher and Yates (e.g. [80]) promoted the value of latin squares in the design and analysis of scientific experiments, stimulating additional interest in latin squares and transversals. Work in the area of group theory and quasigroup theory has progressed our understanding of transversals through study of their algebraic equivalent. The very recent resolution of a longstanding conjecture by Hall and Paige [95] brings news about plexes in latin squares that are based on groups. Independent of the original motivation, transversals have been studied for the greater part of the last century. However, a number of very basic questions about them remain a challenge to mathematicians.

One important unresolved question concerns the next conjecture. The conjecture is attributed to Ryser, 1967 [39, p143].

Conjecture 1.1. Every latin square of odd order has a transversal.

The study of k-plexes for k > 1 spans a much shorter time period than the case k = 1. Originally motivated by statistical application, studies of k-plexes for k = 2, 3, 4 in small latin squares appear in the literature from 1945 [76]. One of the earliest general statements about k-plexes, for k > 1, is the next conjecture which is attributed to Rodney, 1994 [39, p143],[55], [132].

Conjecture 1.2. Every latin square contains a 2-plex.

Several general results for k-plexes for k > 1 appear in [150]. Some earlier results about k-plexes originated from the study of partial latin squares and their possible completion to a latin square [20, 45]. Extending on Wanless's work in [150], the author's recent results in [65] motivate the particular direction and approach of this research. Indeed we continue from that point to consolidate and develop results obtained earlier, and explore questions arising. Several chapters of this thesis incorporate, as appropriately indicated, work submitted in [65] because it is necessary for the exposition of the current results.

The remaining sections of this introductory chapter introduce essential definitions, other motivating problems, and review literature relevant to this study. We report on

various different approaches that have been used to tackle Conjectures 1.1 and 1.2. For further explanation of latin square terminology and concepts we refer the reader to standard reference texts by Dénes and Keedwell [46, 48]. The small amount of graph theory and group theory terminology that we use accords with the standard usage. Any other assumed terminology is likely to be found in the Handbook of Combinatorial Designs [39].

In Chapter 2 we introduce a key lemma along with the infinite families of latin squares and notation used in our work in Chapters 3, 4 and 5. The content of these three main chapters is outlined during this first chapter as we review related work. Further research problems and concluding remarks can be found at the end of Chapters 3, 4 and 5.

1.2 Equivalence of latin squares

The following equivalence relations are of fundamental importance in the theory of latin squares [46].

If L and L' are latin squares of order n and it is possible to transform L to L' by permuting the rows, permuting the columns and permuting symbols, then L' is said to be *isotopic* to L'. The ordered triple of permutations which transforms L to L' is an *isotopy* or *isotopism*. The set of all latin squares isotopic to L form the *isotopy* class of L. An autotopy of L is an isotopy which preserves L. The set of autotopies of L forms the autotopy group of L. The cardinality of an isotopy class of order n is $n!^3$ divided by the order of its autotopy group.

Another important equivalence relation is *conjugacy*. If L' can be obtained from L by permuting the roles of the rows, columns and symbols, then L' is said to be a *conjugate* of L. Since permutation of the three roles generates a group isomorphic to S_3 , the symmetric group of order 3, the number of distinct conjugates of a latin square is always, 1, 2, 3 or 6, the index of a subgroup of S_3 .

Two latin squares L and L' are $main\ class\ isotopic$, or paratopic, if L is isotopic to any conjugate of L'. A map consisting of an isotopy and a permutation which transforms a latin square to one of its conjugates is a paratopy, or paratopism.

The set of all conjugates of the latin squares in an isotopy class is known by several names; main class, paratopy class or species. We use the name species. The number of transversals in a latin square is well known to be a species invariant and it follows from similar reasoning that so is the number of k-plexes in a latin square. The action of the permutations, whether an isotopy or a paratopy, on a latin square L can be thought of as a relabelling of the rows, columns and symbols of L [46, p7]. Thus, such permutations cannot alter the existence of a transversal, a k-plex, or any other similar substructure of a latin square.

A latin square that is equal to its transpose is called *symmetric*. A latin square is *semi-symmetric* if three of its conjugates are equal.

1.3 Plexes, indivisible plexes and partitions

The classical notion of a latin square is a matrix. Equivalently, we may consider a latin square as a set of ordered triples. Set terminology and notation is an advantage in our later workings.

A set L of n^2 ordered triples $(x, y, z) \in \mathcal{I}(L)^3$, where $\mathcal{I}(L)$ is a set of cardinality n, and no two distinct elements of L agree in more than one coordinate, defines a latin square of order n. The set $\mathcal{I}(L)$ is the *index set* of L and we say that L is *indexed* by $\mathcal{I}(L)$. If $(x, y, z) \in L$, then the corresponding matrix has the symbol, or *entry*, z at the intersection of row x and column y. Typically, we will set $\mathcal{I}(L) = \{0, 1, \ldots, n-1\}$ to facilitate calculations in the cyclic group $(\mathbb{Z}_n, +)$, denoted by \mathbb{Z}_n . A subset of a latin square which itself is a latin square is called a *subsquare*. A subsquare of order 2 is called an *intercalate*.

In set terminology, a k-plex is a subset of a latin square consisting of kn entries in which each row, column and symbol occurs k times. Our definition of a k-plex differs from the first usage in [150] as we insist on containment in some latin square of order n. This is further explained in Section 1.12 where we give an alternative definition of a k-plex as a partial latin square.

We call a set of plexes parallel, or disjoint, if no two of them share a common element. The union of an a-plex and a parallel b-plex in a latin square L yields an (a + b)-plex of L. The reverse process, that is dividing an (a + b)-plex into an a-plex parallel to a b-plex, is not necessarily possible. If a k-plex contains no c-plex for 0 < c < k then it is said to be indivisible.

If K is a k-plex in L, it follows that the *complement*, that is $L \setminus K$, is an (n-k)-plex. Together, these plexes form a (k, n-k)-partition. More generally, a $(k_1^{\alpha_1}, k_2^{\alpha_2}, \ldots)$ -partition is a partition with α_1 parts which are k_1 -plexes, α_2 parts which are k_2 -plexes, etc. The tuple $(k_1^{\alpha_1}, k_2^{\alpha_2}, \ldots)$ is called the *type* of a partition. It is of particular interest when all parts are of the same size. We call a $(k^{n/k})$ -partition a k-partition. If each plex in a partition is indivisible, then we say that the partition itself is indivisible. Under set union of two or more parts, the indivisible partitions of a latin square produce all other partitions admitting divisible plexes. Divisible plexes have also been called degenerate plexes [78, 84].

Conjecture 1.3, stated next, is stronger than Conjectures 1.1 and 1.2. Dougherty [55] states Conjecture 1.3 as a revision, apparently by Rodney, of the earlier Conjecture 1.2. Independently, Wanless states Conjecture 1.3 in [150].

Conjecture 1.3. Every latin square of order n contains $\lfloor n/2 \rfloor$ parallel 2-plexes.

Wanless [150] proved, by computation, that Conjecture 1.3 is true for $n \leq 8$. In Chapter 5 we report on computations proving that Conjecture 1.3 is true for n = 9.

Conjecture 1.3 implies, by union of parts, that every latin square of odd order possesses a k-plex for each $k \in \{0, 1, ..., n\}$, and that no latin square will fail to contain a k-plex for any even value of k in that range.

The *plex range*, of a latin square L, is the set of values of k for which a k-plex is contained in L. Let R be the plex range of a latin square L of order n. Then L is said to have a *complete* plex range if $R = \{0, 1, ..., n\}$, an *even* plex range if $R = \{0, 2, 4, ..., n\}$, and a *mixed* plex range if R is neither complete nor even. (Only squares of even order can have an even plex range.)

Existence of a 1-partition implies a complete plex range but the converse does not hold. For example, the shading in (1.1) identifies a $(1^2, 2^2)$ -partition so, by an appropriate union of plexes from the partition, this latin square has a complete plex range. However it does not have a 1-partition.

$$\begin{pmatrix}
5 & 1 & 2 & 3 & 4 & 0 \\
1 & 0 & 3 & 2 & 5 & 4 \\
2 & 3 & 4 & 5 & 0 & 1 \\
3 & 4 & 5 & 0 & 1 & 2 \\
4 & 5 & 0 & 1 & 2 & 3 \\
0 & 2 & 1 & 4 & 3 & 5
\end{pmatrix}$$
(1.1)

Among the small latin squares of even order, Finney [78, 79] for order 6 and Wanless [150] for order 8, report the existence of latin squares with a 3-plex but no transversal, hence latin squares with a mixed plex range. The following conjecture by Wanless [150] remains open.

Conjecture 1.4. For all even n > 4, there exists a latin square of order n with a 3-plex but no transversal.

In Chapter 3 we show that, for all even n > 2, there exists a latin square of order n that has no k-plex for any odd $k < \lfloor \frac{n}{4} \rfloor$, but does have a k-plex for every other $k \leq \frac{1}{2}n$. A corollary is the existence, in the general case, of latin squares which possess a mixed plex range. Latin squares with a complete or even plex range are shown in the next sections by Corollaries 1.17 and 1.28.

Conjecture 1.3, along with evidence from small order studies, including our own in Chapters 4 and 5, suggests that all latin squares are 'highly divisible' in the sense that they possess many different partitions.

Problem 1.5. For any given n, define $\kappa(n)$ to be the largest k such that there exists a latin square of order n containing an indivisible k-plex. What is the asymptotic behaviour of $\kappa(n)$?

A major goal of this thesis is to better understand $\kappa(n)$.

Conjectures 1.2 and 1.3 imply that, for all n > 2, $\kappa(n) < n$. A result in [65], and restated in this thesis as Lemma 4.8, says that, for all odd k > 1, there exists a latin square of order 2k with an indivisible k-partition. In Chapter 4 we will show that $\kappa(n) \geqslant \lfloor \frac{n}{2} \rfloor$ by showing two parallel $\lfloor \frac{n}{2} \rfloor$ -plexes in a latin square for each order n > 2. In Section 4.5, we establish $\kappa(n)$ exactly for $n \leqslant 8$ and show that $\kappa(9)$ is either 6 or 7.

As noted in [154], the existence of proportionally large indivisible plexes, such as those mentioned above, may be surprising. In 1996, Dougherty [55] reports the next conjecture which was made in response to Conjecture 1.2.

Conjecture 1.6. If k > 2, then any k-plex in a latin square can be partitioned into an a-plex and a b-plex for some a and b with a + b = k.

Dougherty was clearly unaware that indivisible 3-plexes in latin squares of order 5 and 6 had been identified by Finney in [77, 78]. Also, Drisko is attributed with a counterexample of order 5 at Dougherty's homepage [54]. Beyond our results about $\kappa(n)$, the data on indivisible partitions in Section 4.5 suggests a prevalence of indivisible plexes in latin squares which may also be surprising.

1.4 Similar concepts in designs and graphs

The problems that we consider are related to a number of problems that have been studied in a variety of combinatorial contexts. Some examples connected with combinatorial designs and graphs are mentioned next. We do not present a thorough coverage. Examples in some other contexts will be mentioned in later sections.

Together a k-plex and its complementary (n-k)-plex are examples of an orthogonal partition of a latin square. Much early literature about k-plexes in small latin squares is expressed using this terminology [76–78]. A precise definition is not important to us but we caution that orthogonality of partitions has been defined in different ways [7, 127]. Another example of an orthogonal partition of a latin square is the obvious partition by its rows, columns or symbols (e.g.[7]). Much interest in orthogonal partitions, especially those with a regular part size which we discuss in Section 1.7, stems from their extensive application in the design and analysis of experiments [48, Ch.10]. Bailey [7] and Gilliland [88] both consider orthogonal partitions in a general setting, of which latin squares are a particular example.

Whenever we refer to a partition of a latin square it can be assumed that we mean a partition into parallel plexes. Finding a partition of a latin square L is equivalent to finding a frequency square orthogonal to L, where frequency square and orthogonality are as defined in [39, p465-466]. The non-existence of a latin square of order 6 with a

1-partition motivated the first studies of 2-plexes by Finney [76–78]. Various further designs involving latin squares of order 6 are discussed in [82–86, 93].

Any latin square yields a transversal design, by the method described in [39, p161]. A k-plex in a latin square corresponds to a k-parallel class of the transversal design. Non-separable k-parallel classes in Steiner triple systems have been investigated [40, p418]. An indivisible k-plex in a latin square corresponds to a non-separable k-parallel class in the associated design. Abel et al. [1] use a 2-plex, of a finite group with no transversal, to construct a class of generalised Bhaskar Rao designs.

In graph theory a latin square of order n may be described in several different ways. For example, a proper edge colouring of the complete bipartite graph $K_{n,n}$ with n colours. In this context each colour represents a symbol in the latin square. A transversal is represented in the graph by a 1-factor (or perfect matching) in which each edge has a distinct colour. A matching whose edges have distinct colours is called a rainbow [158], a rainbow matching [2], an orthogonal matching [141] or multicolored matching [3]. A k-plex is a k-factor of $K_{n,n}$ in which each of the n colours occurs on k edges and a k-partition is a k-factorisation into factors of this type.

Graph factorisation problems have been well studied [39, 126, 148]. One of the earliest results of graph theory, by Petersen in 1891, says that every regular graph of even degree has a 2-factorisation [52, p39]. The corresponding result for k-plexes in latin squares remains open and is the subject of Conjecture 1.3.

The term k-plex in graph theory may take a different meaning to our usage. In the study of graphs of social networks the name k-plex has been used to described a generalised clique [135]. Prior usage of the name k-plex for that purpose was not known in 2002 when it was first adopted in the context relevant to this thesis.

1.5 Partial transversals

Another generalisation of a transversal in a latin square is the following.

A partial transversal of size or length m is a set of m elements of a latin square such that no row, column or symbol occurs more than once.

The following conjecture by Brualdi [46, p103] also remains open after many decades.

Conjecture 1.7. Every latin square of order n has a partial transversal of size n-1.

In 1975, Stein [139] conjectured that the same is true for a more general structure which he calls an *equi-n-square*.

Conjecture 1.8. Every $n \times n$ matrix in which each of n symbols occurs n times has a partial transversal of size n-1.

Erdős and Spencer [70] proved that an $n \times n$ matrix in which no symbol occurs more than (n-1)/16 times has a transversal.

There have been errors in several published results concerning Conjecture 1.7. A fatal flaw in a published proof of Conjecture 1.7, by Derienko [51], is shown by Cameron and Wanless [23]. A recent article by Hatami and Shor [98] explains a "bug" in [136] which was subsequently duplicated in [87]. Hatami and Shor [98] conclude that the following theorem, with a corrected higher constant, is proven and that their method cannot achieve a bound better than $n - O(\log^2 n)$.

Theorem 1.9. Every latin square of order n has a partial transversal of length at least $n - 11.053 \log^2 n$.

The theorem improves on earlier bounds (2n+3)/3 by Koksma [107], 3n/4 by Drake [56] and then $n-\sqrt{n}$ by Woolbright [157] and independently by Brouwer, de Vries and Wieringa [17].

Zaker [159] uses a graph setting to show that the computational complexity of finding a partial transversal of maximum size in a 2-plex is an NP-complete problem.

Cameron and Wanless [23] show that every latin square contains a set of n entries such that every row and column occurs once and no symbol occurs more than twice.

Akabari and Alipour [3] showed that the number of partial transversals of length n-1 is even.

Aharoni and Berger [2] further generalise Conjectures 1.1, 1.7 and 1.8 in the context of hypergraphs.

With some consideration as to an appropriate definition, one might also ask questions about partial k-plexes. A requirement that no row, column or symbol occurs more than k times does not imply, for k > 1, that such a set of elements has cardinality divisible by k. An idea of length m corresponding to k entries in each of m rows (or columns) such that no row, column or symbol occurs more than k times is perhaps the clearest analogy. There is considerable interest in the case where every row, column or symbol contains either 0 or k entries, and the set also satisfies the definition of a latin trade. We will discuss such latin trades in Section 1.13.

1.6 Sets of permutations

Let S_n denote the set of all distinct permutations of $[n] = \{1, 2, ..., n\}$. Let $\bar{x} = x_1x_2...x_n$ and $\bar{y} = y_1y_2...y_n$ be distinct permutations in S_n . Then \bar{x} and \bar{y} are said to agree in position $i \in [n]$ if $x_i = y_i$. The rows of any latin square identify a sharply transitive set of permutations. That is, the rows in L, considered as a set of permutations, form a subset of S_n , and for any $i, j \in [n]$, exactly one permutation in

L takes i to position j. Every pair of rows of a latin square agree in zero positions, and a set of rows of a latin square is maximal in size with regard to this property.

An alternative approach to Conjectures 1.1 and 1.7 emerged in an unpublished manuscript by Kézdy and Snevily [105]. Results of [105] are recorded in [23] and can also be found in [130]. We shall see from their work, that Conjectures 1.1 and 1.7 might be solved by answering the following.

Problem 1.10. Let $S \subset S_n$. What is the minimum |S| such that every permutation in S_n agrees with some member of S in at least two positions?

More generally, let f(n, s) be the minimum size of $S \subseteq S_n$ such that every permutation of S_n agrees in at least s positions with some element of S. For example, as every pair of distinct permutations in S_n disagrees in at least two positions we have $f(n, n-1) = |S_n| = n!$. The function f(n, s) has a nice interpretation as a graph. Define the graph $G_{n,s} = (V, E)$, where V consists of the elements of S_n and E consists of the pairs of vertices which agree in at least s places. Then f(n, s) is equal to the size of the smallest dominating set in $G_{n,s}$.

For s = 2, the following is called the Kézdy-Snevily Conjecture [23].

Conjecture 1.11. If n is even then f(n,2) = n. If n is odd then f(n,2) > n.

Using Hall's Theorem [96], Cameron and Ku [22], and Kézdy and Snevily [105], determined the exact value of f(n, 1):

Theorem 1.12. For all n, $f(n,1) = \lfloor \frac{n}{2} \rfloor + 1$.

The next result, due to Cameron and Wanless [23], gives bounds on f(n,2).

Theorem 1.13. For all n > 2, $\left| \frac{n}{2} \right| + 2 \leqslant f(n,2) \leqslant \frac{4}{3}n + O(1)$.

A latin square has a transversal if and only if there is a permutation in S_n that agrees in exactly one position with each of its rows. Thus, f(n,2) > n only if every latin square of order n has a transversal. Hence, Conjecture 1.1 is implied by Conjecture 1.11. The existence of latin squares with no transversal shows that, for even n, $f(n,2) \leq n$.

Proof that Conjecture 1.11 implies Conjecture 1.7 is given in [23]. The idea of the proof is that given L contradictory to Conjecture 1.7, then every element of S_n must either agree in three positions with some row in L, or agree with two rows of L in at least two places. Append the symbol n+1 to each of the rows of L to give a set $S' \subset S_{n+1}$. It is then argued that any permutation in S_{n+1} agrees at least twice with each member of S' which shows that $f(n+1,2) \leq n$, in contradiction to Conjecture 1.11.

For $n \in \{5,7,9\}$, $f(n,2) \le n+1$ follows from the examples given in [23]. These were found by adding a permutation that agrees with every transversal of the latin square

(considered as a permutation) to the rows of a latin square. Keevash and Ku [104] provide constructions that imply bounds on f(n, s) for general s.

Let g(n, s) be the minimum size of a maximal set of permutations of S_n such that every two of them agree in fewer than s positions. For example, g(n, 1) = n since, by Theorem 1.34 in Section 1.12, every latin rectangle can be extended to a latin square. The following conjecture made by Quistorff in 1999 can be found in [130], although using different terminology.

Conjecture 1.14. If n is even then g(n,2) = n. If n is odd then g(n,2) > n.

Quistorff [130] shows that Conjecture 1.11 implies Conjecture 1.14, which itself implies Conjectures 1.1 and 1.7.

Much of the literature about sets of permutations is expressed in terms of Hamming distance and covering radius of the permutation space. We refer the reader to [23], and the surveys [21, 130], for further coverage of the ideas mentioned in this section and their various connections with problems in coding theory.

1.7 Orthogonal latin squares and k-partitions

Two latin squares L_1 and L_2 of order n are said to be *orthogonal mates* if the n^2 ordered pairs in the matrix obtained by superimposition of L_1 and L_2 are distinct. It follows that these n^2 ordered pairs are distinct if and only if the positions of a fixed symbol in L_1 correspond to the positions occupied by a transversal in L_2 . Thus follows a simple but important consequence.

Theorem 1.15. A latin square has an orthogonal mate if and only if it has a decomposition into disjoint transversals.

We often defer to traditional phrasing of the concept. Clearly, the decomposition in Theorem 1.15 is a 1-partition. A set of latin squares that are pairwise orthogonal mates is called a set of mutually orthogonal latin squares (MOLS).

In 1782 Euler conjectured that if $n \equiv 2 \mod 4$, then there is no pair of MOLS of order n [71]. A 1960 disproof of Euler's conjecture is contained in the next, famous, theorem by Bose, Shrikhande and Parker. The proof unfolded over several papers. Parker [124] showed the case n = 10, Bose and Shrikhande [14, 15] proved n = 22 and infinitely many other cases for $n \equiv 2 \mod 4$. A combined effort [16] proved the theorem.

Theorem 1.16. For all n except 2 and 6, there is a pair of MOLS of order n.

As recorded in [150], Theorem 1.16 together with an example of order 6 such as (1.1), show the following.

Corollary 1.17. For all $n \neq 2$, there exists a latin square of order n with a complete plex range.

In 1896, Moore was the first to show that the maximum cardinality of a set of MOLS of order n is n-1, and that this upper bound is achieved if n is a prime power [120]. A set of MOLS which achieves the upper bound is called a *complete* set of MOLS. Bose proved that a complete set of MOLS is equivalent to an affine plane of order n [13]. A prominent open problem is the following [113, p38].

Conjecture 1.18. There exist a complete set of MOLS of order n if and only if n is a prime power.

The next result, by Bruck and Ryser [18], resolves many cases of Conjecture 1.18.

Theorem 1.19. If $n \equiv 1$ or $2 \mod 4$ and there exists a complete set of MOLS of order n, then there exist integers a and b such that $n = a^2 + b^2$.

Even for relatively small n, such as n=10, it remains open to show the exact maximum cardinality of a set of MOLS. Moreover, for n=10 a set of 3 MOLS has not been found despite considerable effort by many people over many decades. See the list of references in [119]. A computational result by Lam, Thiel and Swiercz [112] proved that there is no complete set of MOLS of order 10. Combining their result with a result by Shrikhande [137], shows that the maximum number of MOLS of order 10 is between 2 and 6 inclusive. Several attempts have been made to find 3 MOLS of order 10 by computer search. The most recent, by McKay, Meynert and Myrvold [119], concluded that if there is a set of 3 MOLS of order 10, then each latin square in the set has a trivial paratopy group.

There are many other important open problems, equivalences and a huge amount of interest in orthogonal latin squares and their applications which we do not detail here. For example, see the reference texts [39], [46], [48] and [113], and the numerous references therein. We will apply the valuable result of Theorem 1.16 in our work but our attention is strongly focused toward indivisible k-plexes and latin squares with a restriction of their transversals which results in the absence of a 1-partition.

Other authors have studied latin squares with a closely related restriction (on transversals) that still permits the existence of an orthogonal mate. Danziger, Wanless and Webb [44] introduce the notion of a Δ -crimped latin square, a latin square that contains two elements γ_1 and γ_2 such that there is a transversal through γ_1 but every such transversal includes γ_2 as well. This concept is defined more precisely in [44] using a version of our Lemma 2.1 in Chapter 2, which is crucial to their result. The key observation of [44] is that a Δ -crimped latin square cannot be a member of a set of 3 MOLS.

A k-maxMOLS(n) is a set of k MOLS of order n that is maximal in the sense that it is not contained in any set of k+1 MOLS.

Problem 1.20. For which integers k and n do k-maxMOLS(n) exist?

For k=1, an answer to the problem is given by Theorem 1.23 in the next section. A 1-maxMOLS, has no orthogonal mate and is called a bachelor latin square. A latin square is called monogamous if it has an orthogonal mate, but is not in any set of 3 MOLS. Thus a monogamous latin square and its orthogonal mate are 2-maxMOLS. Danziger, Wanless and Webb [44] show the existence of Δ -crimped latin squares of order $n \notin \{1, 2, 4\}$. By finding orthogonal mates for many of these latin squares they prove the existence of 2-maxMOLS for all n > 6, except perhaps when n = 2p for a prime $p \geqslant 11$. Prior work on the 2-maxMOLS problem includes the work of Drake, van Rees and Wallis [56, 57, 142].

Conjecture 1.21. For all $n \notin \{1, 2, 4, 5, 6\}$, there exist 2-maxMOLS of order n.

Conjecture 1.21 is stated in [44] and by the results stated there, they conclude that, for n < 100, Conjecture 1.21 is only open for $n \in \{22, 26, 34, 38, 58, 62, 74, 86, 94\}$. The status of Problem 1.20 for other values of k is summarised in [39, p190].

The question of the existence of a latin square with a 1-partition is answered by Theorem 1.16. An analogous question for k-plexes is the following.

Problem 1.22. For which integers n and k does there exist a latin square of order n with an indivisible k-partition?

We present an almost complete answer to Problem 1.22 in Chapter 4. We do not rule out the possibility that there exists some indivisible k-plex for k = n. In other words, an indivisible latin square. Note that Problem 1.22 without the condition "indivisible" is easily answered. Theorem 1.16 along with (1.1) yield a k-partition in a latin square of order mk for each order mk except the two cases for k = 1 which are specifically excluded by Theorem 1.16.

1.8 Bachelor latin squares

A bachelor latin square is a latin square with no orthogonal mate. An entry of a latin square is called transversal-free if no transversal includes the entry. A confirmed bachelor square is a latin square that has at least one transversal-free entry.

The question of the existence of a bachelor latin square for each even order n was resolved by Euler in 1779 (e.g. [46, p445]). For $n \in \{1,3\}$ it is easily found that there is no bachelor latin square of order n. In 1944, Mann showed existence of a bachelor for each order n > 3 when $n \equiv 1 \mod 4$ [117]. In general, for orders $n \equiv 3 \mod 4$, the question of existence of bachelor squares remained open until 2006. The next theorem is due to Wanless and Webb [155].

Theorem 1.23. For any positive integer $n \notin \{1,3\}$, there exists a confirmed bachelor square of order n.

The proof of Theorem 1.23 illustrates the base case of a useful lemma, stated in Chapter 2, and one which underpins the main substance of Chapters 3, 4 and 5. A description of the infinite family used to prove Theorem 1.23 is in Section 2.5.2.

Independently, and using a variation of the same technique, Evans [73] showed a similar result, namely, the existence of a bachelor latin square for each order $n \notin \{1,3\}$.

We present a very simple alternative proof of Theorem 1.23 for odd n > 3 in Section 4.3.

In Chapter 4 we extend the above mentioned result by Mann ([117]) to show that very large parts of a latin square may fail to contain not only transversals, but any odd plexes, and that this happens for each order n.

In Chapter 5 we investigate several types of restrictions they may exist on the transversals in a latin square. We consider, for example, the maximum number of transversals in a partition and the number of transversal-free entries in a latin square. We will see that an absence of transversals is not necessary to the existence of some strong restrictions. Our main result is that for all even $n \ge 10$, except perhaps if n is a power of 2, there exists a latin square of order n that possesses a transversal, but every transversal coincides on a single entry. A theorem by Wanless will show that there exist arbitrarily large latin squares of odd order in which the proportion of entries not in a transversal is asymptotic to one ninth.

In 1990 van Rees [142] conjectured the following.

Conjecture 1.24. The proportion of latin squares of order n which are bachelor squares is asymptotically equal to 1.

The conjecture was based upon evidence of the latin squares of order $n \leq 7$. More recent views expressed in [118] and [155] say that the opposite may be true. Evidence for slightly larger orders [119] suggests that the proportion of bachelor latin squares of order n may be rapidly diminishing.

Many of the families of latin squares that we will define in the next chapter are confirmed bachelor families. Our families will be 'close' to some well known families with block structure, in the sense that not many entries differ. In the next sections we review known results for plexes in latin squares with a well defined structure.

1.9 Step type latin squares

A latin square is said to be of *q*-step type if it is of order mq and can be represented by a matrix of $q \times q$ blocks $B_{i,j}$ as follows:

$$\begin{pmatrix}
B_{0,0} & B_{0,1} & \dots & B_{0,m-1} \\
B_{1,0} & B_{1,1} & \dots & B_{1,m-1} \\
\vdots & \vdots & \ddots & \vdots \\
B_{m-1,0} & B_{m-1,1} & \dots & B_{m-1,m-1}
\end{pmatrix},$$

where each block $B_{i,j}$ is a latin subsquare of order q and two blocks $B_{i,j}$ and $B_{i',j'}$ contain the same symbols if and only if $i + j \equiv i' + j' \mod m$.

Euler first showed in 1779 that a cyclic (or 1-step) latin square of even order has no transversals. Later, Maillet in 1894 proved that if m is even and q is odd, then a q-step type latin square has no transversals [46, p446]. As the next theorem by Wanless records, Maillet's result extends to k-plexes [150].

Theorem 1.25. If m is even and q is odd then a q-step type latin square has no odd plex.

In Chapter 2 we generalise Theorem 1.25 to obtain Lemma 2.1 which gives a necessary condition for k-plexes, and is important to our work following. In the next section, an algebraic interpretation of Theorem 1.25 is stated by Theorem 1.29.

1.10 Quasigroups, loops and groups

Much of the theory of latin squares, and transversals, is stated in the context of quasigroups (e.g. [46]).

A quasigroup (Q, \cdot) is a set Q on which the binary operation \cdot satisfies (i) for all $a, b \in Q$ there is a unique $x \in Q$ such that $a \cdot x = b$; and (ii) for all $a, b \in Q$ there is a unique $y \in Q$ such that $y \cdot a = b$. It follows that the unbordered Cayley table of a quasigroup is a latin square. Conversely, for any latin square L there exists some quasigroup Q whose operation \cdot is specified by L. A loop is a quasigroup with an identity element. An associative loop is a group. A result about the existence of plexes in all loops is true for all latin squares. This follows from the statements in Section 1.2 and considering that, for every latin square, there exists an isotopic latin square which has a natural order on the entries in its first row and first column. Such a latin square is said to be in reduced form [113, p4].

The algebraic equivalent of a transversal in a quasigroup is called a complete mapping and was first described by Mann [116]. A complete mapping of a quasigroup (Q, \cdot) is a permutation f of Q such that the mapping $x \to g(x)$ where $g(x) = x \cdot f(x)$ is

also a permutation of Q. Hence the set of ordered triples $\{(x, f(x), g(x) = x \cdot f(x))\}$ form a transversal. The permutation g(x) is called an *orthomorphism*, terminology introduced in [99].

It is well known [48, p7] that any transversal in a group table can be translated by multiplication of its elements by some $g \in G$ to obtain a disjoint transversal. Varying g across all non identity elements of the group results in n-1 disjoint translates of the first transversal. Thus;

Theorem 1.26. If the Cayley table of a group of order n has a transversal then it has n disjoint transversals.

An analogous statement to Theorem 1.26 for k-plexes is false [150]. For any k, multiplication on the elements of a k-plex by a fixed non identity element of the group will yield some k-plex, but one that is not necessarily parallel to the original when k > 1.

The next theorem is due to Wanless [150].

Theorem 1.27. Let G be a group of finite order n with a non-trivial cyclic Sylow 2-subgroup. The Cayley table of G contains no k-plex for any odd k but has a 2-partition and hence contains a k-plex for every even k in the interval $0 \le k \le n$.

Thus, the Cayley table of \mathbb{Z}_n justifies the next statement.

Corollary 1.28. For all even n, there exists a latin square of order n which has an even plex range.

The proof of Theorem 1.27 makes use of the fact that for any normal subgroup N in G, the quotient G/N presents a block structure in the Cayley table. This means that the absence of odd plexes, when G has a non-trivial cyclic Sylow 2-subgroup, follows from Theorem 1.25. The next statement shows Theorem 1.25 in the context of loops. It is due to Pula [128].

Theorem 1.29. If Q is a loop whose Cayley table has an odd plex, then Q has no $N \leq Q$ such that |N| is odd and $Q/N \cong \mathbb{Z}_{2^m}$ for $m \geqslant 1$.

The normality condition for a loop in Theorem 1.29 follows similar notion as it does for groups. A definition is given in [125, p13].

In Corollary 1.33, we see that Theorem 1.29 also holds for more general objects than odd plexes.

Exciting news regarding plexes in group tables is the resolution of a conjecture by Hall and Paige from 1955 [95]. The result, due to the work of many, has now been stated as a theorem in the literature, for example in [128].

Theorem 1.30. If G is a finite group, then the following conditions are equivalent:

- (A) G has a transversal,
- (B) the Sylow 2-subgroups of G are trivial or non-cyclic,
- (C) there is a permutation of the elements of G which yields a trivial product. That is g_1, g_2, \ldots, g_n , such that $g_1g_2 \ldots g_n = 1$ where 1 denotes the identity element of G.

We summarise the developments which lead to Theorem 1.30. First, Paige in 1951 showed that (A) \implies (C) [122]. In 1955, Hall and Paige proved that (A) \implies (B) and showed that the converse holds for all soluble, symmetric and alternating groups. They conjectured that the converse holds for all finite groups and that statement is referred to as the Hall-Paige Conjecture [95]. In 1989, Dénes and Keedwell showed that for non-soluble groups (B) \implies (C) [47]. They observed that proving the Hall-Paige Conjecture was thus equivalent to proving that all non-soluble groups have complete mappings. In 2003, Vaughan-Lee and Wanless [144] gave an elementary proof of (B) \implies (C); the initial proof in [47] invoked the Feit-Thompson Theorem. Other works up to 1992 are reported by Evans in [72]. Work by Dalla Volta and Gavioli [145–147] stated properties of a minimal counterexample to the conjecture. In 2009, Wilcox showed that any minimal counterexample to the conjecture must be simple [156], and thus reduced the possible candidates to the Tits group or a sporadic simple group, some of which had already been proven to possess transversals. Final cases to complete the proof, other than the Janko group J_4 , are shown by Evans [74]. The last case, J_4 , is reportedly proven in correspondence cited in [74] but has not yet been published.

It follows from Theorems 1.26 and 1.30 that Theorem 1.27 completely categorises the plex ranges for finite groups. That is, if G has a non-trivial cyclic Sylow 2-subgroup then its Cayley table has an even plex range, and otherwise it has a complete plex range.

Corollary 1.31. Conjecture 1.3 is true for any Cayley table of a finite group.

Recent work by Pula [128] investigates products of elements in a loop and a framework for non-associative analogues of the Hall-Paige Conjecture. The remainder of this section concerns the article [128].

For a finite loop Q, let $P^k(Q)$ be the set of elements that can be represented as a product containing each element of Q precisely k times. Using this notation, condition (C) of Theorem 1.30 says that $1 \in P^1(G)$. The notation A(Q) denotes the associator subloop of Q, the smallest normal subloop of Q such that Q/A(Q) is a group. So when Q is a group, $A(Q) = \{1\}$.

A class of finite loops Q is said to satisfy the *HP-condition* if for all $Q \in Q$, the following conditions are equivalent:

 (A^*) the Cayley table of Q has a transversal,

- (B*) there is no $N \subseteq Q$ such that |N| is odd and $Q/N \cong \mathbb{Z}_{2^m}$ for $m \geqslant 1$, and
- (C*) the set $P^1(Q)$ intersects A(Q).

Thus, Theorem 1.29 shows that $(A^*) \implies (B^*)$ for all loops. Pula shows that $(B^*) \iff (C^*)$ holds for all loops. The results by Pula are actually shown to hold for a wider class of structures than k-plexes, as follows.

A row k-plex of a latin square is a collection of triples representing each row exactly k times. A subset $C = \{(x_i, y_i, z_i) : 1 \le i \le m\} \subseteq L$ is call column-entry regular, or just regular for short, if for each symbol s we have $|\{i : y_i = s\}| = |\{i : z_i = s\}|$. That is, s appears as a symbol the same number of times it appears as a column. We denote by C_r , the multiset of indices appearing as rows in C. For example, if C is a k-plex then C_r contains precisely k copies of each index. The next theorem and comments following are also from [128].

Theorem 1.32. If C is a regular subset of the Cayley table of Q, then $P^1(C_r)$ intersects A(Q). In particular, if Q has a k-plex, or just a regular row k-plex, then $P^k(Q)$ intersects A(Q).

Corollary 1.33. If a loop fails to satisfy (B^*) then it has no regular row odd plexes. Thus, Theorem 1.29 also applies to regular row odd plexes in Q.

However, the HP-condition does not hold for all loops. The following example has no transversal but satisfies condition (B^*) . The shading shows a regular row transversal.

In Chapter 3 we show latin squares, for each even order $n \ge 8$, which also fail the HP-condition.

For all odd ordered loops, condition (B*) is always true and so the implication (B*) \implies (A*) is equivalent to Conjecture 1.1. More generally, the question of whether a loop has a 2-plex if and only if $P^2(Q)$ intersects A(Q) would resolve Conjecture 1.2. The condition $P^2(Q)$ intersects A(Q) is proven for all loops [128].

1.11 Diagonally cyclic latin squares

We continue to review results relating to latin squares with a defined structure.

A latin square L is a diagonally cyclic latin square or DCLS if $(x,y,z) \in L \iff (x+1,y+1,z+1) \in L$ where $\mathcal{I}(L) = \{0,1,\ldots,n-1\}$ and calculations are in \mathbb{Z}_n . Thus any DCLS is completely determined by a single row or column. We say that row 0 generates a DCLS if and only if $f(i) = 0 \cdot i$ and g(i) = f(i) - i are permutations of $\{0,1,\ldots,n-1\}$. Thus, f and g guarantee distinct symbols in the rows and columns, respectively. Also, f is a complete mapping of the elements of \mathbb{Z}_n so we see, by the definition of a complete mapping, that every DCLS of order n corresponds to a transversal of the Cayley table of \mathbb{Z}_n . Conversely, given a transversal of the Cayley table of \mathbb{Z}_n , say $T = \{(x,y,x+y)\}$, then row 0 of a DCLS is determined by $f(y) = 0 \cdot y = x+y$ for each element of T. If n is even, by Theorem 1.25, the Cayley table of \mathbb{Z}_n has no transversal so there is no DCLS of even order. More directly, if n is even then $\sum_{i=1}^{n-1} f(i) \equiv \frac{1}{2}n$ but $\sum_{i=1}^{n-1} g(i) = 0$ [91].

DCLSs have a long history of study in connection with orthogonal latin squares and combinatorial designs. We refer the reader to [152] for further results and detailed review of the history of investigation into DCLSs and their applications.

It follows immediately from the definition that a DCLS has the property that every diagonal consists of a cyclic transversal or cyclically generated transversal. That is, for a fixed $\alpha \in \mathcal{I}(L)$, the diagonal $D_{\alpha} = \{(x, \alpha + x, 0 \cdot \alpha + x) : x \in \mathcal{I}(L)\}$ is a transversal. In the next section we consider problems about sets of cyclically generated transversals.

A DCLS is a latin square possessing a cyclic automorphism of order n. More generally, as described in [152], a bordered diagonally cyclic latin square of type B_b includes latin squares with a cyclic automorphism of order n-b. A B_0 type is a DCLS. Collectively, the B_b type latin squares are named Parker squares [152], in recognition of Parker's work on constructing sets of orthogonal latin squares involving these types of latin squares [123]. Parker latin squares are useful to us in Chapter 4 where we consider plexes in latin squares that contain subsquares.

1.12 Partial latin squares and completability

In Section 1.3 we defined a k-plex as a subset of a latin square. We may also consider a k-plex as a partial latin square. Indeed, some of the earliest results about k-plexes, other than small order studies, appear in the context of partial latin squares [4], [20], [45]. The definition of a PLS and information in this section is useful in Sections 4.2.2 and Sections 4.4 where we are interested in extending plexes into larger latin squares.

A partial latin square (PLS) of order n is a subset P of $R \times C \times S$, where each of R, C and S is a set of cardinality n, such that no two distinct elements of P agree in more than one coordinate. Often R = C = S and we then refer to this set as the index set of P, and denote it by $\mathcal{I}(P)$. A PLS P of order n with $|P| = n^2$ is a latin square. The size of a PLS P is the cardinality of P.

A PLS of order n is *completable* if it is a subset of some latin square of order n.

Colbourn [38] showed that the decision problem Is a PLS of order n completable? is an NP-complete problem.

Classical results of M. Hall [94] and Ryser [133] apply Hall's Theorem [96] to state necessary and sufficient conditions for the extension of a latin rectangle to a latin square. The result of [133], stated next, is known as Ryser's Theorem.

Theorem 1.34. An $r \times s$ latin rectangle R with entries from n symbols is completable to a latin square of order n if and only if each symbol occurs in R at least r + s - n times.

Another classical result is the following, due to Evans [75].

Theorem 1.35. For any integers t and $n \ge 2t$, a $t \times t$ latin subsquare is completable to a latin square of order n.

In answer to a well known problem posed in 1960 by Evans [75], which is later referred to as the Evans Conjecture, Smetaniuk [138] proved that any PLS of order n and size at most n-1 is completable. Independently, Andersen and Hilton [6] obtained the same result. Earlier work by Häggkvist [92] proved the case for $n \ge 1111$. The question of completability again arises by insisting that further structure is imposed on the PLS. For example, the next conjecture made in 1980 by Häggkvist is stated in [50].

Conjecture 1.36. Any PLS of order nr whose elements occur in (n-1) disjoint subsquares of order r is completable.

Conjecture 1.36 is true for r = 1, by proof of the Evans Conjecture, and for r = 3 by Denley and Häggkvist [50]. Other special cases are proven in [49] and [109]. Further interest, see for example the survey [53], includes PLSs which are uncompletable or have precisely one completion, and which are critical with respect to this property. The latter are called *critical sets* and are closely connected with *latin trades* which we will discuss in the next section. We first mention some results and problems concerning PLSs of the kind closest to our interest.

A k-plex of order n is a completable PLS of order n and size kn and with the property that each row, column and symbol occurs k times.

Example (1.3) is a PLS of order 5 and size 10 such that each row, column and symbol occurs twice. However, by inspection, it is not completable and is therefore not a 2-plex.

$$\begin{pmatrix}
0 & 1 & \cdot & \cdot & \cdot \\
1 & 0 & \cdot & \cdot & \cdot \\
\cdot & \cdot & 2 & 3 & \cdot \\
\cdot & \cdot & \cdot & 4 & 2 \\
\cdot & \cdot & 4 & \cdot & 3
\end{pmatrix}$$
(1.3)

Following terminology introduced in [19], we call such PLSs, which may or may not be completable, protoplexes and define them as follows. A k-protoplex of order n is a PLS of order n in which each row, column and symbol occurs exactly k times. Thus, a k-plex is equivalent to a completable k-protoplex. The definition of k-plex that we use follows [19, 67] and is a restriction of the original definition in [150]. In defining a k-plex as a PLS, Wanless [150] allowed that a k-plex be not necessarily completable; that is, a k-protoplex. Our definition of a protoplex is very similar to that of a homogeneous PLS which we discuss in Section 1.13. However, unlike protoplexes, homogeneous PLSs are allowed to have empty rows and columns.

The following problem is posed by Donovan in [53].

Problem 1.37. For which integers k and n is there an uncompletable k-protoplex of order n?

A partial answer to Problem 1.37 is given next, by Wanless [150].

Theorem 1.38. If 1 < k < n and $k > \frac{1}{4}n$, then there exists an uncompletable k-protoplex of order n.

Conjecture 1.39, stated next, can be found in Daykin and Häggkvist [45]. They proved that if n = 16m, where m is a positive integer and $k \leq \sqrt{n}/128$, then a k-protoplex is completable. In 1996 Burton [20] showed by computation that all 2-protoplexes of order 8 are completable. Having evidence of uncompletable 2-protoplexes for smaller order he also states the conjecture.

Conjecture 1.39. For $k \leq \frac{1}{4}n$ every k-protoplex of order n is completable, hence is a k-plex of order n.

The next partial extension result, due to Burton [20], can be found in [150].

Theorem 1.40. For $k \leq \frac{1}{4}n$, every k-protoplex of order n is contained in a (k+1)-protoplex of order n.

If Conjecture 1.39 is true, then the k-protoplex in Theorem 1.40 is a k-plex.

In Section 4.2.2 we further consider completability of k-protoplexes of order 2k and thereby derive results about indivisible plexes and partitions. The existence of arbitrarily large indivisible plexes follows from the next theorem by Wanless [150].

Theorem 1.41. For arbitrary k and $n \ge k^2$, there exists an indivisible k-protoplex of order n.

Our results in Chapter 4 realise a suggestion in [150] that the bound k^2 in Theorem 1.41 might be improved. It will follow from our results that, for k > 1, there exist indivisible k-plexes (hence indivisible k-protoplexes) of order $n \in \{2k, 2k + 1\}$.

The next problem posed by Alspach and Heinrich in 1990 [4], in connection with their work on combinatorial designs, concerns protoplexes consisting of parallel transversals.

Problem 1.42. For each k, does there exist an integer N(k) such that any k-protoplex consisting of k transversals of order $n \ge N(k)$ is completable?

Grüttmüller [91] poses the following restriction of Problem 1.42 to diagonally cyclic latin squares.

Problem 1.43. Does there exist for each k an odd integer C(k) such that any k-protoplex of odd order $n \ge C(k)$ and consisting of k cyclically generated transversals is completable to a DCLS?

An idempotent latin square is a latin square L such that $x \cdot x = x$ for all $x \in \mathcal{I}(L)$. As noted in [91], the existence of an idempotent latin square for all $n \neq 2$ shows that N(1) = 3. Grüttmüller [91] proved that:

Theorem 1.44. For all $k \ge 1$, $N(k) \ge 4k - 1$.

Theorem 1.45. For all $k \ge 3$, $C(k) \ge 3k - 1$.

Grüttmüller further proved that C(2) = 3 [90]. Computational results [91] prove that a PLS P, consisting of k cyclically generated transversals, is completable to a DCLS if $2 \le k \le 7$ and n is an odd integer in the range $3k - 1 \le n \le 21$. Also, [91] shows that P is completable (non-cyclically) for k = 2, 3 or 4 and arbitrary n such that $4k - 1 \le n \le 21$. This supports the following conjectures by Grüttmüller [91].

Conjecture 1.46. For $k \ge 3$, N(k) = 4k - 1.

Conjecture 1.47. For $k \ge 3$, C(k) = 3k - 1.

As noted [34], it has not been shown that N(2) exists. Cavenagh, Hämäläinen and Nelson [34] recently proved the following which may suggest that C(3) = 9.

Theorem 1.48. For prime p > 7, a PLS consisting of three cyclically generated transversals of order p is completable to a DCLS.

As noted in Section 1.11, a DCLS of order n consists of n disjoint cyclically generated transversals and there is a bijection between the set of DCLSs of order n and transversals in the Cayley table of \mathbb{Z}_n . Similarly, an equivalent statement to Theorem 1.48 is that every partial transversal of size 3 in the Cayley table of \mathbb{Z}_p , for prime p > 7, can be extended to a transversal.

1.13 Latin trades and homogeneous partial latin squares

A critical set is a uniquely completable PLS P of order n such that every proper subset of P has at least two different completions to a latin square of order n. For a given order, finding the set of achievable sizes of critical sets and the smallest and largest such sizes are among questions of interest e.g. [53]. It has been conjectured by Bate and van Rees [9] that the smallest size of a critical set of a latin square of order n is $\lfloor n^2/4 \rfloor$. We refer the interested reader to a survey by Keedwell [103] for more detailed information on the study of critical sets. It is an area of research closely connected with the study of latin trades.

A PLS T is said to be a *latin trade* if $T \neq \emptyset$ and there exists a PLS T' with *size* |T| = |T'|, and for each $(x, y, z) \in T$ there exists a unique $x' \neq x$, $y' \neq y$ and $z' \neq z$ such that $(x', y, z) \in T'$, $(x, y', z) \in T'$, and $(x, y, z') \in T'$. The pair (T, T') is called a *latin bitrade*.

A latin trade T may be thought of as a non-empty subset of a latin square which may be replaced with another set T' to obtain a new latin square of the same order. That is, $T \subset L$ is a latin trade if there exists a PLS T' with $T \cap T' = \emptyset$ and such that $(L \setminus T) \cup T'$ is a latin square. Thus a bitrade identifies the different elements of two latin squares of the same order. Some early studies, by Drápal and Kepka, used the name exchangeable partial groupoids to describe latin bitrades [63, 64].

A latin trade is *minimal* if no proper subset of it is a latin trade. Much attention has focused on minimal latin trades because of their close connection with critical sets. That well known connection (e.g. [30]) is as follows.

Lemma 1.49. If C is a critical set of a latin square L, then every latin trade $T \subseteq L$ must intersect L in at least one element. For each element (x, y, z) in C there exists a unique minimal latin trade $T \subseteq L$ such that $T \cap L = (x, y, z)$.

A summary of the theory of latin bitrades and their applications is contained in a survey by Cavenagh [30].

Latin trades can be used in the generation of random latin squares [33] and the possibility of compact storage for large catalogues of latin squares is discussed in [151].

As mentioned in [30], latin trades have been used to obtain existence proofs like Theorem 1.23 by [155]. Indeed, most of the work in our later chapters will use a latin trade in the Cayley table of an Abelian group. The trades that we employ are designed to work well with Lemma 2.1 which is stated in the next chapter. Some of them will closely resemble, or could be adapted from, those described in [27] and which involve 3 rows of a latin square. Otherwise they might be obtained using a technique referred to as cycle switching in [151]. Of course, as noted in [29], any latin

square may be thought of as involving some latin trade with another latin square of the same order.

Recent studies of latin trades have identified several interesting connections between latin trades, geometry, topology and permutations. We mention these, briefly, as we next review the rather extensive literature on latin trades. Much of this work is very recent. A reader who is new to the topic may find it helpful to also refer to the survey [30]. Information in this section is presented for context only and it will not be used in our later work.

In contrast to our own designs of latin trades which will be detailed in the next chapter, those latin trades which themselves closest resemble a k-plex are the following. Our review will focus on homogeneous latin trades.

A k-homogeneous latin trade for $k \ge 2$ is a latin trade which intersects each row, each column and each symbol of the latin square either 0 or k times. Hence the size of a k-homogeneous latin trade is equal to km for some integer m.

A minimal 2-homogeneous latin trade is an intercalate and any 2-homogeneous latin trade is necessarily a union of intercalates.

According to Cavenagh [29], one reason for interest in k-homogeneous latin trades is that they often have, compared with other minimal latin trades, large size with respect to the order of the latin square they are contained in and so may be useful in locating small critical sets. Cavenagh also notes that the k-homogeneous property is invariant under conjugacies, and that k-homogeneous latin trades often partition into disjoint partial transversals and thus have implications for partial orthogonality.

Construction of k-homogeneous latin trades for each $k \ge 2$ is shown by Bean et al. [10] but these are not required to be minimal.

Cavenagh, Donovan and Drápal [24] use a geometric method, a hexagonal packing of the Euclidean plane with circles where arcs identify a labelling of rows, columns and symbols, to prove existence of 3-homogeneous latin trades of size 3m for each $m \ge 3$. It is further shown by Cavenagh [29] that the construction in [24] determines all 3-homogeneous latin trades. Thus, in contrast to 2-homogeneous latin trades:

Theorem 1.50. Every 3-homogeneous latin trade is composed of 3 disjoint partial transversals.

It is now known, for example [30], that Theorem 1.50 is implied by an earlier and more general result from 1983 by Negami [121]. Negami's result states that any 6-connected graph with genus 1 is uniquely embeddable on the torus. Other recent proofs of Theorem 1.50 appear in [89, 97, 115]. Grannell, Griggs and Knor [89] point to an even earlier result from 1973, by Altshuler [5], and present an exposition of the relationship of the classification problem for 3-homogeneous latin trades with two other combinatorial classification problems. The alternative proof of Theorem 1.50 in [89] is based on the work of Altshuler and Negami.

Another geometric construction, due to Cavenagh, Donovan and Drápal, involves rectangular packings of the plane with unit circles, where labels are derived by specified intersections and line segments [25]. The authors show 4-homogeneous latin trades of size 8m for each integer $m \ge 4$.

Cavenagh, Donovan and Yazici [31] construct minimal k-homogeneous latin trades of size km for $k \ge 3$ and $m \ge 1.75k^2 + 3$. It is noted in [31] that these trades partition into disjoint transversals.

Although every latin trade can be completed to some latin square, a latin trade of order n need not be a completable PLS (of order n). We will discuss the *embedding* order of a latin trade later in this section. The question of whether or not a latin trade embeds into a latin square of some specific form, for example the Cayley table of an Abelian group, is certainly of further interest to other researchers and in our own pursuits. Cavenagh [28] gives a construction method using the trades of [24] to prove that:

Theorem 1.51. For even m there exists a 3-homogeneous latin trade of size $3m^2$ in the Cayley table of $(\mathbb{Z}_2)^m$.

A well known result by Drápal [58] is the following.

Theorem 1.52. Let m and n be positive integers. Suppose that we can partition the area of an equilateral triangle of side n into m smaller (integer-sided) equilateral triangles, such that each vertex of a triangle occurs as the vertex of at most 3 of the smaller triangles. Then there exists a spherical latin bitrade (T, T') of size m such that T embeds into the Cayley table of \mathbb{Z}_n .

We use the terminology and notation of [35] to explain the adjective "spherical" in the last sentence of the theorem.

Each row x of a latin bitrade (W, B) defines a permutation ψ_x of the symbols of row x, where $\psi_x(z) = z'$ if and only if $(x, y, z) \in W$ and $(x, y, z') \in B$ for some y. If ψ_x is a single cycle, then we say that the row x is separated. Similarly, we may determine that each column and symbol of the bitrade is either separated or non-separated. If every row, column and symbol is separated, then we say that the latin bitrade (W, B) is separated.

Any non-separated bitrade can be transformed into a separated bitrade by sending offending entries of (W, B), say in a smaller cycle of ψ_x , to an empty row. For an example see [30] or [35]. Performing a similar operation, the column and symbol cycles can also be separated. Thus, in some sense, all latin bitrades are considered by the class of separated bitrades.

A latin bitrade (W, B) is said to *primary*, or *connected*, if there exists no latin bitrades (W', B') and (W'', B'') such that $W \cap W'' = \emptyset$, $W = W' \cup W''$ and $B = B' \cup B''$.

Let $R_W = \{r_x : (x, y, z) \in W\}$, $C_W = \{c_y : (x, y, z) \in W\}$ and $S_W = \{s_z : (x, y, z) \in W\}$. Clearly $R_W = R_B$, and similarly for the columns and symbols.

Given a separated, connected latin bitrade (W, B), we may construct a triangulation $\mathcal{G}_{W,B}$ whose vertex set is $R_W \cup C_W \cup S_W$ and whose edges are pairs of vertices corresponding to some triple $(x, y, z) \in W$, or equivalently within some triple of B. If we define white and black faces of $\mathcal{G}_{W,B}$ to be the triples from W and B, respectively, then $\mathcal{G}_{W,B}$ is a face 2-colourable triangulation of some surface [59].

We now orient the edges $\mathcal{G}_{W,B}$ so that each white face contains directed edges from a row to a column, from a column to a symbol and from a symbol to a row. The subgraph induced by the vertices of a face, black or white, has both in-degree and out-degree of each vertex equal to 1. This shows that the surface in which $\mathcal{G}_{W,B}$ is embedded is orientable. Therefore, we may associate with $\mathcal{G}_{W,B}$ the non-negative integer value g for the genus of $\mathcal{G}_{W,B}$ given by Euler's genus formula:

$$g = \frac{1}{2}(2 - f + e - v)$$

$$= \frac{1}{2}(2 - (|W| + |B|) + 3|W| - (|R_W| + |C_W| + |S_W|))$$

$$= \frac{1}{2}(2 + |W| - |R_W| - |C_W| - |S_W|).$$

Thus, we may now explain fully the statement of Theorem 1.52. A spherical or planar latin bitrade is a latin bitrade of genus 0. A spherical latin trade W is a latin trade for which there exists some disjoint mate B such that (W, B) is a spherical latin bitrade. In general, there may exist more than one disjoint mate for W but Cavenagh and Wanless show in [35] that if W is spherical, then the spherical bitrade (W, B) is uniquely determined. In fact they show that any other disjoint mate B' of W is neither primary nor separated. (Hence an associated genus does not exist.)

Lefevre et al. [114] construct, for each genus $g \ge 0$, a latin bitrade of smallest possible size and also a minimal latin bitrade of size 8g + 8.

A main result of Cavenagh and Wanless [35], and one independently shown using an entirely different method by Drápal, Hämäläinen and Kala [61], is the following.

Theorem 1.53. Every spherical latin trade can be embedded in a finite Abelian group.

Cavenagh and Wanless [35] also show that, for every $g \ge 1$, there is a latin trade of genus g which cannot be embedded in any group, and another latin trade of genus g which can be embedded in the Cayley table of \mathbb{Z}_n for some n.

Drápal, Hämäläinen and Kala consider methods for generating dissections of equilateral triangles to arrive at their proof of Theorem 1.53 in [61]. Their approach was motivated by a recent finding of Cavenagh and Lisoněk [26] that planar Eulerian triangulations are equivalent to spherical latin bitrades. In other words, spherical latin bitrades are equivalent to cubic 3-connected bipartite planar graphs.

Drápal, Hämäläinen and Rosendorf [62] report on the computer generation of spher-

ical bitrades up to size 24. An earlier computer enumeration and classification of small latin trades up to size 19, by Wanless [153], also determines the respective $embedding\ order$, the smallest order n such that a latin square of order n contains the latin trade. Species representatives of each latin trade are available at the author's homepage [149].

Several of the recent results on latin trades refer to Drápal's representation of a latin bitrade as a set of three permutations. The idea, which we explain below, is now published in [60] but was earlier circulated at a workshop in Prague in 2003. Also [32] provides proof of the main concepts. Some further terminology is required.

Let (W, B) be a connected and separated latin bitrade. Define the map $\beta_d : W \to B$ where $d \in \{1, 2, 3\}$ and $\beta_d(w_1, w_2, w_3) = (b_1, b_2, b_3)$ implies that $w_d \neq b_d$ and $w_i = b_i$ for $i \neq d$. In other words, with two coordinates fixed, the (different) d-th coordinate identifies a unique triple in B. In addition, let $\tau_1, \tau_2, \tau_3 : W \to W$ be given by $\tau_1 = \beta_2 \beta_3^{-1}$, $\tau_2 = \beta_3 \beta_1^{-1}$ and $\tau_3 = \beta_1 \beta_2^{-1}$. Note that τ_i leaves the i-th coordinate of a triple fixed. The three permutations $\{\tau_1, \tau_2, \tau_3\}$ are called the permutation representation of (W, B).

In the reverse direction, a set of permutations with particular properties defines a latin bitrade. Let $\text{Mov}(\pi)$ denote the set of points that a permutation π acts on. Let τ_1, τ_2, τ_3 be permutations and let $\Omega = \text{Mov}(\tau_1) \cup \text{Mov}(\tau_2) \cup \text{Mov}(\tau_3)$. Define four properties:

- (T1) $\tau_1 \tau_2 \tau_3 = 1$,
- (T2) if ρ_i is a cycle of τ_i and ρ_j is a cycle of τ_j , then $|\text{Mov}(\rho_i) \cap \text{Mov}(\rho_j)| \leq 1$ for $1 \leq i < j \leq 3$,
- (T3) τ_1, τ_2 and τ_3 have no fixed points,
- (T4) the group $\langle \tau_1, \tau_2, \tau_3 \rangle$ is transitive on Ω .

The next theorem, by Drápal [60], states how these properties relate to latin bitrades.

Theorem 1.54. A latin bitrade (T,T') is equivalent (up to isotopism) to three permutations τ_1, τ_2 and τ_3 acting on a set Ω and satisfying properties (T1), (T2) and (T3). If property (T4) is also satisfied then (T,T') is primary. If $\{\tau_1, \tau_2, \tau_3\}$ satisfy (T1), (T2) and (T3) then let $\mathcal{A}_i = \{\rho : \rho \text{ is a cycle of } \tau_i\}$. Then, a separated latin bitrade (T,T') of size $|\Omega|$, and with $|\mathcal{A}_1|$ rows, $|\mathcal{A}_2|$ columns and $|\mathcal{A}_3|$ symbols is specified by

$$T = \{(\rho_1, \rho_2, \rho_3) : \rho_i \in (A)_i \text{ and the } \rho_i \text{ all act on a common point of } \Omega\},$$

and

$$T' = \{(\rho_1, \rho_2, \rho_3) : \rho_i \in (A)_i \text{ and } x, x' \text{ and } x'' \text{ are distinct points of } \Omega \text{ such that } x\rho_1 = x', x\rho_2 = x'' \text{ and } x''\rho_3 = x\}.$$

Lefevre, Donovan and Drápal [115] use a permutation representation to provide a shorter proof, than originally shown by [24] and [25], of the existence of 3 and 4-homogeneous latin trades. The authors [115] also give a further construction for 4-homogeneous latin trades.

Again via a permutation representation, Cavenagh, Drápal and Hämäläinen [32] show how some latin bitrades may be directly derived from groups. They study the special case when the set Ω is the set of elements of a group G and the group acts on itself by right translation.

Let G be a finite group and let a, b and c be distinct non-identity elements of G. Let $A = \langle a \rangle$, $B = \langle b \rangle$ and $C = \langle c \rangle$. Define three properties:

(G1)
$$abc = 1$$
,

(G2)
$$|A \cap B| = |A \cap C| = |B \cap C| = 1$$
,

(G3)
$$\langle a, b, c \rangle = G$$
.

The main tool of [32], and used in conjunction with a number of special conditions to yield particular properties, is the following.

Theorem 1.55. If properties (G1) and (G2) are satisfied, then a latin bitrade (T°, T^{\star}) of size |G| is given by

$$T^{\circ} = \{ (gA, gB, gC) : g \in G \}, \quad T^{\star} = \{ (gA, gB, ga^{-1}C) : g \in G \}.$$

If (G3) is also satisfied then (T°, T^{\star}) is primary.

Some further properties of the bitrade thus constructed are next listed [32].

- (T°, T^{\star}) has |G: A| rows each with |A| entries, |G: B| columns each with |B| entries and |G: C| symbols each occurring |C| times.
- Thus if |A| = |B| = |C| = k, then T° (hence T^{\star}) are k-homogeneous latin trades.
- T° is equivalent to the following sets.

$$\{(g_1A, g_2B, g_3C): g_1, g_2, g_3 \in G \text{ and } |g_1A \cap g_2B \cap g_3C| = 1\}, \text{ and } \{(g_1A, g_2B, g_3C): g_1, g_2, g_3 \in G \text{ and } g_1A \cap g_2B = \{g_3\}\}.$$

Orthogonality and minimality can also be encoded, however we omit the detail for these properties. In one example, using a non-Abelian group of order p^3 where p is an odd prime, the authors [32] show that for each prime p there exists a minimal primary p-homogeneous latin bitrade of size p^3 . In general they prove the existence of minimal, k-homogeneous latin trades for each odd $k \ge 3$.

Behrooz, Bagheri and Mahmoodian [12] give a recursive construction for k-homogeneous latin bitrades in their proof of the next theorem. They also prove the existence of a k-homogeneous latin bitrade of size km for several more cases; for all odd k and $m \ge k$, for all $m \ge k$ where $9 \le k \le 37$, and a special case when k is even

Theorem 1.56. Let $m \ge k$ and $m' \ge k'$. If there exists a k-homogeneous latin bitrade of size km and a k'-homogeneous latin bitrade of size k'm', then there exists a kk'-homogeneous latin bitrade of size kmk'm'.

The next theorem, by Cavenagh and Wanless [36], gives the spectrum of possible sizes of k-homogeneous latin bitrades. It proves a conjecture in [12].

Theorem 1.57. For all $k \ge 3$, a k-homogeneous latin bitrade of size s exists if and only if k divides s and $s \ge k^2$.

The authors [36] conjecture that Theorem 1.57 completely determines all possible sizes of primary k-homogeneous latin bitrades. The proof of Theorem 1.57 shows a connection between k-homogeneous latin trades and transversals in the Cayley table of \mathbb{Z}_n . We mention this further in the next section.

In connection with Problem 1.5 we ask the following.

Problem 1.58. For which integers n and k is there an indivisible k-plex of order n that is a (k-homogeneous) latin trade?

Unfortunately, other than Theorem 1.50 and the obvious indivisibility of 2-homogeneous latin trades, classification according to divisibility is not readily found in the literature. Many of the constructions are elaborate and their proofs are non trivial. Moreover, as illustrated by Theorem 1.55, one generic construction may yield various examples. Incidentally, we observe that a minimal 4-homogeneous latin trade given by Lemma 40 of [25] contains 2 disjoint transversals. It is not obvious that similar divisibility would hold in general as any mapping described by their Remark 38 yields a 4-homogeneous latin trade. An example, given in [25] and [30], of a 4-homogeneous latin trade of size 32 in the Cayley table of $\mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_2$ is divisible. Our data in Section 4.5 shows that there is no indivisible 4-plex in the Cayley table of this group.

1.14 The number of plexes and partitions

1.14.1 Small order studies

As mentioned in Section 1.2, the number of k-plexes and partitions in a latin square is a species invariant. Thus, to consider all plexes among the latin squares of order n,

we need only consider a list of species representatives of order n. However, except for $n \leq 8$, such a list is either very large or unavailable. We rely on a program used in [119] to generate the isotopy or species representatives of order 9. This program was also used in [118] to prove Conjecture 1.1 for the latin squares of order $n \leq 9$. We refer the reader to [119] for a history and details of the investigation of the number of latin squares of small order. The text [101] also gives information on algorithmic issues relevant to the classification of latin squares.

Detailed investigations of the plexes and partitions of latin squares of order $n \leq 6$ are given by the collected works of Finney [76–79], Freeman [84–86] and Johnston and Fullerton [100]. Several of the early works on plexes for k > 1 are well prior to modern technologies so these reports would have taken an impressive number of human hours to compute.

Killgrove et al. [106] give data on the number of transversals and 2-plexes in latin squares of order 6 and 7. Wanless [150] includes the first enumerative data on plexes for species of order 8 and identifies the maximum number of disjoint transversals in each species.

Data on the number of transversals in species of order 9, group tables of order $n \leq 23$, and turn-squares of order 14 is presented in [118].

A research problem [101, p265] suggests a computational check of Conjectures 1.1 and 1.2 for order 9. As noted above, Conjecture 1.1 for order 9 was shown in [118]. Our results in Section 4.5.3 will verify Conjecture 1.2 for order 9.

In Section 4.5 we report on extensive computations of indivisible plexes and partitions in latin squares of order $n \leq 9$. Although several earlier reports in the literature consider plexes in small latin squares, and as mentioned in Section 1.3, Finney [77, 78] found indivisible plexes, the value of $\kappa(n)$ for small n has not been explicitly computed. We compute $\kappa(n)$ exactly for $n \leq 8$ and prove that $\kappa(9) \in \{6,7\}$. We enumerate every different type of indivisible partition for every species representative of order $n \leq 8$.

In Section 5.4 we report on further computations concerning transversals in latin squares of order $n \leq 9$. We will report on the number of transversal-free entries and the number of transversals in a partition. As mentioned in Section 1.3, we will prove Conjecture 1.3 for order 9.

Another useful outcome of our computations is finding latin squares with interesting properties. Generalisation of a small example might lead to proving existence in general of an interesting property. For example, the proof of Theorem 5.6 in the final chapter developed in this manner. The data also provides a solid resource for testing ideas and the development of further theory.

1.14.2 General cases

Conjecture 1.1 is attributed to Ryser but is a weakening of his original statement [134]. Ryser originally conjectured that the number of transversals of a latin square is odd if and only if the latin square is of odd order. The original statement is disproved by the existence of latin squares of odd order, for example of order 7, with an even number of transversals. However, the latin squares of even order satisfy Ryser's original statement as proved by Balasubramanian [8].

Theorem 1.59. The number of transversals in a latin square of even order is even.

As observed in [150], Theorem 1.59 does not generalise in an obvious way to k-plexes. In Section 4.5, we will see that there exist latin squares of even order that possess an odd number of k-plexes for k > 1.

Following notation in [118], we define

 $T(n) = \max \{ t : \text{ some latin square of order } n \text{ possesses } t \text{ transversals} \}.$

McKay, McLeod and Wanless [118] proved the following.

Theorem 1.60. For $n \geqslant 5$,

$$15^{n/5} \leqslant T(n) \leqslant c^n \sqrt{n} \ n!$$

where
$$c = \sqrt{\frac{3-\sqrt{3}}{6}}e^{\sqrt{3}/6} \approx 0.61354$$
.

Also following [118], define

 $t(n) = \min \{ t : \text{ some latin square of order } n \text{ possesses } t \text{ transversals} \}.$

As Conjecture 1.1 is open, $t(n) \ge 0$. We know by Theorem 1.25 that t(n) = 0 for all even n. For odd n, understanding t(n) is apparently difficult. As follows, there are some results about the number of transversals in the Cayley table of \mathbb{Z}_n for odd n.

For odd $n \ge 3$, let t_n denote the number of transversals in the Cayley table of \mathbb{Z}_n .

Vardi [143] conjectured the following. The conjecture was made in connection with study of an equivalent problem known as the *toroidal n-queens problem*. The equivalence is explained in [154].

Conjecture 1.61. There are constants c_1 , c_2 such that $0 < c_1 < c_2 < 1$ and

$$c_1^n n! \leqslant t_n \leqslant c_2^n n!$$

Clark and Lewis [37] make the following conjecture.

Conjecture 1.62.

$$t_n \geqslant n(n-2)(n-4)\dots(3)(1).$$

Cavenagh and Wanless [36] showed that t_n is at least exponential in n. Combining their result with the upper bound given by Theorem 1.60 proves the next theorem.

Theorem 1.63. If n is sufficiently large then

$$(3.246)^n < t_n \le (0.614)^n n!$$

Theorem 1.63 improves on some earlier bounds given by Cooper and Kovalenko [41, 43, 108] and Rivin, Vardi and Zimmermann [131]. Estimates of t_n are also given by Cooper et al. [42]. More recently, Kuznetsov [110, 111] use a fast computational method to estimate t_n for considerably large n, up to n = 205.

Other results, which we next mention, concern the divisors of t_n , and more generally, divisors of the number of transversals in latin squares with some specified structure.

The next two theorems are due to McKay, McLeod and Wanless [118].

Theorem 1.64. The number of transversals in a symmetric latin square of order n is congruent to $n \mod 2$.

Theorem 1.65. If G is a group of order $n \not\equiv 1 \mod 3$, then the number of transversals in the Cayley table of G is divisible by 3.

Let $z_n = t_n/n$. It should be clear that z_n is the number of transversals through a fixed entry in the Cayley table of \mathbb{Z}_n (n is odd) and is independent of the entry chosen.

The following three theorems by Stones and Wanless [140] follow from their study of orthomorphisms of \mathbb{Z}_n .

Theorem 1.66. For odd n

$$z_n \equiv \begin{cases} -2 \bmod n & if \ n \ is \ prime, \\ 0 \bmod n & otherwise. \end{cases}$$

Theorem 1.67. If n is a prime of the form $2 \times 3^k + 1$, then $z_n \equiv 1 \mod 3$.

Theorem 1.68. For odd n

$$z_n \equiv \begin{cases} 0 \mod 3 & \text{if } n \not\equiv 1 \mod 3 \text{ and } n \geqslant 5, \\ \zeta(n) \mod 3 & \text{if } n \equiv 1 \mod 3. \end{cases}$$

where $\zeta(n)$ is the number of partitions of $\{1, 2, ..., n\}$ into parts of size 3 in which each part has sum divisible by n.

Finally, on the subject of transversals in the Cayley table of \mathbb{Z}_n , we mention next a result about pairwise intersection of its transversals.

Theorem 1.69. Let $I(\mathbb{Z}_n)$ denote the set of integers i for which there exist transversals T and T' in the Cayley table of \mathbb{Z}_n such that $|T \cap T'| = i$. For each odd $n \neq 5$, $I(\mathbb{Z}_n) = \{i : 0 \leq i \leq n-3\} \cup \{n\}$, while $I(\mathbb{Z}_5) = \{0, 1, 5\}$.

Theorem 1.69 is due to Cavenagh and Wanless [36] who used it to prove Theorem 1.57. A key observation of [36], and one which exploits the equivalence of transversals in the Cayley table of \mathbb{Z}_n and diagonally cyclic latin squares, is the following.

Lemma 1.70. Let T and T' be two transversals in the Cayley table of \mathbb{Z}_n such that $T \cap T' = n - k$. Then there exists a k-homogeneous latin bitrade of size nk.

All of the general results mentioned in this section concern transversals in latin squares. For k > 1, there are no known general enumerative results for k-plexes. We will give a divisibility result for k-plexes in latin squares that are group tables in Section 4.5. One might hope that publication of the data in that section will inspire some serious results for plexes. We note that, for k = 1, proving the upper bound on T(n) involved considerable work [118]. Generalising results such as Theorems 1.60 and 1.63 to k-plexes may be difficult.

Chapter 2

A key lemma, definitions and latin families

2.1 Introduction

In this chapter we introduce a key lemma, along with the infinite families of latin squares and notation which we will use in our proofs in forthcoming chapters. The families have been designed for a productive application of Lemma 2.1. They are collected together here to facilitate comparison between their structure and to give a central reference location for related results and comments which cross several chapters. When working with a latin family defined in this chapter, we will also assume the named sets and notation conventions which are associated with the family. We suggest the reader should refer back to this chapter to obtain the definition of a latin family as and when it is first introduced in our workings. For example, only the families \mathcal{P}_n and \mathcal{Q}_n in Section 2.4.1 are needed for Chapter 3.

2.2 A necessary condition for plexes

Let G be an Abelian group and let L be a latin square of order |G| where $\mathcal{I}(L) = G$. The function $\Delta: L \to G$ is given by

$$\Delta(e) = z - x - y$$
 for each $e = (x, y, z) \in L$.

Lemma 2.1. Let G be an Abelian group with identity ε and let L be a latin square indexed by G. If K is a k-plex in L, then

$$\sum_{e \in K} \Delta(e) = -k \sum_{g \in G} g = \begin{cases} \omega & \text{if k is odd and G has a unique involution ω,} \\ \varepsilon & \text{otherwise.} \end{cases}$$

Proof. By definition, each row, column and symbol occurs k times in K, so that

$$\sum_{e \in K} \Delta(e) = \sum_{(x,y,z) \in K} \Delta(x,y,z) = k \sum_{z \in G} z - k \sum_{x \in G} x - k \sum_{y \in G} y = -k \sum_{g \in G} g.$$

If G has a unique involution ω , which happens precisely when G has non-trivial cyclic Sylow 2-subgroups, then $\sum_{g \in G} g = \omega$. Otherwise $\sum_{g \in G} g = \varepsilon$ (see, e.g. [46, p.34]). The result follows.

For several of the latin families to follow we will set $\mathcal{I}(L) = \mathbb{Z}_n$ so that, by Lemma 2.1,

$$\sum_{e \in K} \Delta(e) \equiv \begin{cases} \frac{1}{2} n \bmod n & \text{if } k \text{ is odd and } n \text{ is even,} \\ 0 & \text{otherwise.} \end{cases}$$

The simplest case of Lemma 2.1, as above when $G = \mathbb{Z}_n$, is due to Egan and Wanless [65, 67] and independently to Evans [73]. The case k = 1 was applied in [155] to prove Theorem 1.23.

Lemma 2.1 and variants of it have also proved useful in [19, 44, 73, 129].

2.3 Notation

In usage of the notation below we rely on context to make clear which latin square L we are considering.

Define r_x to be the set of elements in row x of L. That is,

$$r_x = \{(x, y, z) \in L : y, z \in \mathcal{I}(L)\}.$$
 (2.1)

To specify $X \subset L$ we identify, for each row of L, the columns in X, as follows.

$$\operatorname{col}(x) = \{ y \in \mathcal{I}(L) : (x, y, z) \in X \text{ for some } z \in \mathcal{I}(L) \}.$$
 (2.2)

As mentioned in Section 1.13, the latin families that we will use involve some latin trade with the Cayley table of a finite Abelian group G, with an identity element ε . The set Δ_* , as we next define, identifies the actual trade with the Cayley table of G. When applying Lemma 2.1 we focus on Δ_* , or perhaps Δ_i for some $i \in G$.

$$\Delta_* = \{ e \in L : \Delta(e) \neq \varepsilon \}, \qquad \Delta_i = \{ e \in \Delta_* : \Delta(e) = i \}. \tag{2.3}$$

2.4 Latin families of even order

For each member L of our families of even order we set $\mathcal{I}(L) = \mathbb{Z}_n$. We define L using notation L[x,y] = z, where all calculations are performed modulo n. We assume that each $i \in \mathcal{I}(L)$ is the least non-negative residue in its congruence class.

We partition $\mathcal{I}(L)$ by parity, as follows.

$$E = \{0, 2, 4, \dots, n-2\} \text{ and } F = \{1, 3, 5, \dots, n-1\}.$$
 (2.4)

2.4.1 \mathcal{P}_n , \mathcal{Q}_n and \mathcal{R}_n

Families \mathcal{P}_n and \mathcal{Q}_n originate in [65, 67]. They will be used in Chapters 3 and 4. The family \mathcal{R}_n appears in [19]. It is used in Chapter 4.

Definition 2.2 (Latin square \mathcal{P}_n). For n=4q=2h, define

$$\mathcal{P}_n[x,y] = \begin{cases} x+y+2 & \text{if } x=n-3 \text{ and } y \in F, \\ x+y-2 & \text{if } x=n-1 \text{ and } y \in F, \\ x+y & \text{otherwise.} \end{cases}$$
 (2.5)

The smallest member \mathcal{P}_4 is isotopic to the Cayley table of $\mathbb{Z}_2 \oplus \mathbb{Z}_2$.

In the case when $L = \mathcal{P}_n$, it is immediate from (2.5) that

$$\Delta_* = \Delta_{-2} \cup \Delta_2, \text{ where,}$$

$$\Delta_2 = \{(x, y, z) \in r_{n-3} : y \in F\},$$

$$\Delta_{-2} = \{(x, y, z) \in r_{n-1} : y \in F\}.$$

Less formally, construction of \mathcal{P}_n is evident by the table below of the value $\Delta(e)$ for each $e = (x, y, z) \in \Delta_*$.

Definition 2.3 (Latin square \mathcal{R}_n). For n = 4q = 2h, define

$$\mathcal{R}_n[x,y] = \begin{cases} x+y-4 & \text{if } x=3 \text{ and } y \in F, \\ x+y+4 & \text{if } x=n-1 \text{ and } y \in F, \\ x+y & \text{otherwise.} \end{cases}$$
 (2.7)

For $L = \mathcal{R}_n$,

if n = 4, $\Delta_* = \emptyset$ (since \mathcal{R}_4 is the Cayley table of \mathbb{Z}_4).

Otherwise,

$$\Delta_* = \Delta_{-4} \cup \Delta_4$$
, where,
 $\Delta_{-4} = \{(x, y, z) \in r_3 : y \in F\},$
 $\Delta_4 = \{(x, y, z) \in r_{n-1} : y \in F\}.$

For \mathcal{R}_n , the value $\Delta(e)$ for each $e \in \Delta_*$ is as follows.

Definition 2.4 (Latin square Q_n). For n=2h where $h \geqslant 3$ is odd, define

$$Q_{n}[x,y] = \begin{cases} n-1 & \text{if } x = y = 0 \text{ or } x = y = n-1, \\ 0 & \text{if } \{x,y\} = \{0,n-1\}, \\ x+y-2 & \text{if } x = 1 \text{ and } y \in F, \\ x+y+2 & \text{if } x = n-1 \text{ and } y \in F \setminus \{n-1\}, \\ x+y & \text{otherwise.} \end{cases}$$
(2.9)

If $L = \mathcal{Q}_n$, then

$$\Delta_* = \Delta_{-2} \cup \Delta_2 \cup I_{\mathcal{Q}} \text{ where } I_{\mathcal{Q}} = \Delta_{-1} \cup \Delta_1,
\Delta_{-2} = \{(x, y, z) \in r_1 : y \in F\},
\Delta_2 = \{(x, y, z) \in r_{n-1} : y \in F \setminus \{n - 1\}\},
\Delta_{-1} = \{(0, 0, n - 1)\},
\Delta_1 = \{(0, n - 1, 0), (n - 1, 0, 0), (n - 1, n - 1, n - 1)\}.$$
(2.10)

For \mathcal{Q}_n , the value $\Delta(e)$ for each $e \in \Delta_*$ is as follows.

Example 2.5. When we illustrate a member of \mathcal{P}_n , \mathcal{Q}_n or \mathcal{R}_n we adopt the convention of writing the rows and columns in the order $0, 2, 4, \ldots, n-2, 1, 3, 5, \ldots, n-1$. The marks $_*$ in (2.12) indicate elements of Δ_* in \mathcal{Q}_{14} . The circled entries identify $I_{\mathcal{Q}}$. The darker shaded cells show a 3-plex K and lighter shaded cells identify an 8-plex named J. It should be apparent that $\mathcal{Q}_{14} \setminus (K \cup J)$ is another 3-plex, so we have at hand a

 $(3^2, 8)$ -partition of this latin square.

The illustrated 3-plex K is given by

$$\begin{array}{lll} \operatorname{col}(0) &= \{0,2,9\}, & \operatorname{col}(1) &= \{1,3,5\}, \\ \operatorname{col}(2) &= \{2,6,11\}, & \operatorname{col}(3) &= \{0,12,5\}, \\ \operatorname{col}(4) &= \{4,8,13\}, & \operatorname{col}(5) &= \{0,2,7\}, \\ \operatorname{col}(6) &= \{6,8,1\}, & \operatorname{col}(7) &= \{4,9,13\}, \\ \operatorname{col}(8) &= \{10,3,7\}, & \operatorname{col}(9) &= \{4,1,11\}, \\ \operatorname{col}(10) &= \{10,5,9\}, & \operatorname{col}(11) &= \{6,3,13\}, \\ \operatorname{col}(12) &= \{12,7,11\}, & \operatorname{col}(13) &= \{8,10,12\}. \end{array}$$

In Chapter 3 we give a generalised definition of J.

2.4.2 $V_{n,j}$

The family $V_{n,j}$ is used in Chapter 4. These latin squares are turn-squares such as studied by Parker; see for example [118].

Definition 2.6 (Latin square $V_{n,j}$). For even $n \ge 6$ and $j \equiv 0 \mod 4$, let u = j/4, h = n/2 and $U = \{0, 1, 2, \dots, u - 1\}$. Define

$$\mathcal{V}_{n,j}[x,y] = \begin{cases} x+y+h & \text{if for some } v \in U, \\ (x,y) \in \{(0,2v), (0,h+2v), (h,2v), (h,h+2v)\}, \\ (x+y) & \text{otherwise.} \end{cases}$$
(2.14)

If $L = \mathcal{V}_{n,i}$, then

$$\Delta_* = \Delta_h \text{ and } |\Delta_*| = j. \tag{2.15}$$

2.4.3 \mathcal{U}_n

The family \mathcal{U}_n will be used in Chapter 5.

Definition 2.7 (Latin square \mathcal{U}_n). For n=2mq where q is odd, $q \ge 3$ and $m=2^t$ for $t \ge 1$. Let $M = \{m, 3m, 5m, \dots, n-m\}$. Define

$$\mathcal{U}_{n}[x,y] = \begin{cases}
n-m & \text{if } \{x,y\} = \{0, n - \frac{1}{2}m\} \text{ or } \{n-m\}, \\
n - \frac{1}{2}m & \text{if } \{x,y\} = \{0\} \text{ or } \{n - \frac{1}{2}m\}, \\
0 & \text{if } \{x,y\} = \{0, n-m\}, \\
y + mq - 2m & \text{if } q > 3, \ x = m \text{ and } y \in M, \\
y - m & \text{if } x = mq - 2m \text{ and } y \in M, \\
y + m & \text{if } x = n - m \text{ and } y \in M \setminus \{n - m\}, \\
x + y & \text{otherwise.}
\end{cases} \tag{2.16}$$

For $L = \mathcal{U}_n$,

$$\Delta_{*} = \Delta_{-\frac{1}{2}m} \cup \Delta_{\frac{1}{2}m} \cup \Delta_{m} \cup \Delta_{2m} \cup \Delta_{m(1-q)} \cup \Delta_{m(q-3)}, \text{ where,}$$

$$\Delta_{\frac{1}{2}m} = \left\{ (n - \frac{1}{2}m, n - \frac{1}{2}m, n - \frac{1}{2}m) \right\},$$

$$\Delta_{-\frac{1}{2}m} = \left\{ (0, 0, n - \frac{1}{2}m), (0, n - \frac{1}{2}m, n - m), (n - \frac{1}{2}m, 0, n - m) \right\},$$

$$\Delta_{m} = \left\{ (0, n - m, 0), (n - m, 0, 0), (n - m, n - m, n - m) \right\},$$

$$\Delta_{2m} = \begin{cases} \left\{ (x, y, z) \in r_{m} \cup r_{n-m} \setminus \Delta_{m} : y \in M \right\} & \text{if } q = 5, \\ \left\{ (x, y, z) \in r_{n-m} \setminus \Delta_{m} : y \in M \right\} & \text{otherwise,}
\end{cases}$$

$$\Delta_{m(1-q)} = \left\{ (x, y, z) \in r_{m(q-2)} : y \in M \right\},$$

$$\Delta_{m(q-3)} = \begin{cases} \emptyset & \text{if } q = 3, \\ \Delta_{2m} & \text{if } q = 5, \\ \left\{ (x, y, z) \in r_{m} : y \in M \right\} & \text{otherwise.}
\end{cases}$$

For \mathcal{U}_n , we table below the value $\Delta(e)$ for each $e \in \Delta_*$.

2.4.4 \mathcal{A}_n and \mathcal{B}_n

Families A_n and B_n will be used in Chapter 5.

Definition 2.8 (Latin square A_n). For n = 16d where $d \ge 1$, define

$$\mathcal{A}_{n}[x,y] = \begin{cases}
0 & \text{if } (x,y) \in \{(d,0), (0,15d)\}, \\
d & \text{if } (x,y) \in \{(4d,0), (d,13d)\}, \\
4d & \text{if } (x,y) \in \{(4d,10d), (10d,0)\}, \\
10d & \text{if } (x,y) \in \{(0,0), (10d,10d)\}, \\
14d & \text{if } (x,y) \in \{(0,10d), (d,14d), (4d,13d)\}, \\
15d & \text{if } (x,y) \in \{(0,14d), (d,15d)\}, \\
x+y & \text{otherwise.}
\end{cases} \tag{2.19}$$

For \mathcal{A}_n , the value of $\Delta(e)$ for each $e \in \Delta_*$ is tabled below.

Definition 2.9 (Latin square \mathcal{B}_n). For n=2h where $h \geq 5$ is an odd integer, define

$$\mathcal{B}_{n}[x,y] = \begin{cases} x+y+1 & \text{if } x=0 \text{ and } y \in \{n-2,n-1\}, \\ x+y-1 & \text{if } x=1 \text{ and } y \in \{0,n-2,n-1\}, \\ x+y+3 & \text{if } x=1 \text{ and } y=n-3, \\ x+y-3 & \text{if } x=4 \text{ and } y \in \{0,n-3\}, \\ x+y+4 & \text{if } x=0 \text{ and } y \in \{0,4,8,\ldots,n-6\}, \\ x+y-4 & \text{if } x=4 \text{ and } y \in \{4,8,\ldots,n-6\}, \\ x+y+h-7 & \text{if } n>14, \ x=6 \text{ and } y \in E, \\ x+y-h+7 & \text{if } n>14, \ x=h-1 \text{ and } y \in E \\ x+y & \text{otherwise.} \end{cases}$$
 (2.21)

For \mathcal{B}_n , there are 8 elements $e \in \Delta_*$ for which $\Delta(e)$ is an odd value:

For a even, $\Delta_a \subset \mathcal{B}_n$ is contained as follows:

If
$$n = 22$$
, $\Delta_4 \subset (r_0 \cup r_6)$ and $\Delta_{-4} \subset (r_4 \cup r_{h-1})$.
Otherwise, $\Delta_4 \subset r_0$ and $\Delta_{-4} \subset r_4$. (2.23)
If $n > 14$ and $n \neq 22$, $\Delta_{h-7} \subset r_6$ and $\Delta_{h+7} \subset r_{h-1}$.

2.5 Latin families of odd order

$\mathbf{2.5.1}$ \mathcal{H}_n

The family \mathcal{H}_n will be used in Chapter 4.

Definition 2.10 (Latin square \mathcal{H}_n). For odd $n \ge 5$, we define the latin square \mathcal{H}_n of order n and indexed by \mathbb{Z}_n .

$$\mathcal{H}_{n}[x,y] = \begin{cases} 0 & \text{if } (x,y) \in \{(1,0), (2,n-1)\}, \\ 1 & \text{if } (x,y) \in \{(0,0), (1,n-1)\}, \\ y+2 & \text{if } x=0 \text{ and } y \in \{1,3,5,\dots,n-2\}, \\ y & \text{if } x=2 \text{ and } y \in \{1,3,5,\dots,n-2\}, \\ x+y & \text{otherwise.} \end{cases}$$
(2.24)

For \mathcal{H}_n ,

$$\Delta_* = \Delta_1 \cup \Delta_{-1} \cup \Delta_2 \cup \Delta_{-2}, \text{ where,}$$

$$\Delta_1 = \{(0, 0, 1), (1, n - 1, 1)\},$$

$$\Delta_{-1} = \{(1, 0, 0), (2, n - 1, 0)\},$$

$$\Delta_2 = \{(x, y, z) \in r_0 : y \in \{1, 3, 5, \dots, n - 2\}\},$$

$$\Delta_{-2} = \{(x, y, z) \in r_2 : y \in \{1, 3, 5, \dots, n - 2\}\}.$$

For \mathcal{H}_n , we table below the value $\Delta(e)$ for each $e \in \Delta_*$.

$\mathbf{2.5.2}$ \mathcal{W}_n

The next family, which we name W_n , is due to Wanless and Webb [155]. It was used to prove Theorem 1.23 for all odd n > 3. We consider W_n in Chapter 5 and will give data for its small members in Chapters 4 and 5. For n = 5, W_n and H_n are of the same species.

Definition 2.11 (Latin square W_n). For odd $n \ge 5$, we define the latin square W_n of order n and indexed by \mathbb{Z}_n .

For $n \equiv 1 \mod 4$,

$$\mathcal{W}_{n}[x,y] = \begin{cases}
0 & \text{if } (x,y) \in \left\{ (0,1), \left(\frac{n-3}{2}, 0 \right), \left(n-1, \frac{n+3}{2} \right) \right\}, \\
1 & \text{if } (x,y) \in \left\{ (0,0), (n-1,1) \right\}, \\
x+2 & \text{if } y=0 \text{ and } x \in \left\{ 1,3,5,\dots, \frac{n-7}{2} \right\}, \\
x & \text{if } y=2 \text{ and } x \in \left\{ 1,3,5,\dots, \frac{n-7}{2} \right\} \cup \left\{ \frac{n-3}{2} \right\}, \\
\frac{n+1}{2} & \text{if } (x,y) \in \left\{ \left(\frac{n-3}{2}, \frac{n+3}{2} \right), (n-1,2) \right\}, \\
x+y & \text{otherwise.}
\end{cases} \tag{2.26}$$

For $n \equiv 3 \mod 4$,

$$\mathcal{W}_{n}[x,y] = \begin{cases}
0 & \text{if } (x,y) \in \{(0,1), (\frac{n-1}{2},0), (n-1,\frac{n+1}{2})\}, \\
1 & \text{if } (x,y) \in \{(0,0), (n-1,1)\}, \\
x+2 & \text{if } y = 0 \text{ and } x \in \{1,3,5,\dots,\frac{n-5}{2}\}, \\
x & \text{if } y = 2 \text{ and } x \in \{1,3,5,\dots,\frac{n-5}{2}\}, \\
\frac{n-1}{2} & \text{if } (x,y) \in \{(\frac{n-1}{2},\frac{n+1}{2}), (n-1,2)\}, \\
x+y & \text{otherwise.}
\end{cases} \tag{2.27}$$

For W_n , we table below the value $\Delta(e)$ for each $e \in \Delta_*$. The table on the left is for $n \equiv 1 \mod 4$. The table on the right is for $n \equiv 3 \mod 4$.

| | 0 | 1 2 | $\frac{n+3}{2}$ | | 0 | 1 | 2 | $\frac{n+1}{2}$ | |
|---------------------------------|-----------------|--------------------|-----------------|-----------------|-----------------|----|-----------------|-----------------|--------|
| 0 | 1 | -1 | | 0 | 1 | -1 | | | |
| 1 | 2 | -2 | | 1 | 2 | | -2 | | |
| 3 | 2 | -2 | | 3 | 2 | | -2 | | |
| 5 | 2 | -2 | | 5 | 2 | | -2 | | (2.28) |
| ÷ | : | : | | : | : | | : | | |
| $\frac{n-7}{2}$ | 2 | -2 | | $\frac{n-5}{2}$ | 2 | | -2 | | |
| $\frac{n-7}{2}$ $\frac{n-3}{2}$ | $\frac{n+3}{2}$ | -2 | $\frac{n+1}{2}$ | $\frac{n-1}{2}$ | $\frac{n+1}{2}$ | | | $\frac{n-1}{2}$ | |
| n-1 | | $1 \frac{n-1}{2}$ | $\frac{n-1}{2}$ | n-1 | | 1 | $\frac{n-3}{2}$ | $\frac{n+1}{2}$ | |

2.5.3 $\mathcal{O}_{m,k}$

The family $\mathcal{O}_{m,k}$ will be used in Chapter 4.

Definition 2.12 (Latin square $\mathcal{O}_{m,k}$). For odd integers, $m \geq 3$ and $k \geq 3$, we define $\mathcal{O}_{m,k}$ of order mk and indexed by $\mathbb{Z}_m \oplus \mathbb{Z}_k$. We test inequalities by assuming that c is the least non-negative residue in its congruence class.

$$\mathcal{O}_{m,k}[(a,b),(c,d)] =$$

$$\begin{cases} (a+c-1,\ b+d+1) & \text{if } a+c=0,\ b=0 \text{ and } c<\lfloor m/2\rfloor,\\ (a+c-1,\ b+d) & \text{if } a+c=0,\ b=0 \text{ and } c=\lfloor m/2\rfloor,\\ (a+c-1,\ b+d-1) & \text{if } a+c=0,\ b=0 \text{ and } c>\lfloor m/2\rfloor,\\ (a+c+1,\ b+d) & \text{if } a+c=m-1 \text{ and } b=0,\\ (a+c,\ b+d-1) & \text{if } a+c=m-1,\ b=1 \text{ and } c<\lfloor m/2\rfloor,\\ (a+c,\ b+d+1) & \text{if } a+c=m-1,\ b=k-1 \text{ and } c>\lfloor m/2\rfloor,\\ (a+c,\ b+d) & \text{otherwise.} \end{cases}$$

Example 2.13. For $\mathcal{O}_{3,3}$ we display in (2.29) the abbreviated ordered pairs $\Delta(e)$ for each $e \in \Delta_*$. For example, $e = ((0,0),(0,2),(2,0)) \in \mathcal{O}_{3,3}$, with $\Delta(e) = (2,1) \neq (0,0)$. Hence, Δ value 21 is at row 00, column 02. Shading indicates a 3-partition of $\mathcal{O}_{3,3}$ for which a definition is later given, by (4.16) in Chapter 4.

| | 00 | 01 | 02 | 10 | 11 | 12 | 20 | 21 | 22 |
|----|----|----|----|----|----|----|----|----|----|
| 00 | 21 | 21 | 21 | | | | 10 | 10 | 10 |
| 01 | | | | | | | | | |
| 02 | | | | | | | 01 | 01 | 01 |
| 10 | | | | 10 | 10 | 10 | 22 | 22 | 22 |
| 11 | | | | | | | | | |
| 12 | | | | | | | | | |
| 20 | 10 | 10 | 10 | 20 | 20 | 20 | | | |
| 21 | 02 | 02 | 02 | | | | | | |
| 22 | | | | | | | | | |

2.5.4 \mathcal{D}_{3m}

The family \mathcal{D}_{3m} will be used in Chapter 5. This construction is due to Ian Wanless but, as explained in Chapter 5, was motivated by joint data.

Definition 2.14 (Latin square \mathcal{D}_{3m}). For odd $m \geq 3$, we define the latin square \mathcal{D}_{3m} of order 3m and indexed by $\mathbb{Z}_3 \oplus \mathbb{Z}_m$.

$$\mathcal{D}[(a,b),(c,d)] =$$

$$\begin{cases} (1, d) & \text{if } (a = b = c = 0) \text{ or } (a = 2, b = 0 \text{ and } c = 1), \\ (0, d) & \text{if } (a = b = 0 \text{ and } c = 1) \text{ or } (a = c = 2 \text{ and } b = 0), \\ (0, d + 1) & \text{if } a = 1 \text{ and } b = c = 0, \\ (1, d + 1) & \text{if } a = 1, b = 0 \text{ and } c = 2, \\ (0, d) & \text{if } a = c = 0 \text{ and } b = 1, \\ (1, b + d + 1) & \text{if } a = c = 2 \text{ and } b \neq 0, \\ (a + c, b + d) & \text{otherwise.} \end{cases}$$

$$(2.30)$$

Let $\ell = m - 1$. For each $e \in \Delta_* \subset \mathcal{D}_{3m}$, we display below the abbreviated ordered pairs $\Delta(e)$. The shaded region shows an $(m-1) \times m$ subrectangle of \mathcal{D}_{3m} . In Chapter 5 we show that this subrectangle consists of transversal-free entries.

| | 00 | 01 | 02 | 0ℓ | 10 | 11 | 12 | 1ℓ | 20 | 21 | 22 | | 2ℓ | | |
|---------|---------|---------|---------|-------------|----|----|----|-------------|----|----|----|----|---------|----|--|
| 00 | 10 | 10 | 10 | 10 | 20 | 20 | 20 | 20 | | | | | | | |
| 01 | 0ℓ | 0ℓ | 0ℓ | 0ℓ | | | | | | | | | | | |
| 02 | | | | | | | | | | | | | | | |
| : | | | | | | | | | | | | | | | |
| 0ℓ | | | | | | | | | | | | | | | |
| 10 | 21 | 21 | 21 | 21 | | | | | 11 | 11 | 11 | | 11 | | |
| 11 | | | | | I | | | | | | | | | | |
| 12 | | | | | | | | | | | | | | (2 | |
| ÷ | | | | | | | | | | | | | | | |
| 1ℓ | | | | | | | | | | | | | | | |
| 20 | | | | | 10 | 10 | 10 | 10 | 20 | 20 | 20 | | 20 | | |
| 21 | | | | | | | | | 01 | 01 | 01 | | 01 | | |
| 22 | | | | | | | | | 01 | 01 | 01 | | 01 | | |
| i | | | | | | | | | • | : | : | ٠. | : | | |
| 2ℓ | | | | | | | | | 01 | 01 | 01 | | • | | |

Chapter 3

Latin squares with no small odd plexes

3.1 Introduction

The main results of this chapter appear in [67].

Latin squares with no small odd plexes were first shown in [65]:

Theorem 3.1 ([65]). For all even n > 2, there exists a latin square of order n which has no k-plex for any odd $k < \lfloor \frac{n}{4} \rfloor$, but does have a k-plex for all even k and some odd $k \leq \frac{1}{2}n$.

We will prove the following stronger statement.

Theorem 3.2. For all even n > 2, there exists a latin square of order n which has no k-plex for any odd $k < \lfloor \frac{n}{4} \rfloor$, but does have a k-plex for every other $k \leq \frac{1}{2}n$.

Theorem 3.2 will follow immediately from Theorems 3.15 and 3.18. In fact we will analyse the latin squares used to prove Theorem 3.2 in sufficient detail to describe all their possible types of partitions. We use the latin families \mathcal{P}_n and \mathcal{Q}_n , and associated notation from Chapter 2.

3.2 Necessary conditions

We begin by showing that \mathcal{P}_n and \mathcal{Q}_n have no small odd plexes.

Lemma 3.3 ([65]). Let k be an odd integer. If $k < \frac{1}{4}n$ or $k > \frac{3}{4}n$, then the latin square \mathcal{P}_n has no k-plex.

Proof. Assume that $K \subset \mathcal{P}_n$ is a k-plex for some odd k. By Lemma 2.1,

$$\sum_{e \in K \cap \Delta_*} \Delta(e) \equiv \frac{1}{2} n \bmod n. \tag{3.1}$$

Since K cannot satisfy (3.1) without inclusion of either $\frac{1}{4}n$ elements of Δ_{-2} or $\frac{1}{4}n$ elements of Δ_2 , there must be a row in K with at least $\frac{1}{4}n$ elements. As n is even, the complement of K is an (n-k)-plex with (n-k) odd. Hence $n-k \geqslant \frac{1}{4}n$, and so $k \leqslant \frac{3}{4}n$.

Corollary 3.4. If q is even, then a partition of the latin square \mathcal{P}_{4q} has at most two odd plexes.

Proof. Any partition has an even number of odd plexes, as 4q is even. Since q is even, Lemma 3.3 implies that an odd k-plex has $k \ge q + 1$. Thus, four or more odd plexes in a partition is impossible.

Lemma 3.5 ([65]). Let k be an odd integer. If $k < \frac{1}{4}(n-2)$ or $k > \frac{1}{4}(3n+2)$, then the latin square Q_n has no k-plex.

Proof. Assume that $K \subset \mathcal{Q}_n$ is a k-plex for some odd k. Then, by Lemma 2.1, (3.1) holds. Since $h = \frac{1}{2}n$ is odd, it is immediate that K contains an odd number of elements of $I_{\mathcal{Q}}$ as defined by (2.10). We consider two cases:

Case 1: $\sum_{e \in K \cap I_O} \Delta(e) = \pm 1$.

Then (3.1) requires that $|\Delta_{-2} \cap K| \geqslant \frac{1}{2}(h-1)$ or $|\Delta_2 \cap K| \geqslant \frac{1}{2}(h-1)$.

Case 2: $\sum_{e \in K \cap I_O} \Delta(e) = 3$.

Then (3.1) requires that either (i) $|\Delta_{-2} \cap K| \ge \frac{1}{2}(h+3)$, or (ii) $|\Delta_2 \cap K| \ge \frac{1}{2}(h-3)$. However, in subcase (ii), K also has two elements of $I_{\mathcal{Q}}$ in r_{n-1} .

Hence each possibility implies the existence of at least $\frac{1}{2}(h-1) = \frac{1}{4}(n-2)$ elements in the same row of K. As n is even, the complement of K is an (n-k)-plex with (n-k) odd. Thus $n-k \geqslant \frac{1}{4}(n-2)$, so $k \leqslant \frac{1}{4}(3n+2)$.

Plexes in squares of order six have been studied in detail. Our square \mathcal{Q}_6 belongs to species I in Fisher and Yates' catalogue [81]. Hence, due to Finney [78], we know that \mathcal{Q}_6 has a $(1^2, 2^2)$ -partition but no $(1^4, 2)$ -partition. Also, none of its eight distinct transversals intersect $I_{\mathcal{Q}} \cap r_5$. More generally:

Lemma 3.6. Let $h \equiv 3 \mod 4$ and $K \subset \mathcal{Q}_{2h}$ be a k-plex where $k = \frac{1}{2}(h-1)$. Then K contains precisely one element of $I_{\mathcal{Q}}$ and that element must be in r_0 .

Proof. It is clear from Lemma 3.5 that k is the minimal odd value for which a k-plex is possible. Assume that $|I_Q \cap r_{n-1} \cap K| > 0$. By Lemma 2.1 we have two cases:

Case 1: $\sum_{e \in K \cap I_{\mathcal{O}}} \Delta(e) = 3$.

Then $|I_{\mathcal{Q}} \cap r_{n-1} \cap K| = 2$ so Lemma 2.1 implies that either $|\Delta_{-2} \cap K| \ge \frac{1}{2}(h+3)$ or $|\Delta_2 \cap K| + |I_{\mathcal{Q}} \cap r_{n-1} \cap K| \ge \frac{1}{2}(h-3) + 2 = \frac{1}{2}(h+1)$.

Case 2: $\sum_{e \in K \cap I_Q} \Delta(e) = 1$.

Then $|I_{\mathcal{Q}} \cap r_{n-1} \cap K| \ge 1$ and Lemma 2.1 implies that either $|\Delta_{-2} \cap K| \ge \frac{1}{2}(h+1)$, or, $|\Delta_2 \cap K| + |I_{\mathcal{Q}} \cap r_{n-1} \cap K| \ge \frac{1}{2}(h-1) + 1 = \frac{1}{2}(h+1)$.

Hence in all cases, K has at least $\frac{1}{2}(h+1) > k$ elements in either r_1 or r_{n-1} which is a contradiction. Since $|I_{\mathcal{Q}} \cap K|$ is odd, K contains precisely one element of $I_{\mathcal{Q}}$ in r_0 . \square

Lemma 3.7. A partition of the latin square Q_{2h} has at most two odd plexes.

Proof. The latin square is of even order so the number of odd plexes in the partition is even. By Lemma 3.5, the number is not more than four. We now show that four is impossible. Assume there is a (k_1, k_2, \ldots, k_m) -partition where k_i is odd for i = 1, 2, 3, 4. We have two cases:

Case 1: $h \equiv 1 \mod 4$.

Since $\frac{1}{4}(n-2)$ is even, Lemma 3.5 implies that $k_i \ge \frac{1}{4}(n+2) > \frac{1}{4}n$ for i=1,2,3,4 which is impossible.

Case 2: $h \equiv 3 \mod 4$.

It follows from Lemma 3.6 that at most two of the k_i are minimal, say k_1 and k_2 . Therefore $k_3, k_4 \ge k_1 + 2 = \frac{1}{2}(h+3)$. However, then we have $\sum_{i=1}^4 k_i \ge n+2$ which is a contradiction.

3.3 Partitions of \mathcal{P}_n

In this section we show partitions of \mathcal{P}_n and determine its plex range. Throughout this section we assume that n = 2h = 4q for an integer $q \ge 1$.

We first show the even values in the plex range of \mathcal{P}_n .

Lemma 3.8 ([65]). The latin square \mathcal{P}_n has a 2-partition and hence has a k-plex for all even k in the interval $0 \leq k \leq n$.

Proof. For each $a \in E$, a 2-plex J_a is given by

$$\operatorname{col}(x) = \begin{cases} \{x+a, \ x+1+a\} & \text{if } x \in E, \\ \{x-3+a, \ x+a\} & \text{if } x \in F \text{ and } x < n-3, \\ \{x-3+a, \ x+2+a\} & \text{if } x = n-3, \\ \{x-3+a, \ x-2+a\} & \text{if } x = n-1. \end{cases}$$

Next is a partition with odd plexes.

Lemma 3.9 ([65]). The latin square \mathcal{P}_n has an (h-1,h+1)-partition.

Proof. An (h-1)-plex K is given by

$$\begin{aligned} & \operatorname{col}(n-2) = E \setminus \{h\}, \\ & \operatorname{col}(h-2) = F \setminus \{h+1\}, \\ & \operatorname{col}(1) = \{y \in E : \ y \geqslant h\} \cup \{y \in F : \ y < h-2\}, \\ & \operatorname{col}(n-1) = \{y \in E : \ y < h-2\} \cup \{y \in F : \ y > h\}. \end{aligned}$$

For $x \in \{0, 2, 4, \dots, h-4\}$,

$$col(n - 4 - x) = E \setminus \{n - 2 - x\},\$$

$$col(h - 4 - x) = F \setminus \{n - 1 - x\},\$$

$$col(h + 1 + x) = E \setminus \{x\},\$$

$$col(3 + x) = F \setminus \{1 + x\}.$$

It is routine to check that K is an (h-1)-plex.

The next corollary follows from Lemma 3.3.

Corollary 3.10 ([65]). For all $n \ge 8$, the latin square \mathcal{P}_n has a mixed plex range.

We point out that the latin square \mathcal{P}_4 is of the same species as the Cayley table of $\mathbb{Z}_2 \times \mathbb{Z}_2$. It has a complete plex range.

Our next two lemmas use partitions of the plexes constructed in Lemma 3.9 to create smaller plexes.

Lemma 3.11. The (h+1)-plex in \mathcal{P}_n defined in the proof of Lemma 3.9 contains a set of $\lfloor \frac{n}{8} \rfloor$ parallel 2-plexes.

Proof. Let $m = 4(\lfloor \frac{n}{8} \rfloor - 1)$. For each $a \in \{0, 4, 8, \dots, m\}$, a 2-plex J_a is given by

$$\operatorname{col}(x) = \begin{cases} \{n-4+x-a, \ n-2+x-a\} & \text{if } x \in E \text{ and } x < h, \\ \{1+x-a, \ 3+x-a\} & \text{if } x \in E \text{ and } x \geqslant h, \\ \{h-5+x-a, \ h-3+x-a\} & \text{if } x \in F \text{ and } x < h, \\ \{x-h-a, \ 2+x-h-a\} & \text{if } x \in F \text{ and } h < x < n-3, \\ \{2+x-h-a, \ 4+x-h-a\} & \text{if } x = n-3, \\ \{x-2-h-a, \ x-h-a\} & \text{if } x = n-1. \end{cases}$$

Lemma 3.12. If q is even, then the (h-1)-plex in \mathcal{P}_n defined in the proof of Lemma 3.9 contains a set of $\frac{n}{8}-1$ parallel 2-plexes.

Proof. Let $m=4\left(\frac{n}{8}-2\right)=h-8$. For each $a\in\{0,4,8,\ldots,m\}$ we show a 2-plex J_a :

For $x \in \{0, 4, 8, \dots, h - 4\},\$

$$\operatorname{col}(x) = \{1 + x - a, \ 3 + x - a\},\$$

$$\operatorname{col}(2 + x) = \{h + 1 + x - a, \ h + 3 + x - a\},\$$

$$\operatorname{col}(h + x) = \{x - a, \ 2 + x - a\},\$$

$$\operatorname{col}(h + 2 + x) = \{h + x - a, \ h + 2 + x - a\},\$$

$$\operatorname{col}(1 + x) = \{h - 5 + x - a, \ h - 3 + x - a\},\$$

$$\operatorname{col}(3 + x) = \{n - 5 + x - a, \ n - 3 + x - a\},\$$

$$\operatorname{col}(h + 1 + x) = \{h - 4 + x - a, \ h - 2 + x - a\},\$$

$$\operatorname{col}(h + 3 + x) = \{n - 4 + x - a, \ n - 2 + x - a\}.$$

We can now say which odd plexes exist in \mathcal{P}_{4q} when q is even.

Lemma 3.13. If q is even, then the latin square \mathcal{P}_{4q} has a $\left(2^{q-1}, (q+1)^2\right)$ -partition and hence has a k-plex for all odd k in the interval $\frac{1}{4}n+1 \leqslant k \leqslant \frac{3}{4}n-1$.

Proof. We have the following refinements of the (h-1,h+1)-partition of Lemma 3.9. Lemma 3.11 gives a division of the (h+1)-plex into $\frac{n}{8}=\frac{1}{2}q$ parallel 2-plexes and a parallel (odd) (q+1)-plex. Lemma 3.12 gives a division of the (h-1)-plex into $\frac{n}{8}-1=\frac{1}{2}q-1$ parallel 2-plexes and a parallel (q+1)-plex. Thus we have the claimed partition. The union of a (q+1)-plex with as many of the 2-plexes as required forms a k-plex, for each odd k in the interval $\frac{1}{4}n+1\leqslant k\leqslant \frac{3}{4}n-1$.

In (3.2) we illustrate the $(2^3, 5^2)$ -partition of \mathcal{P}_{16} in Lemma 3.13. The circled elements show the 2-plex given by Lemma 3.12. It is contained within the (dark) 7-plex defined in the proof of Lemma 3.9. The two 2-plexes from Lemma 3.11 are shown by paler

shading. The marks * identify elements of Δ_* .

| | 0 | 2 | 4 | 6 | 8 | 10 | 12 | 14 | 1 | 3 | 5 | 7 | 9 | 11 | 13 | 15 | |
|----|----|----|----|----|-----|----|------|------|----------|-----|-------|-------|------|-----|-----|-----|-------|
| 0 | 0 | 2 | 4 | 6 | 8 | 10 | 12 | 14 | 1 | 3 | 5 | 7 | 9 | 11 | 13 | 15 | |
| 2 | 2 | 4 | 6 | 8 | 10 | 12 | 14 | 0 | 3 | 5 | 7 | 9 | 11 | 13 | 15 | 1 | |
| 4 | 4 | 6 | 8 | 10 | 12 | 14 | 0 | 2 | 5 | 7 | 9 | (11) | 13 | 15 | 1 | 3 | |
| 6 | 6 | 8 | 10 | 12 | 14 | 0 | 2 | 4 | 7 | 9 | 11 | 13 | 15 | 1 | 3 | (5) | |
| 8 | 8 | 10 | 12 | 14 | 0 | 2 | 4 | 6 | 9 | 11 | 13 | 15 | 1 | 3 | 5 | 7 | |
| 10 | 10 | 12 | 14 | 0 | 2 | 4 | 6 | 8 | 11 | 13 | 15 | 1 | 3 | 5 | 7 | 9 | |
| 12 | 12 | 14 | 0 | 2 | 4 | 6 | 8 | 10 | 13 | 15 | 1 | 3 | 5 | 7 | 9 | 11 | |
| 14 | 14 | 0 | 2 | 4 | 6 | 8 | (10) | (12) | 15 | 1 | 3 | 5 | 7 | 9 | 11 | 13 | (3.2) |
| | | | | | | | | | | | | ı | | | | | (9.2) |
| 1 | 1 | 3 | 5 | 7 | 9 | 11 | 13 | 15 | 2 | (4) | (6) | 8 | 10 | 12 | 14 | 0 | |
| 3 | 3 | 5 | 7 | 9 | 11 | 13 | 15 | 1 | 4 | 6 | 8 | 10 | 12 | 14 | 0 | 2 | |
| 5 | 5 | 7 | 9 | 11 | 13 | 15 | 1 | 3 | 6 | 8 | 10 | (12) | (14) | 0 | 2 | 4 | |
| 7 | 7 | 9 | 11 | 13 | 15 | 1 | 3 | 5 | 8 | 10 | 12 | 14 | 0 | 2 | 4 | 6 | |
| 9 | 9 | 11 | 13 | 15 | 1 | 3 | 5 | 7 | 10 | 12 | 14 | 0 | 2 | 4 | 6 | 8 | |
| 11 | 11 | 13 | 15 | 1 | 3 | 5 | 7 | 9 | 12 | 14 | 0 | 2 | 4 | 6 | 8 | 10 | |
| 13 | 13 | 15 | 1 | 3 | (5) | 7 | 9 | 11 | 0* | 2* | 4_* | 6* | 8* | 10* | 12* | 14* | |
| 15 | 15 | 1 | 3 | 5 | 7 | 9 | 11 | 13 | 14_{*} | 0* | 2* | 4_* | 6* | 8* | 10* | 12* | |

We have seen that when q is even a partition of \mathcal{P}_{4q} has at most two odd plexes. In contrast, when q is odd \mathcal{P}_{4q} has a partition containing four odd plexes.

Lemma 3.14. If q is odd, then the latin square \mathcal{P}_n has an h-partition in which each h-plex can be divided in the following ways:

- 1. into q parallel 2-plexes,
- 2. into two parallel q-plexes.

Thus, when q is odd, \mathcal{P}_{4q} has a q-partition and a $(2^q, q^2)$ -partition. Hence, when q is odd, \mathcal{P}_n has a k-plex for all odd k in the interval $\frac{1}{4}n \leq k \leq \frac{3}{4}n$.

Proof. We first show two parallel h-plexes, H and H', each of which is formed by the union of q parallel 2-plexes. We will then divide each of H and H' into two parallel q-plexes.

We partition the sets E and F as follows:

```
E_0 = \{i \in E : i \equiv 0 \bmod 4\}, \quad E_2 = \{i \in E : i \equiv 2 \bmod 4\},
F_1 = \{i \in F : i \equiv 1 \bmod 4\}, \quad F_3 = \{i \in F : i \equiv 3 \bmod 4\}.
```

We use the parallel 2-plexes J_a as defined in the proof of Lemma 3.8. Let $H = \bigcup_{a \in E_0} J_a$ and $H' = \bigcup_{a \in E_2} J_a$. Now H and H' each contain q parallel 2-plexes and together they form an h-partition.

We need to show that H and H' each contain a q-plex.

A q-plex $K_0 \subset H$ is given by

$$\operatorname{col}(x) = \begin{cases} E_0 & \text{if } (x \in E_0 \text{ and } x \geqslant h-2) \text{ or } (x \in F_3 \text{ and } x < h), \\ E_2 & \text{if } (x \in E_2 \text{ and } x < h) \text{ or } (x \in F_1 \text{ and } x \geqslant h-1), \\ F_1 & \text{if } (x \in E_0 \cup F_1 \text{ and } x < h-2) \text{ or } x = n-1, \\ F_3 & \text{if } (x \in E_2 \text{ and } x \geqslant h) \text{ or } (x \in F_3 \text{ and } h < x < n-1). \end{cases}$$

A q-plex $K_2 \subset H'$ is given by

$$\operatorname{col}(x) = \begin{cases} E_2 & \text{if } (x \in E_0 \text{ and } x \geqslant h-2) \text{ or } (x \in F_3 \text{ and } x < h), \\ E_0 & \text{if } (x \in E_2 \text{ and } x < h) \text{ or } (x \in F_1 \text{ and } x \geqslant h-1), \\ F_3 & \text{if } (x \in E_0 \cup F_1 \text{ and } x < h-2) \text{ or } x = n-1, \\ F_1 & \text{if } (x \in E_2 \text{ and } x \geqslant h) \text{ or } (x \in F_3 \text{ and } h < x < n-1). \end{cases}$$

It is routine to check that each of the above q-plexes is contained in its respective h-plex. It follows that $K_1 = H \setminus K_0$ and $K_3 = H' \setminus K_2$ are also q-plexes so we have divided the h-partition into a q-partition.

Dividing H in one way, and H' in the other, gives a $(2^q, q^2)$ -partition. Using this partition, the union of one q-plex with as many of the 2-plexes as required yields a k-plex, for each odd k in the interval $\frac{1}{4}n \leq k \leq \frac{3}{4}n$.

Combining the above results, we can now establish exactly which partitions of \mathcal{P}_n are possible.

Theorem 3.15. The plex range of the latin square \mathcal{P}_n is the set

$$\left\{\frac{1}{4}n \leqslant k \leqslant \frac{3}{4}n : k \text{ is odd}\right\} \cup \left\{0, 2, 4, \dots, n\right\}.$$

Moreover, \mathcal{P}_n possesses a partition of every type consistent with its plex range.

Proof. Any type of partition which has no odd plexes can be obtained by the union of plexes from the 2-partition of Lemma 3.8.

If $n \equiv 0 \mod 8$, then a (q+1)-plex is the smallest odd plex permitted by Lemma 3.3. A partition with odd plexes has exactly two odd plexes, by Lemma 3.4, so it is of a type which can be generated by the $(2^{q-1}, (q+1)^2)$ -partition of Lemma 3.13.

In the case that $n \equiv 4 \mod 8$, a partition with four odd plexes is a q-partition. A partition with precisely two odd plexes can, for all possible partition types, be formed

by the union of plexes from a $(2^q, q^2)$ -partition. Thus the partitions of Lemma 3.14 generate every possible partition type for odd plexes in this case.

3.4 Partitions of Q_n

In this section we investigate partitions of Q_n and determine its plex range. We assume throughout this section that n = 2h for an odd integer $h \ge 3$.

Lemma 3.16 ([65]). The latin square Q_n has a 2-partition and hence has a k-plex for all even k in the interval $0 \le k \le n$.

Proof. For each $a \in E$, a 2-plex J_a is given by

$$\operatorname{col}(x) = \begin{cases} \{x + a, \ x - 1 + a\} & \text{if } x \in E, \\ \{x + 1 + a, \ x - 2 + a\} & \text{if } x \in F \setminus \{1, n - 1\}, \\ \{x + 1 + a, \ x - 4 + a\} & \text{if } x = 1, \\ \{x + 1 + a, \ x + a\} & \text{if } x = n - 1. \end{cases}$$

Notice that the 2-plex J_0 defined in the proof of Lemma 3.16 contains all of I_Q . Thus, it follows from Lemma 2.1 that $Q_n \setminus J_0$ is an (n-2)-plex which contains no odd plexes. We now show that Q_n itself contains many odd plexes. In the next lemma, the specified value of k is the smallest odd value that is not prohibited by Lemma 3.5.

Lemma 3.17. The latin square Q_{2h} has a $(k^2, 2^{h-k})$ -partition where $k = 2\lfloor \frac{h}{4} \rfloor + 1$.

Proof. We will define an even plex J which is the union of h-k parallel 2-plexes. Then, we will show an odd k-plex K, contained in the 2k-plex $\mathcal{Q}_{2h}\setminus J$, thus collectively giving a $(k^2, 2^{h-k})$ -partition.

Let m = 2(h - k - 1). For each $a \in \{0, 2, 4, \dots, m\}$, a 2-plex J_a is given by

$$\operatorname{col}(x) = \begin{cases} \{x+1+a, \ x-2-a\} & \text{if } x \in E \text{ or } (x \in F \setminus \{1, n-1\} \text{ and } a < m), \\ \{x+1+a, \ x-4-a\} & \text{if } x = 1 \text{ and } a < m, \\ \{x+1+a, \ x-a\} & \text{if } x = n-1 \text{ and } a < m, \\ \{x+1+a, \ n-1\} & \text{if } x = 1 \text{ and } a = m, \\ \{x+1+a, \ x\} & \text{if } x \in F \setminus \{1, n-1\} \text{ and } a = m, \\ \{x+1+a, \ 1\} & \text{if } x = n-1 \text{ and } a = m. \end{cases}$$

Then $J = \bigcup_{a \in \{0,2,4,\ldots,m\}} J_a$. (An illustration of $J \subset \mathcal{Q}_{14}$ was given in Example 2.5.)

Next, we show the odd k-plex K. Let $t = \lfloor \frac{h}{4} \rfloor$, and let

$$T_x = \{x + 2, x + 4, x + 6, \dots, x + 2t\},\$$

$$U_x = \{n - 1 - 4t + x, n + 1 - 4t + x, n + 3 - 4t + x, \dots, n - 3 - 2t + x\},\$$

$$V_x = \{h + x + 4, h + x + 8, h + x + 12, \dots, h + x + 4t\},\$$

$$W_x = \{x + 2, x + 6, x + 10, \dots, x - 2 + 4t\}.$$

We specify K in two cases.

Case 1: $h \equiv 1 \mod 4$

For $x \in E$,

$$\operatorname{col}(x) = \begin{cases} T_x \cup \{x\} \cup U_x & \text{if } x \leqslant 2t \text{ or } x \geqslant h + 2t + 1, \\ T_x \cup \{x - 1\} \cup U_x & \text{otherwise,} \end{cases}$$

and for $x \in F$,

$$\operatorname{col}(x) = \begin{cases} \{0\} \cup \{1, 3, 5, \dots, h - 2\} & \text{if } x = 1, \\ \{h - 1, h + 1, h + 3, \dots, n - 2\} & \text{if } x = n - 1, \\ V_x \cup W_x \cup \{x + 4t\} & \text{if } h - 2t \leqslant x \leqslant h + 2t, \\ \{h + x\} \cup V_x \cup W_x & \text{otherwise.} \end{cases}$$

Case 2: $h \equiv 3 \mod 4$

In the case h=7, we rely on (2.13). For $h\neq 7$, we define K as follows.

For $x \in E$,

$$\operatorname{col}(x) = \begin{cases} T_x \cup \{x\} \cup U_x & \text{if } x \leqslant h - 1, \\ T_x \cup \{x - 1\} \cup U_x & \text{otherwise,} \end{cases}$$

and for $x \in F$,

$$\operatorname{col}(x) = \begin{cases} \{1, 3, 5, \dots, h-2\} & \text{if } x = 1, \\ \{h+x+2\} \cup V_x \cup W_x & \text{if } 3 \leqslant x \leqslant h-4, \\ V_x \cup \{h+x+10\} \cup W_x & \text{if } x = h-2, \\ \left(V_x \setminus \{h+x+8\}\right) \cup \{h+x+10, x+2+4t\} \cup W_x & \text{if } h \leqslant x \leqslant n-7, \\ V_x \cup W_x \cup \{x+2+4t\} & \text{if } n-5 \leqslant x \leqslant n-3, \\ \{h+1, h+3, h+5, \dots, n-2\} & \text{if } x = n-1. \end{cases}$$

Combining the above results, we can now establish exactly which partitions of Q_n are possible.

Theorem 3.18. The plex range of the latin square Q_n is the set

$$\left\{\frac{1}{4}(n-2) \leqslant k \leqslant \frac{1}{4}(3n+2) : k \text{ is odd}\right\} \cup \{0, 2, 4, \dots, n\}.$$

Moreover, Q_n possesses a partition of every type consistent with its plex range.

Proof. The 2-partition of Lemma 3.16 generates an example of a partition of any type that has no odd plex. A partition with odd plexes has exactly two odd plexes by Lemma 3.7. Therefore, we can obtain a partition of any type with odd plexes (except those ruled out by Lemma 3.5) by using the partition described in Lemma 3.17. \square

Corollary 3.19 ([65]). If $n \ge 10$, then the latin square Q_n has a mixed plex range.

The latin square illustrated in (1.1) is Q_6 . It has a complete plex range.

3.5 Concluding remarks

8

9

Theorem 3.2 shows existence of latin squares of even order with a plex range which is quite distinct from those which had previously been described, either in general or from small order studies. We summarise in Table 3.1 the latin squares of order $n \leq 9$ by the named plex ranges. The data for the table for orders $n \leq 8$ is due to the work of several authors: Finney [76–78], Freeman [84–86], Johnston and Fullerton [100], Killgrove et al. [106] and Wanless [150]. For order 9, the data is shown by our work in Chapter 5.

| Order n | Plex | range | Number of species | |
|-----------|----------|-------|-------------------|-----|
| | Complete | Even | Mixed | |
| 1 | 1 | | | 1 |
| 2 | | 1 | | 1 |
| 3 | 1 | | | 1 |
| 4 | 1 | 1 | | 2 |
| 5 | 2 | | | 2 |
| 6 | 6 | 4 | 2 | 12 |
| 7 | 147 | | | 147 |

1

32

283657

19270853541

283624

Table 3.1: Species of order $n \leq 9$ according to plex range.

The 34 species, of order $n \leq 9$ with a mixed plex range, consist of latin squares with a 3-plex but no transversal. Conjecture 1.4 says that a latin square with this property exists for every even order n > 4, and while that particular problem remains open, we are now seeing examples, such as \mathcal{P}_{16} and \mathcal{Q}_{18} , which have neither a 3-plex nor a 1-plex, but do have a 5-plex. Wanless [150] observes that the only latin squares of order $n \leq 8$, without the maximum number of parallel 3-plexes, are the step-type latin squares of Theorem 1.25. He asks whether or not other larger latin squares can fail to possess a 3-plex. A question raised by Pula [129] is the following.

Problem 3.20. Do there exist latin squares outside of those explained by Theorem 1.25 which fail to contain any odd plexes?

The example in (1.2) is a latin square of order 6 which fails the HP-condition, as described in Section 1.10, but it does possess a 3-plex. The latin squares \mathcal{P}_n and \mathcal{Q}_n offer further examples of latin squares which fail the same condition. For $n \geq 8$, the families \mathcal{P}_n and \mathcal{Q}_n satisfy condition (B^*) but fail to contain transversals.

The construction of \mathcal{P}_n and \mathcal{Q}_n could conceivably be adapted to produce examples of latin squares with other plex ranges. For example, the family \mathcal{U}_n which we use in Chapter 5 has some similarity in its construction with \mathcal{Q}_n , but \mathcal{U}_n does possess transversals. On the other hand, it is not obvious how an argument relying on Lemma 2.1 might show a latin square in which the smallest odd k-plex is larger than that shown by \mathcal{P}_n and \mathcal{Q}_n .

Problem 3.21. What is the maximum value of n/k such that there exists a latin square L of order n that possesses an odd k-plex but L has no c-plex for any odd c < k?

Whenever k is the smallest odd value in the plex range of a latin square of even order, every k-plex in the latin square is indivisible. Theorems 3.15 and 3.18 prove the next statement which answers a question raised in [150].

Theorem 3.22. For all odd k > 1 and each $n \in \{4k - 4, 4k - 2, 4k, 4k + 2\}$, there exists a latin square of order n that has a k-plex but every k-plex is indivisible.

Problem 3.23. For k > 2, does there exist a latin square L that possesses a k-plex and every k-plex in L is indivisible but k is not the least odd value in the plex range of L?

Conjecture 1.3 implies that an example, if it exists, is of even order and k is odd.

It would be interesting to know if a latin square can have a plex range substantially different from examples seen thus far.

Problem 3.24. Does there exist a latin square which has an a-plex and a c-plex but does not have a b-plex for odd integers a < b < c?

An analogous problem, concerning k-factors of graphs, is answered in the negative by Katerinis [102].

The latin squares \mathcal{P}_n and \mathcal{Q}_n contain relatively large indivisible k-plexes, in the sense that k grows linearly with the order of the square. In Section 4.2.1 we will show indivisible $(\frac{1}{2}n)$ -partitions in these families. Theorems 3.15 and 3.18 were shown by identifying partitions which, with a small case exception, are indivisible partitions. In the spirit of the next chapter, we record the following. Theorem 3.25 follows immediately from the plex range of \mathcal{P}_{4q} .

Theorem 3.25. The following partitions of \mathcal{P}_{4q} are indivisible:

- 1. For q > 1, the 2-partition of Lemma 3.8.
- 2. For odd q > 1, the 2-partition of Lemma 3.14.
- 3. For odd q > 1, the $(2^q, q^2)$ -partition of Lemma 3.14.
- 4. For odd q, the q-partition of Lemma 3.14.
- 5. For even q, the $(2^{q-1}, (q+1)^2)$ -partition of Lemma 3.13.

The constraint q > 1 is necessary for Parts 1, 2 and 3 of Theorem 3.25. In Section 4.5 we see that \mathcal{P}_4 (isotopic to $\mathbb{Z}_2 \oplus \mathbb{Z}_2$) possesses no indivisible 2-plex.

Theorem 3.26. The following partitions of Q_{2h} are indivisible:

- 1. The 2-partition of Lemma 3.16.
- 2. For $k = 2\lfloor \frac{h}{4} \rfloor + 1$, the $(k^2, 2^{h-k})$ -partition of Lemma 3.17.

Justification of the theorem for Q_6 , which has transversals, follows from the observation immediately after the proof of Lemma 3.16. Otherwise, the theorem follows immediately from the plex range of Q_{2h} .

Chapter 4

Indivisible plexes and partitions

4.1 Introduction

Motivated by Problem 1.5, we continue to investigate indivisible partitions of the latin families used in Chapter 3. In Section 4.2 we will show that, for all even n > 2, there exists a latin square of order n that can be decomposed into two parallel indivisible $(\frac{1}{2}n)$ -plexes. We then investigate ways of embedding such plexes into larger latin squares.

In Section 4.3 we find indivisible partitions, by direct construction, in latin squares of odd order. Collecting our results for even and odd orders together, we prove our two main theorems, the first of which addresses Problem 1.22.

Theorem 4.1. If k divides n where n > 2, k < n and $(n, k) \neq (6, 1)$, then there exists a latin square of order n with an indivisible k-partition.

Theorem 4.2 addresses Problem 1.5.

Theorem 4.2. For all n > 2, there exists a latin square of order n with two parallel indivisible $\lfloor \frac{n}{2} \rfloor$ -plexes. Hence, for all n > 2, $\kappa(n) \geqslant \lfloor \frac{n}{2} \rfloor$.

In Section 4.5, we report on computations of $\kappa(n)$ and indivisible partitions and plexes in latin squares of small order. Beforehand, in Section 4.4, we offer explanation for the absence of odd plexes in some examples of interest, including the largest indivisible plex that we found for order 9.

4.2 Latin squares of even order

The results in this section appear in [19].

4.2.1 Indivisible partitions of \mathcal{P}_{2h} , \mathcal{Q}_{2h} and \mathcal{R}_{2h}

In this section we prove the following theorem.

Theorem 4.3. For all integers $k \ge 2$, there exists a latin square of order 2k which contains two parallel indivisible k-plexes.

Theorem 4.3 will follow immediately from Lemmas 4.4, 4.5 and 4.8.

We prove Theorem 4.3 in three cases. In each case, we identify an h-partition of an appropriate latin square and then prove that each of its two parts is indivisible.

Throughout, assume that $h = \frac{1}{2}n$ where n is the order of the latin square in context; either \mathcal{P}_n , \mathcal{Q}_n or \mathcal{R}_n .

For our work in this section we introduce the following notation. Let K be a k-plex in L and let $X \subseteq L$. Define

$$X^K = |X \cap K|.$$

Lemma 4.4. For all odd q, the latin square \mathcal{R}_{4q} has an indivisible h-partition.

Proof. It is easy to check that \mathcal{R}_4 has an indivisible 2-partition. For q > 1, an h-plex $H \subset \mathcal{R}_{4q}$ is given by

$$\operatorname{col}(x) = \begin{cases} E & \text{if } x < h, \\ F & \text{otherwise.} \end{cases}$$
 (4.1)

Note that since q is odd, $h \equiv 2 \mod 4$. We now show that H is indivisible. Since $\Delta_4^H = h$ and $\Delta_{-4}^H = 0$, a j-plex J contained in H has $j = \Delta_4^J = \Delta_*^J$. If j is odd, then Lemma 2.1 says that $4j \equiv h \mod 4q$. However $4j \equiv 0 \mod 4$ and $h \equiv 2 \mod 4$, so we can conclude that j is not odd. On the other hand, if j is even then Lemma 2.1 says that $4j \equiv 0 \mod 4q$ and thus, since $j \neq q$, we have $j \in \{0, h\}$. Hence H is indivisible. Similarly, $H' = \mathcal{R}_{4q} \setminus H$ is indivisible.

The previous lemma proves Theorem 4.3 when $k \equiv 2 \mod 4$. The next lemma proves the case $k \equiv 0 \mod 4$.

Lemma 4.5. For all even q, the latin square \mathcal{P}_{4q} has an indivisible h-partition.

Proof. An h-plex $H \subset \mathcal{P}_{4q}$ is given by

$$\operatorname{col}(x) = \begin{cases} E & \text{if } (x \in E \text{ and } x < h) \text{ or } (x \in F \text{ and } h - 1 \leqslant x < n - 1), \\ F & \text{otherwise.} \end{cases}$$
(4.2)

Note that since q is even, $h \equiv 0 \mod 4$. We have $\Delta_{-2}^H = h$ and $\Delta_2^H = 0$. We rely on Lemma 2.1 and follow the process of the proof of Lemma 4.4.

As an aside to our proof of Theorem 4.3, note that (4.2) defines an h-partition of \mathcal{P}_{4q} regardless of the parity of q. If q is odd, then the h-partition consists of plexes which contain no even plex, but do, as easily verified, contain q-plexes. On the other hand, a slightly different h-plex of \mathcal{P}_{4q} , given by (4.3) below, puts all of Δ_* into one plex. The result is an h-partition in which both parts contain only even plexes.

$$col(x) = \begin{cases} E & \text{if } x < h, \\ F & \text{otherwise.} \end{cases}$$
 (4.3)

We do not know if \mathcal{P}_{4q} has an indivisible h-partition when q is odd, although we did find h-partitions in which one part is indivisible.

We now return to the proof of Theorem 4.3. In the third and final case we rely on conditions other than Lemma 2.1.

The structure of the latin square Q_n of order n = 2h may be thought of as an h-step type latin square (see Section 1.9) which has been *corrupted* by one element in each block. More specifically,

$$\mathcal{Q}_n = \left(\begin{array}{cc} B_{00} & B_{01} \\ B_{10} & B_{11} \end{array}\right),\,$$

We now consider the number of elements in a k-plex found in the $h \times h$ blocks of a c-corrupted h-step type latin square.

Lemma 4.6 ([65]). Suppose that L is a c-corrupted h-step type latin square of order n = 2h where $L = \begin{pmatrix} B_{00} & B_{01} \\ B_{10} & B_{11} \end{pmatrix}$ and c entries of B_{00} have symbols from F, and the remaining $h^2 - c$ entries of B_{00} have symbols from E. If $K \subseteq L$ is a k-plex, then the following conditions are satisfied.

- 1. $B_{00}^K = B_{11}^K$, and $B_{01}^K = B_{10}^K$.
- 2. $-2c \leqslant B_{00}^K B_{01}^K \leqslant 2c$.
- 3. If both k and h are odd, then $-2c < B_{00}^K B_{01}^K < 2c$.

Proof. Each of the h rows of $B_{00} \cup B_{01}$ and each of the h columns of $B_{01} \cup B_{11}$ is represented k times in K. Therefore $B_{00}^K + B_{01}^K = kh = B_{01}^K + B_{11}^K$, hence $B_{00}^K = B_{11}^K$. Similarly $B_{01}^K = B_{10}^K$, thus part 1 of the lemma is shown.

Next we partition each B_{ij} into A_{ij} and C_{ij} such that for all $e = (x, y, z) \in B_{ij}$,

$$e \in A_{ij}$$
 if $(z \in F \text{ and } i \neq j)$ or $(z \in E \text{ and } i = j)$,
 $e \in C_{ij}$ otherwise. (4.4)

Consider part 2 of the lemma. Each symbol in E is represented k times in K so $A_{00}^K + A_{11}^K + C_{01}^K + C_{10}^K = k|E| = kh$. Hence, rearranging terms;

$$A_{00}^K + A_{11}^K = kh - \left(C_{01}^K + C_{10}^K\right). \tag{4.5}$$

Similarly, counting symbols of F in K;

$$A_{01}^K + A_{10}^K = kh - \left(C_{00}^K + C_{11}^K\right). \tag{4.6}$$

Now assume that $B_{00}^K - B_{01}^K > 2c$. Then $B_{11}^K - B_{10}^K > 2c$. Summing these two inequalities and substituting $B_{ij}^K = A_{ij}^K + C_{ij}^K$ gives

$$\left(A_{00}^K + A_{11}^K + C_{00}^K + C_{11}^K\right) - \left(A_{01}^K + A_{10}^K + C_{01}^K + C_{10}^K\right) > 4c. \tag{4.7}$$

Substituting (4.5) and (4.6) into (4.7) we obtain that $C_{00}^K + C_{11}^K - C_{01}^K - C_{10}^K > 2c$, which is a contradiction since $C_{00}^K \leqslant c$ and $C_{11}^K \leqslant c$. It follows, by symmetry, that $B_{00}^K - B_{01}^K < -2c$ must also be false. Thus part 2 of the lemma has been shown.

Now if k and h are both odd, then $B_{00}^K - B_{01}^K = B_{00}^K - (kh - B_{00}^K) = 2B_{00}^K - kh$ is odd, so part 3 of the lemma follows.

Example 4.7 ([65]). The shading in (4.8) identifies a 7-plex, which we call H, in \mathcal{Q}_{14} . Observe that $\Delta_{-2}^H = 0$, $\Delta_2^H = 3$ and $I_{\mathcal{Q}}^H = 1$. Entries marked $_*$ are elements of Δ_* . The circled entries show $I_{\mathcal{Q}} = \bigcup_{i,j \in \{0,1\}} C_{ij}$, where $C_{i,j}$ is as defined in the proof of Lemma 4.6. In proving the next lemma we will show that this 7-partition is

indivisible.

The illustrated 7-plex H is defined by:

$$\operatorname{col}(x) = \begin{cases} E & \text{if } x \in \{1, 3, 5, 6, 10, 12\}, \\ F & \text{if } x \in \{7, 9, 11\}, \end{cases}$$

$$\operatorname{col}(0) = F \setminus \{13\} \cup \{10\},$$

$$\operatorname{col}(2) = F \setminus \{1\} \cup \{12\},$$

$$\operatorname{col}(4) = F \setminus \{3\} \cup \{2\},$$

$$\operatorname{col}(8) = F \setminus \{11\} \cup \{0\},$$

$$\operatorname{col}(13) = \{4, 6, 8\} \cup \{1, 3, 11, 13\}.$$

Now we are ready to prove Theorem 4.3 in the case that k is odd.

Lemma 4.8 ([65]). For all odd $h \ge 3$, the latin square Q_{2h} has an indivisible h-partition.

Proof. First we identify an h-partition. A 7-partition of Q_{14} is shown in (4.8). For $n = 2h \neq 14$, we show an h-plex H:

$$col(0) = \{n - 2\} \cup F \setminus \{n - 1\},$$

$$col(n - 1) = \{y \in E : y \equiv 2 \mod 4\} \cup \{y \in F : y \equiv 1 \mod 4\},$$

$$(4.9)$$

For $x \in F \setminus \{n-1\}$,

$$\operatorname{col}(x) = \begin{cases} E & \text{if } x < h, \\ F & \text{if } x \geqslant h. \end{cases} \tag{4.11}$$

For $x \in E \setminus \{0\}$, we specify col(x) in three cases, as follows.

Case 1: $h \equiv 1 \mod 4 \ (h \geqslant 5)$.

$$\operatorname{col}(x) = \begin{cases} \{x - 4\} \cup F \setminus \{x - 3\} & \text{if } x \in E \setminus \{0\}, \ x < h \text{ and } x \equiv 0 \bmod 4, \\ \{x - 2\} \cup F \setminus \{x - 1\} & \text{if } x \in E, \ x > h \text{ and } x \equiv 2 \bmod 4, \\ E & \text{otherwise.} \end{cases}$$

Case 2: $h \equiv 3 \mod 8 \ (h \geqslant 3)$.

For
$$h = 3$$
, $c(2) = \{0, 4, 3\}$, $c(4) = \{0, 2, 5\}$.

For $h \geqslant 11$,

$$col(h-3) = \{0\} \cup F \setminus \{1\},\$$

For
$$a \in \{0, 4, 8, \dots, \frac{1}{2}(h-11)\},\$$

$$col(h-7-a) = \{h-3-a\} \cup F \setminus \{h-2-a\},\$$

$$col(h-5-a) = \{n-6-a\} \cup F \setminus \{n-5-a\},\$$

$$col(h+1+a) = \{4+a\} \cup F \setminus \{5+a\},\$$

$$col(h+3+a) = \{h+1+a\} \cup F \setminus \{h+2+a\}.$$

Otherwise, col(x) = E.

Case 3: $h \equiv 7 \mod 8 \ (h \ge 15)$

For
$$a \in \{0, 4, 8, \dots, \frac{1}{2}(h-7)\},\$$

$$\begin{split} & \operatorname{col}(h-7-a) = \{h-3-a\} \cup F \setminus \{h-2-a\}, \\ & \operatorname{col}(h-1-a) = \{n-6-a\} \cup F \setminus \{n-5-a\}, \\ & \operatorname{col}(h+1+a) = \{a\} \cup F \setminus \{1+a\}, \\ & \operatorname{col}(h+3+a) = \{h+1+a\} \cup F \setminus \{h+2+a\} \quad \text{if } a \leqslant \frac{1}{2}(h-15). \end{split}$$

Otherwise, col(x) = E.

This completes all cases for $h \mod 8$. It is routine to check that H is an h-plex.

We proceed by contradiction to show that H and $H' = \mathcal{Q}_{2h} \setminus H$ are indivisible.

With c = 1, consider the sets A_{ij} , B_{ij} and C_{ij} of \mathcal{Q}_{2h} as described in (4.4). By definition, $\bigcup_{i,j\in\{0,1\}} C_{ij} = I_{\mathcal{Q}}$ where $I_{\mathcal{Q}}$ is given by (2.10).

Note that all cases of H, including that given separately in (4.8), obey (4.11) for their definition of $H \cap (B_{10} \cup B_{11}) \setminus r_{n-1}$. Further, by (4.8), (4.9) and (4.10), we see that $H \cap I_{\mathcal{Q}} = C_{11}$, $H \cap r_{n-1} \cap B_{11} = H \cap \Delta_*$ and $H \cap r_{n-1} \cap A_{11} = H \cap \Delta_2$. We rely on these facts in the following argument.

Assume that H is divisible. Then H contains an odd k-plex $K \subset H$ with k < h. It follows from Lemma 2.1 that $K \cap \Delta_* = H \cap \Delta_*$. Therefore $k = r_{n-1}^K \geqslant \Delta_*^K = \frac{1}{2}(h+1)$. Also $I_Q \cap H = C_{11} \subset (K \cap \Delta_*)$, so if $e \in K \setminus C_{11}$, then $e \in A_{ij}$ for some i, j. Consider $K \cap (A_{10} \cup A_{11}) \setminus r_{n-1}$. Following from (4.11) there are $\frac{1}{2}(h-1)$ rows r_x such that $H \cap r_x \subset A_{10}$ and $\frac{1}{2}(h-1)$ rows r_x such that $H \cap r_x \subset A_{11}$. Hence,

$$B_{11}^K = (A_{11} \setminus r_{n-1})^K + \Delta_*^K \geqslant \frac{1}{2}k(h-1) + \frac{1}{2}(h+1), \tag{4.12}$$

$$B_{10}^K = (A_{10} \setminus r_{n-1})^K + (r_{n-1} \setminus \Delta_*)^K \leqslant \frac{1}{2}k(h-1) + \frac{1}{2}(h-1). \tag{4.13}$$

Lemma 4.6 requires that $B_{11}^K - B_{10}^K \leq 1$ so the inequalities in (4.12) and (4.13) must be equalities. Also $B_{00}^K = B_{11}^K$ so $(B_{00} \cup B_{11})^K = 2B_{11}^K = kh - k + h + 1$ using (4.12). Counting occurrences of the symbols, $k|E| = (B_{00} \cup B_{11} \setminus C_{11})^K$ implies that kh = kh - k + h. This contradicts k < h and thus shows that H is indivisible.

Next we consider H'. In this case $I_{\mathcal{Q}} \cap H' = I_{\mathcal{Q}} \setminus C_{11}$. Following similar reasoning as for H we find that if $K \subset H'$ is an odd k-plex, then $(B_{10} \setminus r_{n-1})^K = (A_{10} \setminus r_{n-1})^K = \frac{1}{2}k(h-1)$, and $(B_{11} \setminus r_{n-1})^K = (A_{11} \setminus r_{n-1})^K = \frac{1}{2}k(h-1)$. Hence, by Lemma 4.6,

$$(r_{n-1} \cap B_{10})^K - (r_{n-1} \cap B_{11})^K = \pm 1. \tag{4.14}$$

We also note that $r_{n-1} \cap A_{11} \cap H' = \Delta_2 \cap H'$, with $\Delta_2^{H'} = \frac{1}{2}(h-1)$ and that $\Delta_{-2} = r_1 \cap H'$ so $\Delta_{-2}^{H'} = h$.

Let $J = H' \setminus K$ be a j-plex (with j > 0 even). By Lemma 2.1, $\sum_{e \in J \cap I_Q} \Delta(e)$ is even. Thus we consider two cases.

Case 1: $\sum_{e \in J \cap I_O} \Delta(e) = 0$.

Lemma 2.1 requires that $\Delta_2^J = \Delta_{-2}^J$. Also $j = \Delta_{-2}^J$ so it follows that $(r_{n-1} \cap B_{10})^J = j - \Delta_2^J = 0$. Then $(r_{n-1} \cap B_{10})^K = (r_{n-1} \cap B_{10})^{H'} = \frac{1}{2}(h+1)$. However now $(r_{n-1} \cap B_{10})^K > (r_{n-1} \cap B_{11})^{H'}$ so by (4.14), $(r_{n-1} \cap B_{11})^K = \frac{1}{2}(h-1)$ and $j = \Delta_2^J = 0$, which is a contradiction.

Case 2: $\sum_{e \in J \cap I_{\mathcal{O}}} \Delta(e) = 2$.

Lemma 2.1 requires that $\Delta_{-2}^J = \Delta_2^J + 1$ so $j = \Delta_2^J + 1$, and $(r_{n-1} \cap B_{10})^K = \frac{1}{2}(h-1)$. Thus, the right hand side of (4.14) is positive, so $\Delta_2^K = \frac{1}{2}(h-3)$ and we have k = h-2 and j = 2. By the Case 2 assumption, $C_{00} \subset K$, and $(C_{01} \cup C_{10}) \subset J$, hence $B_{11}^J = A_{11}^J = \Delta_2^J + \frac{1}{2}j(h-1) = h$, so by Lemma 4.6, $A_{00}^J = B_{00}^J = B_{11}^J = h$. Thus $A_{00}^J + A_{11}^J + (C_{01} \cup C_{10})^J = 2h+2$. Since this is not equal to j|E| = jh, we have a contradiction.

The contradictions in the two cases show that H' is indivisible.

4.2.2 Embedding plexes into larger latin squares

The results in this section are jointly due to Bryant, Egan, Maenhaut and Wanless [19].

In this section we will show ways to extend plexes into larger latin squares. By applying our methods to the indivisible plexes of Theorem 4.3 we will prove Theorems 4.9 and 4.10.

Theorem 4.9. For integers $k \ge 2$ and $m \ge 1$, there exists a latin square of order 2km with an indivisible k-partition.

We extend on Theorem 4.9 in Section 4.3 where we show indivisible k-partitions in latin squares of odd order.

Theorem 4.10.

$$\kappa(n) \geqslant \max(\frac{n}{p}, \lfloor \frac{n}{5} \rfloor) \text{ where } p \text{ is the smallest prime divisor of } n.$$
(4.15)

We will improve the result of Theorem 4.10 for odd n in Section 4.3.

We begin with some preliminary results on combining (proto) plexes into larger ones. We use definitions and notation associated with partial latin squares from Section 1.12.

Let P and P' be partial latin squares with pairwise disjoint index sets $\mathcal{I}(P)$ and $\mathcal{I}(P')$. We define the *direct sum* $P \oplus P'$ to be the PLS $P \cup P'$ with index set $\mathcal{I}(P) \cup \mathcal{I}(P')$. We define the *product* $P \otimes P'$ to be the PLS

$$P \otimes P' = \{((x, x'), (y, y'), (z, z')) : (x, y, z) \in P \text{ and } (x', y', z') \in P'\}.$$

with index set $\mathcal{I}(P) \times \mathcal{I}(P')$.

The next two results follow immediately from these definitions.

Lemma 4.11. If $K = K_1 \oplus K_2 \oplus \ldots \oplus K_s$ then K contains a c-protoplex if and only if K_i contains a c-protoplex for each $i \in \{1, 2, \ldots, s\}$.

Lemma 4.12. If K is an a-plex of order n and K' is a b-plex of order n', then $K \otimes K'$ is an ab-plex of order nn'. Furthermore, if K_1, K_2, \ldots, K_s is a partition of a latin square L and K'_1, K'_2, \ldots, K'_t is a partition of a latin square L', then $\{K_i \otimes K'_j : i = 1, 2, \ldots, s, j = 1, 2, \ldots, t\}$ is a partition of $L \otimes L'$.

An immediate consequence of Lemma 4.12 is that if K_1, K_2, \ldots, K_s is a k-partition of L and K'_1, K'_2, \ldots, K'_t is a k'-partition of L', then $L \otimes L'$ has a kk'-partition.

Lemma 4.13. Let K_0 and K_1 be parallel k-plexes in a latin square of order 2k. Then there is a k-partition of a latin square of order 4k all parts of which are isotopic to $K_0 \oplus K_1$.

Proof. We define a latin square of order 4k with index set $\mathcal{I}(K_0) \cup \{i' : i \in \mathcal{I}(K_0)\}$ and a k-partition $\{P_1, P_2, P_3, P_4\}$ where

$$P_{1} = K_{0} \cup \{(x', y', z') : (x, y, z) \in K_{1}\},$$

$$P_{2} = \{(x', y, z') : (x, y, z) \in K_{0}\} \cup \{(x, y', z) : (x, y, z) \in K_{1}\},$$

$$P_{3} = \{(x, y', z') : (x, y, z) \in K_{0}\} \cup \{(x', y, z) : (x, y, z) \in K_{1}\},$$

$$P_{4} = \{(x', y', z) : (x, y, z) \in K_{0}\} \cup \{(x, y, z') : (x, y, z) \in K_{1}\}.$$

It is clear that each P_i is isotopic to $K_0 \oplus K_1$.

We are now ready to prove Theorem 4.9.

Proof. (of Theorem 4.9) Let L be the latin square of order 2k with an indivisible k-partition guaranteed by Theorem 4.3. For $m \notin \{2,6\}$, we apply Lemma 4.12 to L and L' where L' is a latin square of order m possessing a 1-partition. For m=2 we apply Lemma 4.13 to L. For m=6 we apply Lemma 4.12 to L and a latin square of order 3, followed by Lemma 4.13. In all cases, Lemma 4.11 guarantees that the resulting k-partition is indivisible.

We will soon construct latin squares of various additional orders which contain k-plexes that are direct sums of two k-protoplexes. By taking one of these to be a k-protoplex from Theorem 4.3, the resulting k-plexes will be indivisible. First we prove restrictions on the order of a latin square that contains a k-plex that is a direct sum of two k-protoplexes.

Lemma 4.14. If $K = K_0 \oplus K_1$ is a k-plex, where K_0 is a k-protoplex of order m and K_1 is a k-protoplex of order n - m, then $n^2 - (3m + k)n + 3m^2 \ge 0$.

Proof. Let L be the latin square containing K and define

$$X = \{(x, y, z) \in L : x \in \mathcal{I}(K_0), y \in \mathcal{I}(K_1)\}.$$

Consider the m symbols of $\mathcal{I}(K_0)$. The number of occurrences of each of these in the rows indexed by $\mathcal{I}(K_0)$ is m, and the number of occurrences in K_0 is k. Hence the number of occurrences of each of these symbols in X is at most m-k. Similarly, the number of occurrences of each of the n-m symbols of $\mathcal{I}(K_1)$ in X is at most n-m-k. Since the total number of occurrences of symbols in X is m(n-m), it follows that $m(n-m) \leq m(m-k) + (n-m)(n-m-k)$. Rearranging, we obtain the required inequality.

Putting m = 2k in Lemma 4.14, we obtain $(n - 3k)(n - 4k) \ge 0$ and thus $n \le 3k$ or $n \ge 4k$. However, n < 3k implies that K_1 is a k-protoplex of order less than k, which is impossible. Thus, we have the following corollary which tells us the orders

of latin squares in which it may be possible to embed our indivisible k-protoplexes of order 2k.

Corollary 4.15. If $K = K_0 \oplus K_1$ is a k-plex, where K_0 is a k-protoplex of order 2k and K_1 is a k-protoplex of order n - 2k, then either n = 3k or $n \ge 4k$.

In what follows, we will show that there exists a latin square of order n with an indivisible k-plex for all n allowed (given that our k-plex is a direct sum involving a k-protoplex of order 2k) by Corollary 4.15, except for n in the range 4k < n < 5k.

Lemma 4.16. A k-protoplex K of order $m \ge 2k$ can be extended to a k-plex of order 2m - k that is a direct sum involving K.

Proof. The result is trivial in the case m=3 and k=1 so assume otherwise. Let K_0 be a k-protoplex of order m and let L_1 be a latin square of order m-k containing a k-plex K_1 . Since $m \ge 2k$ (and $(m,k) \ne (3,1)$), L_1 exists by Corollary 1.17. We now show that $K_0 \oplus L_1$ is completable, thus showing that $K_0 \oplus K_1$ is the required k-plex. To complete $K_0 \oplus L_1$ we need to

- (1) fill the rectangle $R_{01} = \{(x, y) : x \in \mathcal{I}(K_0), y \in \mathcal{I}(K_1)\},$
- (2) fill the rectangle $R_{10} = \{(x, y) : x \in \mathcal{I}(K_1), y \in \mathcal{I}(K_0)\}, \text{ and }$
- (3) fill the unfilled cells with row and column indices in $\mathcal{I}(K_0)$.

Each of (1)-(3) may be done independently, and each is equivalent to finding a 1-factorisation of an (m-k)-regular bipartite graph with m vertices in each part. The existence of such factorisations is well known, and is an easy consequence of Hall's Theorem [96].

Lemma 4.17. Let $m \ge k \ge 1$ and $n \ge 2m + k$. For any k-plex K of order m, there exists a k-plex $K \oplus K'$ of order n.

Proof. Let L be a latin square of order m containing a k-plex K. Let S be a set of n-m symbols distinct from $\mathcal{I}(L)$ and let γ be a permutation applying a single n-m cycle to S. By [152], and as mentioned in Section 1.11, there exists a (Parker) latin square $L' \supset L$ of order n, indexed by $\mathcal{I}(L) \cup S$ and such that L' has an automorphism α that applies γ simultaneously to the row, column and symbol indices. We obtain the required k-plex by taking the union of K with k orbits of α on entries whose row, column and symbol indices are in S. The condition $n \geq 2m + k$ ensures that $|S| - m \geq k$, so there are enough suitable orbits for this construction.

Note that applying Lemma 4.16 or Lemma 4.17 to the indivisible k-plex of order 2k from Theorem 4.3, we obtain an indivisible k-plex of order n for n=3k and all $n \geq 5k$, respectively. Also, applying Theorem 4.9 with m=2 gives us, for each $k \geq 2$, a latin square of order 4k with an indivisible k-plex. We state these facts in the following corollary.

Corollary 4.18. For all $k \ge 2$, there exists an indivisible k-plex of order n for n = 3k, n = 4k and all $n \ge 5k$.

We are now ready to prove Theorem 4.10 which states that $\kappa(n) \geqslant \max(\frac{n}{p}, \lfloor \frac{n}{5} \rfloor)$ where p is the smallest prime divisor of n.

Proof. (of Theorem 4.10) Let p be the smallest prime divisor of n. If p=2 or p=3, then the existence of a latin square of order n with an indivisible $(\frac{n}{p})$ -plex is guaranteed by Theorem 4.3 or the n=3k result in Corollary 4.18, respectively. If $p \geq 5$, then the existence of a latin square of order n with an indivisible $\lfloor \frac{n}{5} \rfloor$ -plex follows from the $n \geq 5k$ result in Corollary 4.18.

4.3 Latin squares of odd order

We now investigate, by direct construction, indivisible plexes in latin squares of odd order.

4.3.1 Indivisible partitions of $\mathcal{O}_{m,k}$

We use the latin family $\mathcal{O}_{m,k}$ which is defined for all odd integers $m, k \geq 3$.

Lemma 4.19. For odd integers $m \ge 3$ and $k \ge 3$, the latin square $\mathcal{O}_{m,k}$ of order mk has an indivisible k-partition.

Proof. For each $i \in \mathbb{Z}_m$, a k-plex K_i is given by

$$\operatorname{col}((a,b)) = \{(a+i, d) : d \in \mathbb{Z}_k \text{ for each } (a,b) \in \mathcal{I}(\mathcal{O}_{m,k})\}.$$
(4.16)

It is routine to verify that (4.16) identifies a k-partition of $\mathcal{O}_{m,k}$. We need to show that each K_i is indivisible.

Assume that $C \subset K_i$ is a c-plex for some 0 < c < k and some $i \in \mathbb{Z}_m$. Consider the projection $\pi : \mathcal{O}_{m,k} \to \mathbb{Z}_k$ of $\Delta(\mathcal{O}_{m,k})$ onto the second coordinate. Let $\pi_* = \left(\pi^{-1}(1) \cup \pi^{-1}(-1)\right) \cap C$. The construction of $\mathcal{O}_{m,k}$ and K_i imply that π is zero on $C \setminus \pi_*$ and that $|\pi_*| \in \{c, 2c\}$. Specifically, if $i \in \{0, m-1\}$, then $\pi_* \subseteq \pi^{-1}(1)$ and $|\pi_*| = c$. If $i \in \{1, 3, 5, \ldots, m-2\}$ then $\pi_* \subseteq \pi^{-1}(-1)$ and $|\pi_*| = 2c$. Otherwise $i \in \{2, 4, 6, \ldots, m-3\}$, so $\pi_* \subseteq \pi^{-1}(1)$ and $|\pi_*| = 2c$. In each case, Lemma 2.1 implies that k divides k which contradicts k0 k1. Thus k2 is indivisible.

Lemma 4.19 yields no improvement to the lower bound on $\kappa(n)$ stated by (4.15). However, for any odd n divisible by 3 or 5, we match the bound of (4.15) with a simpler construction than was used to prove Theorem 4.10. Using $\mathcal{O}_{m,k}$, we can

apply the technique in the proof of Theorem 4.9 to obtain a latin square of order dmk with a neat indivisible k-partition, except in cases $d \in \{2, 6\}$.

We can now prove Theorem 4.1.

Proof. (of Theorem 4.1) Assume that n = mk where m, k are positive integers such that $k \neq n > 2$ and $(m, k) \neq (6, 1)$. For k = 1 we rely on Theorem 1.16. For all odd integers $k \geqslant 3$ and $m \geqslant 3$, Lemma 4.19 shows the required k-partition. For all even m and $k \geqslant 2$, we use an indivisible k-partition given by Theorem 4.9.

4.3.2 Indivisible partitions of \mathcal{H}_n

We now use the family \mathcal{H}_n which is defined for all odd $n \geq 5$.

Lemma 4.20. If P is (k, k+1)-partition of \mathcal{H}_{2k+1} such that Δ_2 is in one part and Δ_{-2} is in the other, then P consists of an indivisible k-plex and a (k+1)-plex that contains no c-plex for 1 < c < k.

Proof. Let $K \subset \mathcal{H}_{2k+1}$ be a k-plex such that K satisfies the hypothesis. Now $|\Delta_2| = |\Delta_{-2}| = k$, so respecting the rows of K and the sum required by Lemma 2.1, K has exactly one element of $\Delta_1 \cup \Delta_{-1}$ and it must be in r_1 . Any c-plex $C \subset K$ must intersect Δ_* , but no proper subset of $\Delta_* \cap K$ will satisfy Lemma 2.1. Thus K is indivisible. Similarly, if $C \subset K'$ is a c-plex for some c > 1, then C contains some element of Δ_a for $a = \pm 2$. By Lemma 2.1, C must then contain k such elements and it follows that C is an indivisible k-plex.

Lemma 4.21. The latin square \mathcal{H}_n has an indivisible $(1, k^2)$ -partition.

Proof. We show two parts of the partition.

A transversal T is given by

$$\operatorname{col}(x) = \begin{cases} \left\{ n - \frac{1}{2}x \right\} & \text{if } x \text{ is even,} \\ \left\{ \frac{1}{2}(n-x) \right\} & \text{otherwise.} \end{cases}$$
 (4.17)

For each $(x, y, z) \in T$ let t(x) = y. We define a k-plex K_1 , parallel to T, as follows:

$$\operatorname{col}(x) = \begin{cases} \{1, 3, 5, \dots, n-2\} & \text{if } x = 0, \\ \{2, 3, 4, \dots, t(x) - 1\} \cup \{n-1\} & \text{if } x = 1, \\ \{0, 2, 4, \dots, n-3\} & \text{if } x = 2, \\ \{t(x) + 1, t(x) + 2, t(x) + 3, \dots, t(x) + k\} & \text{otherwise.} \end{cases}$$

It is routine to show that K_1 is a k-plex and that it is parallel to T. Thus, $K_2 = \mathcal{H}_n \setminus (K_1 \cup T)$ is a k-plex and we have shown a $(1, k^2)$ -partition. By Lemma 4.20, this partition is indivisible.

In (4.18) we illustrate the indivisible partition of Lemma 4.21 in \mathcal{H}_9 . Entries marked $_*$ identify elements of Δ_* .

We now prove Theorem 4.2 and so conclude the main results of this chapter.

Proof. (of Theorem 4.2) The case for all even n > 2 is shown by Theorem 4.3. For n = 3, the species of \mathbb{Z}_3 possesses a 1-partition. The case for all odd n > 3 is shown by Lemma 4.21.

Although an argument using Lemma 2.1 for an indivisible (k+1)-plex in \mathcal{H}_{2k+1} is not obvious, we wonder if this family does admit larger indivisible plexes. Computations verify that \mathcal{H}_n has an indivisible (k, k+1)-partition for all $n \leq 11$. Also, for $n \leq 9$, we confirmed that \mathcal{H}_n has no larger indivisible plex. In (4.19) we give an example of an indivisible (k, k+1)-partition of \mathcal{H}_9 (left) and \mathcal{H}_{11} (right). In each case, other than the absence of transversals in the larger part, Lemma 4.20 explains the indivisibility. The entries marked * show elements of Δ_* .

| | | | | | | | | | | | 0 | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | 10 |
|---|----|----|---|----|---|---------|---|-------|------------------|----|----|----|----|----|----|-------|----|-------|----|----|------------------|
| | 0 | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 0 | 1* | 3* | 2 | 5* | 4 | 7_* | 6 | 9* | 8 | 0* | 10 |
| 0 | 1* | 3* | 2 | 5* | 4 | 7_{*} | 6 | 0* | 8 | 1 | 0* | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | 10 | 1* |
| 1 | 0* | 2 | 3 | 4 | 5 | 6 | 7 | 8 | $\overline{1_*}$ | 2 | 2 | 1* | 4 | 3* | 6 | 5* | 8 | 7_* | 10 | 9* | 0 |
| 2 | 2 | 1* | 4 | 3* | 6 | 5* | 8 | 7_* | 0_{*} | 3 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | 10 | 0 | 1 | 2 |
| 3 | 3 | 4 | 5 | 6 | 7 | 8 | 0 | 1 | 2 | 4 | 4 | 5 | 6 | 7 | 8 | 9 | 10 | 0 | 1 | 2 | 3 |
| 4 | 4 | 5 | 6 | 7 | 8 | 0 | 1 | 2 | 3 | 5 | 5 | 6 | 7 | 8 | 9 | 10 | 0 | 1 | 2 | 3 | 4 |
| 5 | 5 | 6 | 7 | 8 | 0 | 1 | 2 | 3 | 4 | 6 | 6 | 7 | 8 | 9 | 10 | 0 | 1 | 2 | 3 | 4 | 5 |
| 6 | 6 | 7 | 8 | 0 | 1 | 2 | 3 | 4 | 5 | 7 | 7 | 8 | 9 | 10 | 0 | 1 | 2 | 3 | 4 | 5 | 6 |
| 7 | 7 | 8 | 0 | 1 | 2 | 3 | 4 | 5 | 6 | 8 | 8 | 9 | 10 | 0 | 1 | 2 | 3 | 4 | 5 | 6 | 7 |
| 8 | 8 | 0 | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 9 | 9 | 10 | 0 | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 |
| , | | | | | | | | | | 10 | 10 | 0 | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 |
| | | | | | | | | | | | 1 | | | | | | | | | | $\overline{(4)}$ |

Curiously, computation shows that in \mathcal{H}_9 every indivisible 5-plex occurs in a (4,5)-partition obeying the hypothesis of Lemma 4.20. Thus, every indivisible 5-plex in

 \mathcal{H}_9 occurs in an indivisible (4,5)-partition. In smaller members of \mathcal{H}_n alternative partitions of Δ_* can yield an indivisible (k, k+1)-partition. In \mathcal{H}_7 the complement of an indivisible 4-plex in \mathcal{H}_7 is not necessarily indivisible. The smallest family member \mathcal{H}_5 does exhibit every one of its indivisible 3-plexes in an indivisible (2,3)-partition. However, every 2-plex in \mathcal{H}_5 is indivisible since it does not have two parallel transversals. It would be interesting to know if the observed condition in \mathcal{H}_9 occurs in larger latin squares.

Problem 4.22. For n > 9, does there exist a latin square L of order n that possesses an indivisible k-plex and every indivisible k-plex in L is in an indivisible (k, n - k)-partition?

As an aside to indivisible plexes, we next point out that \mathcal{H}_n is a confirmed bachelor. We thus offer, next, a very simple alternative proof of Theorem 1.23 for the case of all odd n > 3.

Theorem 4.23. The elements (1,0,0) and (1,n-1,1) of \mathcal{H}_n are transversal-free. Hence, for all odd n > 3, the latin square \mathcal{H}_n is a confirmed bachelor latin square of order n.

Proof. Assume that $T \subset \mathcal{H}_n$ is a transversal and that $(1,0,0) \in T$. No other element of $\Delta_1 \cup \Delta_{-1}$ is possible in T as each of them agrees with (1,0,0) in at least one coordinate. Considering r_0 and r_2 , T has at most one element from each of Δ_2 and Δ_{-2} , but then $\sum_{e \in T} \Delta(e) \equiv 0 \mod n$ is impossible. Thus, by Lemma 2.1, (1,0,0) is not in a transversal. Similarly, no transversal in \mathcal{H}_n contains (1, n-1, 1).

4.3.3 Indivisible partitions of \mathbb{Z}_n

The existence of indivisible 2-partitions in \mathbb{Z}_n for n even is shown by Theorem 1.27. The next result is first evidence, in general, of indivisible plexes in the Cayley table of \mathbb{Z}_n when it does possess transversals.

Lemma 4.24. For all odd n and integers $t \equiv n \mod 4$, the Cayley table of \mathbb{Z}_n has an indivisible $(1^t, 2^{(n-t)/2})$ -partition.

Proof. The existence of a 1-partition is well known. For $a \in \{0, 1, ..., n-1\}$, a transversal T_a is given by $\operatorname{col}(x) = \{x + a\}$.

For each $d \in D = \{0, 1, 2, \dots, 2 \left| \frac{n}{4} \right| - 1\}$, a 2-plex J_d is next shown.

If d is even,

$$\operatorname{col}(x) = \begin{cases} \{x + 2d + 1, \ x + 2d + 4\} & \text{if } x = \left\lfloor \frac{n}{2} \right\rfloor - 2, \\ \{x + 2d + 1, \ x + 2d + 2\} & \text{if } (x = \left\lfloor \frac{n}{2} \right\rfloor - 1) \text{ or } (x = n - 2), \\ \{x + 2d + 2, \ x + 2d + 3\} & \text{if } x = n - 1, \\ \{x + 2d + 1, \ x + 2d + 3\} & \text{otherwise,} \end{cases}$$

and if d is odd,

$$\operatorname{col}(x) = \begin{cases} \{x + 2d, \ x + 2d + 1\} & \text{if } x = \left\lfloor \frac{n}{2} \right\rfloor - 2, \\ \{x + 2d + 1, \ x + 2d + 2\} & \text{if } (x = \left\lfloor \frac{n}{2} \right\rfloor - 1) \text{ or } (x = n - 2), \\ \{x + 2d - 1, \ x + 2d + 2\} & \text{if } x = n - 1, \\ \{x + 2d, \ x + 2d + 2\} & \text{otherwise.} \end{cases}$$

If d is even, then J_d contains the 3 elements (n-3, 2d, 2d-3), (n-2, 2d, 2d-2) and (n-2, 2d-1, 2d-3). Since each pair of these triples agree in some coordinate, J_d cannot be divided into parallel transversals. Similarly, if d is odd, using elements in rows $\left\lfloor \frac{n}{2} \right\rfloor - 2$ and $\left\lfloor \frac{n}{2} \right\rfloor - 3$ shows that J_d is indivisible.

It is routine to show that, for a fixed even $d \in D$, the union $J_d \cup J_{d+1}$ is equal to the 4-plex given by

$$col(x) = \{x + 2d + 1, x + 2d + 2, x + 2d + 3, x + 2d + 4\}.$$

Thus it follows that for a fixed even $d \in D$,

$$J_d \cup J_{d+1} = T_{2d+1} \cup T_{2d+2} \cup T_{2d+3} \cup T_{2d+4}$$
.

An example of the required partition is given by

$$\{T_a: a \in \{0, n-1, n-2, \dots, n-t-2\}\} \cup \{J_d: d \in \{0, 1, 2, \dots, 2\left|\frac{n-t}{4}\right|-1\}\}.$$

For $n \equiv 3 \mod 4$, Lemma 4.24 falls short of an indivisible $(1, 2^{(n-1)/2})$ -partition which we expect by Conjecture 4.34 in Section 4.6.

In Section 4.5 we will table data on the indivisible partitions of every type occurring in \mathbb{Z}_n for $n \leq 9$.

4.4 Subsquares and restrictions on odd plexes

In this section we investigate ways in which plexes and partitions may fail to contain odd plexes.

Assume that L is a latin square of order n=a+b and that $A\subseteq L$ is a subsquare of order a. A well known necessary condition is that $a\leqslant b$. By Theorem 1.35, the condition that $a\leqslant b$ is sufficient for embedding of a latin square A' of order a into some latin square of order n=a+b. Existence of the subsquare A in L implies that, up to permutation of the rows and columns, L has the form:

$$L = \begin{pmatrix} A & S \\ R & B \end{pmatrix}, \tag{4.20}$$

where B, R and S are $b \times b$, $b \times a$, $a \times b$ submatrices respectively. Let C denote the subset of B whose symbols are in A and let $D = B \setminus C$. Suppose that K_A is a (b-a)-plex in A. Simple counting shows that $K_A \cup D$ is a (b-a)-plex of L.

Let K be a k-plex in L and assume that $|A \cap K| = x$. Counting cells of K in the rows of A implies that $|K \cap C| = ak - x$. Similarly, $|K \cap R| = ak - x$. It follows that

$$|K \cap D| = bk - 2(ak - x).$$
 (4.21)

In particular, $|K \cap D|$ is odd if and only if b and k are both odd. In 1944, Mann [117] used this argument to prove that a latin square of order $n \equiv 1 \mod 4$, with a subsquare of order (n-1)/2, has no orthogonal mate. As Mann showed, the number of disjoint transversals of such a latin square L is at most |D| = b < n. By a similar argument using (4.21), if b is odd and $D \subseteq P \subseteq L$ for some plex P, then $L \setminus P$ contains no odd plexes. For example, shading in (4.22) identifies a 2-plex P for which the complementary 6-plex contains no odd plex.

$$\begin{pmatrix}
2 & 1 & 0 & 5 & 6 & 3 & 4 & 7 \\
1 & 0 & 2 & 6 & 7 & 5 & 3 & 4 \\
0 & 2 & 1 & 7 & 5 & 4 & 6 & 3 \\
\hline
5 & 6 & 7 & 3 & 4 & 0 & 1 & 2 \\
6 & 7 & 5 & 4 & 3 & 2 & 0 & 1 \\
3 & 5 & 4 & 0 & 2 & 1 & 7 & 6 \\
4 & 3 & 6 & 1 & 0 & 7 & 2 & 5 \\
7 & 4 & 3 & 2 & 1 & 6 & 5 & 0
\end{pmatrix}$$

$$(4.22)$$

More generally:

Lemma 4.25. Suppose that n and j are positive integers such that $\frac{2}{3}n \leq j \leq n$ and $j \equiv 2(n-1) \mod 4$. Then there exists a latin square of order n with a j-plex that contains no odd plex.

Proof. First consider small n. If $n \in \{1, 3, 4, 7\}$, then the lemma is true vacuously. If $n \in \{2, 5\}$, then (n, j) = (2, 2) or (5, 4). Case (2, 2) is trivial, and (5, 4) is shown by a latin square of order 5 whose 3 transversals coincide on a single entry. The complement of any transversal is therefore a 4-plex containing no odd plex.

To prove all remaining cases we will show existence of an appropriate L of the form (4.20), such that b is odd. We shall identify a j-plex $J \subseteq L$ such that $D \cap J = \emptyset$. The lemma will then immediately follow from (4.21).

Set a = j/2. The hypothesis $\frac{2}{3}n \leqslant j \leqslant n$ implies that $\frac{1}{3}n \leqslant a \leqslant \frac{1}{2}n$. Now $a \leqslant \frac{1}{2}n$ is necessary and sufficient to obtain a latin square of the form (4.20). Also $a \geqslant \frac{1}{3}n$ ensures that $a \geqslant n - 2a = b - a$, so the subsquare A is big enough to contain a (b-a)-plex. Corollary 1.17 confirms that $(b-a) \leqslant a$ is sufficient for us to choose a latin square A, of order a > 2, that possesses a (b-a)-plex K_A .

For odd $n \ge 9$, $j = 2a \equiv 0 \mod 4$ implies that a is even. Also, $a \ge \frac{1}{3}n \ge 3$. On the other hand, for even $n \ge 6$, $j = 2a \equiv 2 \mod 4$ implies that a is odd, and $4 \le \frac{2}{3}n \le 2a \equiv 2 \mod 4$ implies that $a \ge 3$. In either case, b = n - 3a is an odd integer and $a \ge 3$.

Now $K = K_A \cup D$ is a (b-a)-plex in L. Thus, $J = L \setminus K$ is the required even j-plex.

For $j=n\equiv 2 \mod 4$, Lemma 4.25 says no more than a well known fact; a latin square of order n=2a with a subsquare of odd order a possesses no odd plex. (e.g. Theorem 1.25.) There is no guarantee in Lemma 4.25 that L possesses any odd plexes, other than L itself when n is odd. We next show that certain odd plexes exist when extra structural constraints are imposed on L. We need the following result which is an immediate consequence of the existence of Parker squares [152, Thm 6]. Parker squares were mentioned in Section 1.11.

Lemma 4.26. There exists a Latin square L of the form (4.20) which has a cyclic automorphism α of order b. The subsquare A of L occurs in the rows and columns indexed by the fixed points of α .

Theorem 4.27. Assume that n, j, k are positive integers such that $\frac{2}{3}n \leq j \leq n$, $j \equiv 2(n-1) \mod 4$ and k=n-j. Then there exists a latin square of order n with a $(1^k, j)$ -partition in which the j-plex contains no odd plex.

Proof. We follow the proof of Lemma 4.25, but use the extra structure from Lemma 4.26. The cases where $n \in \{1, 2, 3, 4, 5, 7\}$ are exactly as in the previous proof. For all a except 2 and 6, we use as the subsquare A, a latin square of order a with a 1-partition guaranteed by Theorem 1.16. In the proof of Lemma 4.25 we have a > 2, thus our only concern is if a = 6. If a = 6, then there exists a latin square with a $(1^4, 2)$ -partition (see Section 4.5, [76] or [150]) which satisfies all but one case. The exception is when n = 17, a = 6, b = 11 and k = 5. However, example (4.23)

then shows the required $(1^5, 12)$ -partition. Five parallel transversals are marked by subscripts a, b, c, d and e. Together they cover D so the complementary 12-plex contains no odd plex, by (4.21).

| / | 0 | 1_b | 2 | 3_d | 4_e | 5 | 6 | 7 | 8 | 9 | 10 | 11_c | 12 | 13 | 14 | 15_a | 16 \ | | | |
|---|--------|---------|-------|--------|--------|--------|--------|--------|--------|--------|-------|--------|--------|--------|--------|--------|--------|---|--------|----|
| 1 | 1_a | 2_e | 3 | 4 | 5 | 6 | 7 | 0 | 9 | 10 | 11 | 12 | 13 | 14_c | 15_b | 16 | 8_d | | | |
| ı | 2_b | 3_c | 4_a | 5 | 6 | 7 | 0 | 1 | 10 | 11_e | 12 | 13 | 14 | 15_d | 16 | 8 | 9 | l | | |
| l | 3 | 4_d | 5 | 6_a | 7 | 0_e | 1 | 2 | 11 | 12 | 13 | 14_b | 15_c | 16 | 8 | 9 | 10 | | | |
| I | 4_c | $ 5_a $ | 6_b | 7 | 0 | 1 | 2 | 3 | 12_d | 13 | 14 | 15 | 16 | 8 | 9_e | 10 | 11 | l | | |
| ı | 5 | 6 | 7_e | 0 | 1 | 2 | 3_a | 4 | 13 | 14 | 15 | 16 | 8 | 9 | 10_d | 11_b | 12_c | l | | |
| I | 6 | 7 | 0_c | 1 | 2_a | 3 | 4 | 5_d | 14 | 15 | 16 | 8 | 9 | 10_e | 11 | 12 | 13_b | l | | |
| l | 7 | 0 | 1_d | 2_c | 3_b | 4 | 5 | 6 | 15 | 16 | 8 | 9 | 10 | 11 | 12 | 13_e | 14_a | | | |
| ١ | 8 | 9 | 10 | 11 | 12 | 13 | 14 | 15 | 16_a | 0_b | 1_c | 2_d | 3_e | 4 | 5 | 6 | 7 | l | (4.23) | ,) |
| ١ | 9 | 10 | 11 | 12 | 13 | 14 | 15 | 16_c | 7_b | 8_a | 0_d | 1_e | 2 | 3 | 4 | 5 | 6 | l | | |
| | 10 | 11 | 12 | 13 | 14 | 15 | 16_b | 8_e | 6_c | 7_d | 9_a | 0 | 1 | 2 | 3 | 4 | 5 | | | |
| ١ | 11 | 12 | 13 | 14 | 15 | 16_d | 8_c | 9_b | 5_e | 6 | 7 | 10_a | 0 | 1 | 2 | 3 | 4 | l | | |
| ١ | 12 | 13 | 14 | 15 | 16 | 8_b | 9_d | 10 | 4 | 5_c | 6_e | 7 | 11_a | 0 | 1 | 2 | 3 | l | | |
| ١ | 13_d | 14 | 15 | 16_e | 8 | 9_c | 10 | 11 | 3 | 4 | 5 | 6 | 7 | 12_b | 0_a | 1 | 2 | | | |
| l | 14_e | 15 | 16 | 8 | 9 | 10 | 11 | 12 | 2 | 3 | 4_b | 5 | 6_d | 7_a | 13_c | 0 | 1 | | | |
| 1 | 15 | 16 | 8 | 9 | 10_c | 11 | 12_e | 13_a | 1 | 2 | 3 | 4 | 5_b | 6 | 7 | 14_d | 0 | | | |
| \ | (16 | 8 | 9 | 10_b | 11_d | 12_a | 13 | 14 | 0 | 1 | 2 | 3 | 4 | 5 | 6 | 7_c | 15_e | 1 | | |

In all remaining cases, each of the required k disjoint transversals can be found by taking the union of a transversal of A with an appropriate orbit of α in B.

Corollary 4.28. For $n \ge 6$ and even $j \le \frac{1}{3}n$, there exists a latin square of order n with a j-plex J that can be refined into parallel 2-plexes but J contains no odd plex.

Proof. We choose L according to Lemma 4.26 and ensure that $A = \mathbb{Z}_j$. The condition $j \leq \frac{1}{3}n$ implies $j = a \leq (b-a) = n-2j$. So D contains at least j orbits of α that include the b symbols of L that do not occur in A. Let K_B be the union of j such orbits. Then $J = A \cup K_B$ is a j-plex of L. Now $A = \mathbb{Z}_j$ has a 2-partition but has no odd plex, by Theorem 1.27. By combining a 2-plex of A with two orbits of α from K_B we obtain a 2-plex of L. Thus J can easily be refined into parallel 2-plexes. However it contains no odd plex, since A contains no odd plex.

In the proof of the next lemma, we construct, among other things, an (n-2)-plex with no odd plex in a latin square of order $n \equiv 2 \mod 4$ that also possesses transversals.

Lemma 4.29. For even integers $n \ge 6$ and $j \le n$ such that $j \equiv 0 \mod 4$, there exists a latin square L of order n such that

- 1. Any partition of L has at most j odd parts.
- 2. If $j' \equiv 0 \mod 4$ and $0 \leqslant j' \leqslant j$, then L has an indivisible $(1^{j'}, 2^{(n-j')/2})$ -partition.

3. If j'' is even and $j'' \leq n - j$, then L has a j''-plex that contains no odd plex.

Proof. We prove the statement for members of the family $\mathcal{V}_{n,j}$, as defined in Section 2.4.2. We recall that, as specified for $\mathcal{V}_{n,j}$, u=j/4, h=n/2 and $U=\{0,1,2,\ldots,u-1\}$. Also, $|\Delta_*|=j$ and every element e in Δ_* has $\Delta(e)=h$. Thus, Part 1 of the lemma is shown by Lemma 2.1.

For each $v \in U$ and each $i \in \{0, 1, 2, 3\}$, we a transversal $T_{v,i}$ is given by

$$\operatorname{col}(x) = \begin{cases} \{x + 2v\} & \text{if } (i = 2 \text{ and } x = 0) \text{ or } (i = 3 \text{ and } x = h), \\ & \text{or } (i = 0 \text{ and } 0 < x < h) \text{ or } (i = 1 \text{ and } x > h) \end{cases}$$

$$\{x + 2v + 1\} & \text{if } (i = 0 \text{ and } x \geqslant h) \text{ or } (i = 1 \text{ and } x < h),$$

$$\{x + h + 2v\} & \text{if } (i = 0 \text{ and } x = 0) \text{ or } (i = 1 \text{ and } x = h) \\ & \text{or } (i = 2 \text{ and } 0 < x < h) \text{ or } (i = 3 \text{ and } x > h),$$

$$\{x + h + 2v + 1\} & \text{if } (i = 2 \text{ and } x \geqslant h) \text{ or } (i = 3 \text{ and } x < h).$$

It is routine to show that the collection $\{T_{v,i}\}$ for $v \in U$ and $i \in \{0, 1, 2, 3\}$ defines a $(1^j, n-j)$ -partition. It should be clear that for a fixed $v \in U$, the union, $\bigcup_{i \in \{0,1,2,3\}} T_{v,i}$ is equivalent to the 4-plex given by

$$col(x) = \{x + 2v, \ x + 2v + 1, \ x + h + 2v, \ x + h + 2v + 1\}. \tag{4.24}$$

In what follows, we will define 2-plexes which will either refine this 4-plex, or are parallel to it.

Let $H = \{0, 1, 2, ..., h - 1\}$. For each $d \in H$, we show a 2-plex J_d , as follows. For $n \equiv 2 \mod 4$,

$$col(x) = \{x + d, x + h + d\},\tag{4.25}$$

and for $n \equiv 0 \mod 4$,

$$col(x) = \begin{cases} \{x+d, \ x+h+d\} & \text{if } x < h-1 \text{ or } x = n-1, \\ \{x+d+(-1)^d, \ x+h+d+(-1)^d\} & \text{otherwise.} \end{cases}$$
(4.26)

We next show that J_d is indivisible. If $n \equiv 2 \mod 4$, then J_d is the union of h intercalates and hence is indivisible. So suppose that $n \equiv 0 \mod 4$ and let q = n/4. In the case when d is even, consider J_d in rows n-2, n-1, q-1 and 3q-2. These entries lie outside of Δ_* . They are

$$\begin{array}{ll} (n-2,\ d-1,\ d-3), & (n-2,\ h+d-1,\ h+d-3), \\ (n-1,\ d-1,\ d-2), & (n-1,\ h+d-1,\ h+d-2), \\ (q-1,\ q+d-1,\ h+d-2), & (q-1,\ 3q+d-1,\ d-2), \\ (3q-2,\ q+d-1,\ d-3), & (3q-2,\ 3q+d-1,\ h+d-3). \end{array}$$

In any partition of these 8 triples into two sets, one of the sets will contain a pair of triples that agree in some coordinate. Hence J_d cannot be divided into parallel transversals. The case for odd d is similarly shown by using the triples in rows h-2, h-1, q-2 and 3q-1. This proves that J_d is indivisible.

Comparing (4.24) with (4.25) and (4.26), shows that for a fixed $v \in U$,

$$J_{2v} \cup J_{2v+1} = \bigcup_{i \in \{0,1,2,3\}} T_{v,i}. \tag{4.27}$$

Proof of Part 2 and Part 3: For $j' \equiv 0 \mod 4$ and $0 \leqslant j' \leqslant j$, an indivisible $(1^{j'}, 2^{(n-j')/2})$ -partition is given by

$$\left\{T_{v,i}: v \in \{0, 1, 2, \dots, \frac{1}{4}j' - 1\}, i \in \{0, 1, 2, 3\}\right\} \cup \left\{J_d: d \in \left\{\frac{1}{2}j', \frac{1}{2}j' + 1, \dots, h - 1\right\}\right\}.$$

The equivalence (4.27) implies that, as required, the plexes in the partition are parallel. Thus we have shown Part 2. For Part 3, set j' = j in the $(1^{j'}, 2^{(n-j')/2})$ -partition of Part 2. Since, by Part 1, this partition has at most j odd parts we form, by union of an appropriate number of 2-plexes from this partition, a j''-plex for each even $j'' \leq n - j$. This completes our proof of the lemma.

Permitting n=4 for the family $\mathcal{V}_{n,j}$ in the proof of Lemma 4.29 yields $\mathcal{V}_{4,0}$ and $\mathcal{V}_{4,4}$, the species of \mathbb{Z}_4 and $\mathbb{Z}_2 \oplus \mathbb{Z}_2$, respectively. The plexes and partitions given in (4.24), (4.26) and (4.27) work for n=4 but Parts 2 and 3 of Lemma 4.29 fail for $\mathcal{V}_{4,4}$. As recorded in [76], and confirmed in the next section, every 2-plex in $\mathbb{Z}_2 \oplus \mathbb{Z}_2$ is divisible.

Corollary 4.30. For all $n \ge 5$ and

$$k = \begin{cases} 1 & \text{if } n \equiv 1 \mod 4, \\ 3 & \text{if } n \equiv 3 \mod 4, \\ 2 & \text{if } n \text{ is even,} \end{cases}$$

there exists a latin square L of order n such that L has at least k disjoint transversals and L possesses an (n-k)-plex that contains no odd plex. If $n \not\equiv 2 \mod 4$ then the (n-k)-plex occurs in a $(1^k, n-k)$ -partition.

Proof. The case $n \equiv 2 \mod 4$ is shown by Lemma 4.29. Theorem 4.27 shows every other case, except (n, k) = (7, 3). For the exception we rely on an example. Table 4.6 in the next section confirms the existence of a latin square of order 7 with an indivisible $(1^3, 4)$ -partition.

We leave open the possibility of an (n-1)-plex with no odd plex among the latin squares of order $n \equiv 3 \mod 4$. If an example exists it must be of order at least 11 since, by computation, among the latin squares of order 7 every 6-plex contains an odd plex.

4.5 Latin squares of small order

The computational results in this section have been independently confirmed by Ian Wanless.

As mentioned in Section 1.14.1, several earlier reports in the literature consider plexes in small latin squares but the value of $\kappa(n)$ for small n has not been explicitly computed. We compute $\kappa(n)$ exactly for $n \leq 8$ and find that $\kappa(9) \in \{6, 7\}$. In the process we show that Conjecture 1.2 is true for the latin squares of order 9.

We summarise, in Section 4.5.2, the results of a complete enumeration of the plexes, indivisible plexes and indivisible partitions in all species representatives of order $n \leq 8$. Our computations for order 9 are later explained in Section 4.5.3. These computations were expensive. We did not keep a precise account of the computation time for this project but estimate that in total it was several decades.

To establish that a given k-plex was indivisible we checked, by exhaustive search, that it contained no smaller c-plex, for $1 \leq c \leq \frac{1}{2}k$. The author and Wanless independently wrote programs and ran them on a representative of each species for orders $n \in \{4, 5, 6, 7, 8\}$. We thus obtained two independent sets of raw data from which the tables are drawn. Information on the paratopy group size of each species representative was used in the final stage of calculations.

4.5.1 Definitions and explanation of tables

A distinctive type of indivisible partition is one that occurs in some, but not all species of order n. A category is a set of species that possess the same (distinctive) types of indivisible partitions. Tables in this section contain data on categories for each order $n \in \{6,7,8\}$. The marks in these tables are \checkmark or \checkmark depending on whether or not the specified indivisible partition type occurs. For each order $n \in \{6,7,8\}$ we will also present a table summarising enumerative data. The data under each of the column headings of these tables means as follows:

| Column heading | Explanation |
|----------------------------|---------------------------------------------------------------|
| Plex/indiv. partition type | A short description of the object counted. |
| | " k -plex" means all k -plexes including any indivisible. |
| | "indiv k -plex" means only indivisible k -plexes. |
| | A type is always an indivisible partition type. |
| Exist num. of species | The number of species in which the object exists. |
| Maximum | The maximum, over all of the species of order n , |
| | of the count of the object in a species. |
| Minimum | Similarly, the minimum count over all species. |
| Non-zero minimum | Similarly, the minimum count over all species when |
| | restricted to species in which the object exists. |
| Mean species to 3 s.f. | Arithmetic mean to 3 significant figures where |
| | each species representative has equal weight. |
| Weighted mean to 3 s.f. | Similarly, where each species has weight proportional |
| | to the number of latin squares in the species. |
| | Equivalently, the expected value for a latin square |
| | of order n that is chosen uniformly at random. |

Within all tables, notation for partition types is abbreviated by omitting both parentheses and separating commas. This is unambiguous since the order is less than 10.

For each order $n \in \{6, 7, 8\}$ we will table enumerative data for species representatives of all group tables. In these tables the entry in the group column may contain a mark '*'. This mark indicates that the specified group achieves the maximum value, over all species of order n, for the corresponding "Plex/indiv. partition type".

4.5.2 Species of order less than 9

We vary slightly from the general table layout for very small orders n = 3, 4, 5 to give a complete count for each species. In the trivial cases $n \in \{1, 2\}$, clearly $\kappa(n) = n$.

Data notes: 1 species of order n = 3. The unique species is of the group table of \mathbb{Z}_3 . It contains exactly three 1-plexes and has a (unique) 1-partition. There are no indivisible 2-plexes so $\kappa(3) = 1$.

Data notes: 2 species of order n = 4. The two species are the groups \mathbb{Z}_4 and $\mathbb{Z}_2 \oplus \mathbb{Z}_2$. Their data is in Table 4.1. The species of $\mathbb{Z}_2 \oplus \mathbb{Z}_2$ and \mathbb{Z}_3 are the only known species with no set of $\lfloor n/2 \rfloor$ parallel indivisible 2-plexes. Neither species of order 4 has an indivisible partition of type $(1^2, 2)$. The largest indivisible plex is a 2-plex, thus $\kappa(4) = 2$.

Data notes: 2 species of order n = 5. In Table 4.2 we present data on the unique group \mathbb{Z}_5 and the single other non-group species. The largest indivisible plex is a 3-plex found in the non-group table, hence $\kappa(5) = 3$. Neither species has an indivisible partition of type $(1^2, 3)$ or $(1^3, 2)$.

Table 4.1: Plexes and indivisible partitions in species of order n = 4.

| Plex/indivisible | S | Species |
|------------------|----------------|------------------------------------|
| partition type | \mathbb{Z}_4 | $\mathbb{Z}_2 \oplus \mathbb{Z}_2$ |
| 1-plex | 0 | 8 |
| 2-plex | 12 | 12 |
| indiv 2-plex | 12 | 0 |
| 2^{2} | 6 | 0 |
| 1^4 | 0 | 2 |

Table 4.2: Plexes and indivisible partitions in species of order n = 5.

| Plex/indivisible | | Species |
|------------------|----------------|-----------|
| partition type | \mathbb{Z}_5 | Non-group |
| 1-plex | 15 | 3 |
| 2-plex | 130 | 30 |
| indiv 2-plex | 100 | 30 |
| indiv 3-plex | 0 | 12 |
| 23 | 0 | 12 |
| 12^{2} | 150 | 9 |
| 1^{5} | 3 | 0 |

Table 4.3: Categories of order n = 6.

| Category | Dis | tinct | Number of | | | | |
|----------|--------------|--------------|--------------|--------------|--------------|--------------|---------|
| | 24 | 3^{2} | 123 | $1^{3}3$ | 1^22^2 | $1^{4}2$ | species |
| 1 | √ | √ | √ | Х | ✓ | √ | 1 |
| 2 | \checkmark | \checkmark | \checkmark | Х | \checkmark | × | 1 |
| 3 | \checkmark | \checkmark | X | X | × | Х | 1 |
| 4 | X | \checkmark | \checkmark | Х | \checkmark | \checkmark | 1 |
| 5 | X | \checkmark | \checkmark | Х | \checkmark | × | 1 |
| 6 | × | \checkmark | × | × | × | × | 1 |
| 7 | X | X | \checkmark | \checkmark | \checkmark | \checkmark | 1 |
| 8 | x | x | \checkmark | \checkmark | × | \checkmark | 1 |
| 9 | х | X | Х | Х | × | Х | 4 |

Data notes: 12 species of order n = 6. Integer partitions of 6 identify 11 types of indivisible partitions to consider. No species possesses a 1-partition (see Theorem 1.16) but every species does possess an indivisible 2-partition. No species of order 6 possesses an indivisible 5-plex and there are no indivisible $(1^2, 4)$ -partitions. Thus 4 types of indivisible partition, $(6), (1, 5), (1^2, 4)$ and (1^6) never occur and 1 type is common to all species. We found that 6 types occur in some species but not all, so these are the distinctive types. In particular, we record that $\kappa(6) = 4$.

Considering the six distinctive types of indivisible partitions, we found as shown in Table 4.3, that the latin squares of order 6 fall into 9 different categories. The four species of category 9, including the two group tables; \mathbb{Z}_6 and D_6 , possess no odd plexes and do not have an indivisible 4-plex. They are the only known species of latin squares of order n > 2 that possess only one type of indivisible partition. Three species, categories 1, 2 and 3, possess an indivisible 4-plex. Each of these three species has an indivisible (2,4)-partition. The example (4.28) is in category 2.

| Plex/ | Exist | Maximum | Minimum | Non-zero | Mean | Weighted |
|----------------|---------|---------|---------|----------|---------------------|---------------------|
| indivisible | num. of | | | minimum | species | mean |
| partition type | species | | | | to 3 s.f. | to 3 s.f. |
| 1-plex | 6 | 32 | 0 | 8 | 7.33 | 6.86 |
| 2-plex | 12 | 1539 | 239 | 239 | 490 | 303 |
| 3-plex | 8 | 1792 | 0 | 512 | 523 | 615 |
| indiv 2-plex | 12 | 1539 | 195 | 195 | 459 | 289 |
| indiv 3-plex | 8 | 640 | 0 | 200 | 310 | 422 |
| indiv 4-plex | 3 | 72 | 0 | 12 | 8.67 | 12.4 |
| 24 | 3 | 72 | 0 | 12 | 8.67 | 12.4 |
| 3^{2} | 6 | 320 | 0 | 88 | 90.7 | 141 |
| 123 | 6 | 696 | 0 | 144 | 147 | 158 |
| 2^3 | 12 | 5949 | 131 | 131 | 917 | 283 |
| $1^{3}3$ | 2 | 72 | 0 | 40 | 9.33 | 1.84 |
| 1^22^2 | 5 | 144 | 0 | 4 | 16.3 | 15.2 |
| $1^{4}2$ | 4 | 56 | 0 | 2 | 7.50 | 1.99 |

Table 4.4: Plexes and indivisible partitions in species of order n = 6.

Table 4.5: Plexes and indivisible partitions in groups of order n = 6.

| Plex/indivisible | Species |
|------------------|----------------------|
| partition type | \mathbb{Z}_6 D_6 |
| indiv 2-plex | 1539* 567 |
| 2^{3} | 5949*945 |

Every species with a 3-plex has an indivisible 3-plex, however two of these species, categories 7 and 8, have no indivisible 3-partition. Indivisible partitions of type $(1^4, 2)$ and (1, 2, 3) were found to occur if and only if a species has enough disjoint transversals. Enumerative data for the species of order n = 6 is summarised in Table 4.4. Enumerative data for the two species of group tables of order n = 6 is in Table 4.5.

Example order n = 6: an indivisible 4-plex. Shading identifies an indivisible (2,4)-partition in (4.28). This latin square, from category 2, is the smallest member of the latin family \mathcal{Q}_n . Lemma 2.1 can be used to explain that the 4-plex is indivisible.

| Category | Di | stin | ctiv | e ind | ivisib | le pa | rtitio | n ty | ре | Number of |
|----------|----------|--------------|-----------------------|--------------|--------------|--------------|--------------|--------------|--------------|-----------|
| | 25 | 34 | 124 | $1^{3}4$ | $1^{2}23$ | $1^{4}3$ | $1^3 2^2$ | $1^{5}2$ | 1^7 | species |
| 1 | √ | \checkmark | \checkmark | х | √ | х | √ | х | х | 1 |
| 2 | X | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | 2 |
| 3 | X | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | x | 33 |
| 4 | X | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | × | X | 62 |
| 5 | X | \checkmark | \checkmark | \checkmark | \checkmark | × | \checkmark | × | X | 9 |
| 6 | X | \checkmark | \checkmark | \checkmark | \checkmark | × | × | × | X | 2 |
| 7 | X | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | 1 |
| 8 | X | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | X | 10 |
| 9 | X | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | 1 |
| 10 | X | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | × | X | 6 |
| 11 | X | \checkmark | \checkmark | × | \checkmark | × | \checkmark | × | X | 12 |
| 12 | х | \checkmark | \checkmark | × | \checkmark | × | × | × | x | 5 |
| 13 | X | \checkmark | \checkmark | × | x | × | × | × | x | 1 |
| 14 | X | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | × | \checkmark | 1 |
| 15 | Х | х | х | х | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | 1 |

Table 4.6: Categories of order n = 7.

To see that the 2-plex is indivisible consider the entries marked *.

$$\begin{pmatrix}
5 & 2 & 4 & 1 & 3 & 0 \\
2 & 4 & 0 & 3_* & 5_* & 1 \\
4 & 0 & 2 & 5_* & 1 & 3 \\
1 & 3 & 5 & 0 & 2 & 4 \\
3 & 5 & 1 & 4 & 0 & 2 \\
0 & 1 & 3 & 2 & 4 & 5
\end{pmatrix}$$
(4.28)

Data notes: 147 species of order n = 7. There are 15 partition types to be checked. We found that no species has an indivisible partition of type (7), (1,6) or $(1^2,5)$. Every species possesses an indivisible partition of type $(1,3^2)$, $(2^2,3)$ and $(1,2^3)$. There are 9 distinctive types of indivisible partitions and these lead to the 15 different categories shown in Table 4.6. Precisely one species, category 1, has an indivisible (2,5)-partition and it is illustrated below in (4.29). Hence, $\kappa(7) = 5$.

Evidently, an indivisible 5-plex of order 7 is rare, however all but one species possesses an indivisible 4-plex. The exception is the species of the unique group \mathbb{Z}_7 (category 15). All but two species possess an indivisible (3, 4)-partition. The exceptions are the Cayley table of \mathbb{Z}_7 and the Steiner quasigroup (see [40, p24]) of order 7 (category 14). In Table 4.7 we summarise enumerative data for order 7. In Table 4.8 we give data for the group \mathbb{Z}_7 . Wanless [150] observed that \mathbb{Z}_7 achieves the maximum number of k-plexes for each k. We see by Table 4.8 that it achieves the maximum value for

Table 4.7: Plexes and indivisible partitions in species of order n=7.

| Plex/ | Exist | Maximum | Minimum | Non-zero | Mean | Weighted |
|----------------|---------|---------|---------|----------|---------------------|---------------------|
| indivisible | num. of | | | minimum | species | mean |
| partition type | species | | | | to 3 s.f. | to 3 s.f. |
| 1-plex | 147 | 133 | 3 | 3 | 23.0 | 20.4 |
| 2-plex | 147 | 23184 | 2676 | 2676 | 3500 | 3330 |
| 3-plex | 147 | 310198 | 42731 | 42731 | 47200 | 45100 |
| indiv 2-plex | 147 | 19404 | 2676 | 2676 | 3370 | 3240 |
| indiv 3-plex | 147 | 54880 | 17360 | 17360 | 35900 | 36400 |
| indiv 4-plex | 146 | 10284 | 0 | 328 | 3020 | 3160 |
| indiv 5-plex | 1 | 4 | 0 | 4 | 0.0272 | 0.0125 |
| 25 | 1 | 4 | 0 | 4 | 0.0272 | 0.0125 |
| 34 | 145 | 10212 | 0 | 136 | 2460 | 2600 |
| 124 | 146 | 2292 | 0 | 72 | 596 | 591 |
| 13^{2} | 147 | 29351 | 5996 | 5996 | 13500 | 12900 |
| $2^{2}3$ | 147 | 566538 | 28728 | 28728 | 64200 | 62500 |
| $1^{3}4$ | 109 | 672 | 0 | 1 | 9.74 | 4.17 |
| $1^{2}23$ | 146 | 120540 | 0 | 192 | 5010 | 3290 |
| 12^{3} | 147 | 976864 | 1442 | 1442 | 13100 | 5580 |
| $1^{4}3$ | 117 | 7546 | 0 | 1 | 74.1 | 14.7 |
| $1^{3}2^{2}$ | 139 | 189042 | 0 | 3 | 1630 | 205 |
| $1^{5}2$ | 47 | 9702 | 0 | 1 | 70.2 | 2.14 |
| 1^7 | 6 | 635 | 0 | 1 | 4.43 | 0.0202 |

| Plex/indivisible | Species |
|------------------|----------------|
| partition type | \mathbb{Z}_7 |
| 1-plex | 133* |
| 2-plex | 23184* |
| 3-plex | 310198* |
| indiv 2-plex | 19404* |
| indiv 3-plex | 54880* |
| 13^{2} | 29351* |
| $2^{2}3$ | 566538* |
| $1^{2}23$ | 120540* |
| 12^{3} | 976864* |
| $1^{4}3$ | 7546* |
| $1^{3}2^{2}$ | 189042* |
| $1^{5}2$ | 9702* |
| 1^7 | 635* |

Table 4.8: Plexes and indivisible partitions in group of order n=7.

every type of indivisible plex and partition that is has.

Example order n = 7: an indivisible 5-plex. Shaded entries in the left hand example in (4.29) identify an indivisible (2,5)-partition of the unique species of order 7 possessing an indivisible 5-plex. This species is in category 1. It is symmetric and has automorphism (34)(56). Together, these generate a paratopism group of order 4. It has exactly four indivisible (2,5)-partitions, which comprise a single orbit of the paratopism group. It has 23 transversals, 1 subsquare of order 3 and 14 intercalates. On the right, the shading identifies an indivisible (3,4)-partition in the same species.

$$\begin{pmatrix}
0 & 1 & 2 & 3 & 4 & 5 & 6 \\
1 & 2 & 0 & 5 & 6 & 4 & 3 \\
2 & 0 & 1 & 6 & 5 & 3 & 4 \\
3 & 5 & 6 & 4 & 2 & 1 & 0 \\
4 & 6 & 5 & 2 & 3 & 0 & 1 \\
5 & 4 & 3 & 1 & 0 & 6 & 2 \\
6 & 3 & 4 & 0 & 1 & 2 & 5
\end{pmatrix}$$

$$\begin{pmatrix}
0 & 1 & 2 & 3 & 4 & 5 & 6 \\
1 & 2 & 0 & 5 & 6 & 4 & 3 \\
2 & 0 & 1 & 6 & 5 & 3 & 4 \\
3 & 5 & 6 & 4 & 2 & 1 & 0 \\
4 & 6 & 5 & 2 & 3 & 0 & 1 \\
5 & 4 & 3 & 1 & 0 & 6 & 2 \\
6 & 3 & 4 & 0 & 1 & 2 & 5
\end{pmatrix}$$

$$(4.29)$$

Data notes: 283657 species of order n = 8. There are 22 types to be considered. No species has an indivisible partition of type $(8), (1,7), (1^2,6)$ or (2,6). The other 18 types are distinctive partition types. We found that 4600 species possess an indivisible 5-plex. Hence, $\kappa(8) = 5$.

The categories of order 8 are tabled in two parts according to whether or not the squares contain an indivisible 5-plex. Table 4.9 collates the 4600 species with an

Table 4.9: Categories of order n=8 for species possessing an indivisible 5-plex.

| Cat. | | | | I | Distino | ctive in | ndivisi | ble p | artitio | on typ | e | | | | Num. of |
|------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|---------|
| | 35 | 125 | 134 | $1^{3}5$ | 1^224 | 1^23^2 | $12^{2}3$ | $1^{4}4$ | $1^{3}23$ | 1^22^3 | $1^{5}3$ | 1^42^2 | $1^{6}2$ | 1^{8} | species |
| 1 | √ | √ | √ | \checkmark | √ | \checkmark | х | 2 |
| 2 | \checkmark | × | X | 2 |
| 3 | \checkmark | \checkmark | \checkmark | × | \checkmark | 3 |
| 4 | \checkmark | \checkmark | \checkmark | × | \checkmark | Х | 295 |
| 5 | \checkmark | \checkmark | \checkmark | × | \checkmark | × | Х | 216 |
| 6 | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | × | \checkmark | × | Х | 29 |
| 7 | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | x | X | 1 |
| 8 | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | 1 |
| 9 | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | × | \checkmark | x | X | 24 |
| 10 | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | × | × | × | Х | 11 |
| 11 | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | × | × | \checkmark | × | × | x | X | 1 |
| 12 | \checkmark | × | \checkmark | × | 2 |
| 13 | \checkmark | × | \checkmark | x | X | 1 |
| 14 | \checkmark | × | \checkmark | × | \checkmark | 26 |
| 15 | \checkmark | × | \checkmark | × | \checkmark | X | 2221 |
| 16 | \checkmark | × | \checkmark | × | \checkmark | × | Х | 1245 |
| 17 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | × | \checkmark | X | × | 50 |
| 18 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | × | × | x | X | 2 |
| 19 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | 1 |
| 20 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | X | \checkmark | 1 |
| 21 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | × | Х | 1 |
| 22 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | 1 |
| 23 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | × | \checkmark | × | \checkmark | 4 |
| 24 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | × | \checkmark | X | × | 35 |
| 25 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | × | × | × | Х | 9 |
| 26 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | × | × | × | X | × | 1 |
| 27 | \checkmark | × | \checkmark | × | \checkmark | \checkmark | \checkmark | × | × | \checkmark | × | × | × | Х | 3 |
| 28 | \checkmark | × | × | × | × | × | × | × | × | × | × | × | X | × | 6 |
| 29 | × | \checkmark | X | 2 |
| 30 | x | \checkmark | \checkmark | × | \checkmark | 5 |
| 31 | × | \checkmark | \checkmark | × | \checkmark | x | 250 |
| 32 | × | \checkmark | \checkmark | × | \checkmark | × | X | 142 |
| 33 | × | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | × | \checkmark | × | x | 2 |
| 34 | × | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | × | \checkmark | × | X | 3 |
| 35 | × | × | \checkmark | х | 2 |

Table 4.10: Categories of order n=8 for species with no indivisible 5-plex.

| Cat. | | | | D | istinct | ive in | ndivis | ible j | partit | ion ty | pe | | | | Num. of |
|------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|---------|
| | 4^2 | 134 | $2^{2}4$ | 23^{2} | $1^{2}24$ | 1^23^2 | $12^{2}3$ | $1^{4}4$ | $1^{3}23$ | 1^22^3 | $1^{5}3$ | 1^42^2 | $1^{6}2$ | 1^{8} | species |
| 36 | \checkmark | √ | √ | √ | √ | √ | √ | \checkmark | √ | √ | \checkmark | √ | \checkmark | \checkmark | 1866 |
| 37 | \checkmark | X | 206706 |
| 38 | \checkmark | × | \checkmark | 31 |
| 39 | \checkmark | × | × | 69708 |
| 40 | \checkmark | × | \checkmark | \checkmark | \checkmark | 1 |
| 41 | \checkmark | × | \checkmark | \checkmark | × | 6 |
| 42 | \checkmark | × | \checkmark | × | × | 408 |
| 43 | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | 41 |
| 44 | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | × | 15 |
| 45 | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | × | \checkmark | 4 |
| 46 | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | × | × | 14 |
| 47 | \checkmark | × | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | 15 |
| 48 | \checkmark | × | \checkmark | \checkmark | × | \checkmark | \checkmark | × | 4 |
| 49 | \checkmark | × | \checkmark | \checkmark | × | \checkmark | × | \checkmark | 19 |
| 50 | \checkmark | × | \checkmark | \checkmark | × | \checkmark | × | × | 159 |
| 51 | \checkmark | × | \checkmark | \checkmark | × | × | × | X | 25 |
| 52 | \checkmark | × | × | \checkmark | × | × | × | × | 3 |
| 53 | \checkmark | × | \checkmark | \checkmark | × | × | × | × | × | × | × | × | × | X | 26 |
| 54 | X | \checkmark | 1 |
| 55 | X | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | × | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | 2 |
| 56 | X | × | \checkmark | × | × | х | x | × | × | × | × | × | × | X | 1 |
| 57 | × | X | X | ✓ | Х | ✓ | ✓ | X | ✓ | ✓ | ✓ | ✓ | ✓ | \checkmark | 2 |

Table 4.11: Plexes and indivisible partitions in species of order n=8.

| Plex/ | Exist | Maximum | Minimum | Non-zero | Mean | Weighted |
|----------------|---------|------------------|----------|-----------------------------------------|-----------|-----------|
| indivisible | num. of | TVICAZEIIII GIII | William | minimum | species | mean |
| partition type | species | | | 111111111111111111111111111111111111111 | to 3 s.f. | to 3 s.f. |
| 1-plex | 283624 | 384 | 0 | 8 | 61.1 | 61.0 |
| 2-plex | 283657 | 465976 | 23592 | 23592 | 57600 | 57500 |
| 3-plex | 283656 | 28979840 | 0 | 2211280 | 3630000 | 3630000 |
| 4-plex | 283657 | 114209676 | 10890348 | 10890348 | 14300000 | 14300000 |
| indiv 2-plex | 283657 | 460096 | 23457 | 23457 | 56800 | 56800 |
| indiv 3-plex | 283656 | 14388224 | 0 | 1908032 | 3170000 | 3170000 |
| indiv 4-plex | 283655 | 10098036 | 0 | 5376 | 1410000 | 1410000 |
| indiv 5-plex | 4600 | 19168 | 0 | 1 | 0.265 | 0.137 |
| 35 | 4194 | 18356 | 0 | 1 | 0.240 | 0.123 |
| 4^2 | 283651 | 2679882 | 0 | 48 | 67100 | 66800 |
| 125 | 989 | 828 | 0 | 1 | 0.0261 | 0.0149 |
| 134 | 283622 | 4809618 | 0 | 4608 | 812000 | 812000 |
| $2^{2}4$ | 283655 | 9249292 | 0 | 25344 | 2380000 | 2380000 |
| 23^{2} | 283656 | 1901046272 | 0 | 34296746 | 89000000 | 88800000 |
| $1^{3}5$ | 11 | 2 | 0 | 1 | 0.0000458 | 0.0000356 |
| $1^{2}24$ | 283622 | 500568 | 0 | 1600 | 61100 | 61100 |
| 1^23^2 | 283624 | 70983936 | 0 | 36232 | 1150000 | 1140000 |
| $12^{2}3$ | 283624 | 2101675776 | 0 | 1728990 | 13200000 | 13100000 |
| 2^4 | 283657 | 3402475552 | 372045 | 372045 | 6450000 | 6350000 |
| $1^{4}4$ | 283224 | 8646 | 0 | 1 | 130 | 130 |
| $1^{3}23$ | 283617 | 91225856 | 0 | 768 | 116000 | 113000 |
| $1^2 2^3$ | 283623 | 736603392 | 0 | 7710 | 339000 | 326000 |
| $1^{5}3$ | 282808 | 609792 | 0 | 1 | 165 | 148 |
| $1^4 2^2$ | 283569 | 36365952 | 0 | 2 | 2740 | 2150 |
| $1^{6}2$ | 211470 | 634368 | 0 | 1 | 14.2 | 4.12 |
| 18 | 2024 | 70272 | 0 | 1 | 0.930 | 0.0147 |

indivisible 5-plex. Each of these species possesses an indivisible partition of types (4^2) , $(2^2, 4)$, $(2, 3^2)$ and (2^4) .

In Table 4.10 we report on the majority; that is the 279057 species with no indivisible 5-plex. All of these species possess an indivisible (2⁴)-partition. Category 37 shows that most species do not have an indivisible 5-plex or a 1-partition but do have an indivisible partition for every type not ruled out by the aforementioned constraints. All but two species possess an indivisible 4-plex. The exceptions are the species of the group tables of $\mathbb{Z}_4 \oplus \mathbb{Z}_2$ and $\mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_2$ which form category 57. All but one species possesses an indivisible 3-plex. The exception, category 56, is the species of \mathbb{Z}_8 which has no odd plexes by Theorem 1.27. It is unique with respect to this property. In Table 4.11 we summarise the enumerative data for order n = 8.

Example order n=8: Species with most indivisible 5-plexes. The mean number of indivisible 5-plexes in Table 4.11 reflects that, at this order, indivisible 5-plexes are uncommon. Indeed, some 2783 of the 4600 species possessing an indivisible 5-plex have only one of them. In this sense, the latin square in (4.30), of category 9, is extreme. It uniquely has the maximum number, that is 19168 indivisible 5-plexes. On the left, one indivisible (3,5)-partition is marked. On the right, the 4 intercalates marked do not intersect any of the 16 transversals in the square.

$$\begin{pmatrix} 0 & 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 1 & 0 & 3 & 2 & 5 & 4 & 7 & 6 \\ 2 & 3 & 1 & 4 & 6 & 7 & 5 & 0 \\ 3 & 2 & 4 & 1 & 7 & 6 & 0 & 5 \\ 4 & 5 & 6 & 7 & 2 & 0 & 1 & 3 \\ 5 & 4 & 7 & 6 & 0 & 2 & 3 & 1 \\ 6 & 7 & 5 & 0 & 1 & 3 & 2 & 4 \\ 7 & 6 & 0 & 5 & 3 & 1 & 4 & 2 \end{pmatrix}$$

$$\begin{pmatrix} 0 & 1 & 2 & 3 & 4 & 5 & 6 & 7 \\ 1 & 0 & 3 & 2 & 5 & 4 & 7 & 6 \\ 2 & 3 & 1 & 4 & 6 & 7 & 5 & 0 \\ 3 & 2 & 4 & 1 & 7 & 6 & 0 & 5 \\ 4 & 5 & 6 & 7 & 2 & 0 & 1 & 3 \\ 5 & 4 & 7 & 6 & 0 & 2 & 3 & 1 \\ 6 & 7 & 5 & 0 & 1 & 3 & 2 & 4 \\ 7 & 6 & 0 & 5 & 3 & 1 & 4 & 2 \end{pmatrix}$$

$$(4.30)$$

Group tables of order n=8. In Table 4.12, we present data on the number of plexes, indivisible plexes and indivisible partitions for the five group tables of order n=8. None of the five species possesses an indivisible 5-plex or an indivisible $(1^4, 4)$ -partition. The Abelian groups in Table 4.12 account for many of the maximum values recorded in Table 4.11. The absence of odd plexes in \mathbb{Z}_8 is explained by Theorem 1.27. Each of the non-cyclic groups of order 8 has 384 transversals, as explained by Bedford and Whitaker [11]. Theorem 1.59, by Balasubramanian, states that a latin square of even order has an even number of transversals. However, as mentioned in Section 1.14.2, Balasubramanian's statement does not generalise in an obvious way to k-plexes. Nevertheless, it is striking that every value in Table 4.12 is an even number. This is explained by the next result, due to Ian Wanless.

Lemma 4.31. For integers $n \ge k \ge 1$, let m be the greatest common divisor of n and k. Let L be the Cayley table of a group (G, \star) of order n. The number of k plexes in L is a multiple of n/m. Likewise, the number of indivisible k-plexes in L is a multiple of n/m.

| Plex/indivisible | | | Group table | | |
|------------------|----------------|------------------------------------|--------------------------------------------------------|-----------|-----------|
| partition type | \mathbb{Z}_8 | $\mathbb{Z}_4 \oplus \mathbb{Z}_2$ | $\mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_2$ | D_8 | Q_8 |
| 1-plex | 0 | 384* | 384* | 384* | 384* |
| 2-plex | 460096 | 462120 | 465976* | 240008 | 236744 |
| 3-plex | 0 | 28979840* | 28907648 | 14590848 | 14625792 |
| 4-plex | 113925996 | 114021324 | 114209676* | 57461196 | 57353676 |
| indiv 2-plex | 460096* | 432744 | 435736 | 210344 | 207272 |
| indiv 3-plex | 0 | 11326720 | 11203584 | 5707008 | 5831424 |
| indiv 4-plex | 28416 | 0 | 0 | 6144 | 6912 |
| 4^2 | 0 | 0 | 0 | 768 | 0 |
| 134 | 0 | 0 | 0 | 6144 | 9216 |
| $2^{2}4$ | 420096 | 0 | 0 | 26112 | 32256 |
| 23^{2} | 0 | 1135096064 | 1143668736 | 278490880 | 283778304 |
| $1^{2}24$ | 0 | 0 | 0 | 4608 | 4608 |
| 1^23^2 | 0 | 70983936* | 66060288 | 36537600 | 39034368 |
| $12^{2}3$ | 0 | 2101675776* | 2063308800 | 499349760 | 506898432 |
| 2^4 | 3402475552* | 2717863186 | 2906655710 | 333162978 | 303664290 |
| $1^{3}23$ | 0 | 91225856* | 83091456 | 41943040 | 42633216 |
| 1^22^3 | 0 | 719491968 | 736603392* | 169825152 | 157148160 |
| $1^{5}3$ | 0 | 569856 | 344064 | 555264 | 609024 |
| 1^42^2 | 0 | 28231296 | 36365952* | 15490560 | 13766784 |
| $1^{6}2$ | 0 | 272384 | 634368* | 373376 | 233472 |
| 1^{8} | 0 | 23040 | 70272* | 33408 | 32256 |

Table 4.12: Plexes and indivisible partitions in groups of order n = 8.

Proof. Let Γ be the group of right translations acting on the k-plexes of L by $K \mapsto \{(r, c \star g, s \star g) : (r, c, s) \in K\}$ for each $g \in G$. Let K be a k-plex of L and Γ_K the stabiliser of K under the action of Γ . Then Γ_K acts semi-regularly on the k entries in the first row of K, which means $|\Gamma_K|$ divides k. The orbit of K under Γ has cardinality $n/|\Gamma_K|$, so $|\Gamma_K|$ divides n and hence also divides m. As this is true for arbitrary K, the total number of k-plexes in L must be a multiple of n/m. The same is true for indivisible plexes since Γ preserves indivisibility of plexes.

4.5.3 Species of order 9

Data: 19270853541 species of order n = 9. As mentioned in Section 1.14.1, we used a program described in [119] to generate isotopy class representatives of order 9. This program was also used in [118] in confirming Conjecture 1.1 for order 9, thereby showing that $\kappa(9) < 9$. We found, in the following manner, that $\kappa(9) < 8$. First, we obtained a list of isotopy class representatives which possess a transversal that

has no parallel transversal. In other words, we identified all isotopy classes with a (1,8)-partition in which the 8-plex contains no transversal. This list reduced to a list of 36007 species representatives, which we shall refer to as Ω . Subsequently, we established that for each species in Ω , a 2-plex is contained in those 8-plexes with no transversal, thereby confirming that $\kappa(9) < 8$. As we have just mentioned, each species of order 9 has either 2 parallel transversals or possesses a 2-plex. Hence, Conjecture 1.2 is true for order 9.

Unfortunately, we were unable to perform an exhaustive search for an indivisible 7-plex. In a small sample, we found that many species have well over a million 2-plexes so it is hard to test a lot of species. Although finding all 2-plexes by our programs is not viable, we did investigate the possibility of an indivisible $(1^2, 7)$ -partition, a much easier computation. The computation was made a little more efficient by adjusting the latin square generator to output species representatives, rather than isotopy representatives. The adjustment was made by Brendan McKay who also contributed to a program designed to record species representatives with a $(1^2, 7)$ -partition in which the 7-plex contains no transversal. We thus recorded, in a list which we call Ψ , precisely 10270 such species representatives. Exactly 2093 species representatives appear in both Ψ and Ω .

Among the species in Ω and Ψ we found that there are no indivisible 7-plexes. We did not search all of them for an indivisible 6-plex. However, we found one species in Ω that has two indivisible 6-plexes. Thus, $\kappa(9) \in \{6,7\}$. We also conclude that if $\kappa(9) \neq 6$, then there exists an indivisible (2,7)-partition.

Example order n = 9: a species of Ω with indivisible 6-plexes. We illustrate in (4.31) the two, slightly different, indivisible 6-plexes which prove that $\kappa(9) \ge 6$. In each case, the 3-plex marked divides in two different ways to give indivisible (1, 2, 6)-partitions. The latin square has a trivial paratopy group so the plexes are in different orbits.

In both cases, equation (4.21) shows that the 6-plex has no odd plexes. (The bold entries comprise the set D, as defined in Section 4.4.) We leave open explanation of the absence of even plexes in these 6-plexes.

$$\begin{pmatrix} 0 & 1 & 2 & 3 & 7 & 6 & 5 & 8 & 4 \\ 1 & 0 & 3 & 2 & 6 & 5 & 4 & 7 & 8 \\ 2 & 3 & 0 & 1 & 4 & 8 & 7 & 5 & 6 \\ 3 & 2 & 1 & 0 & 8 & 7 & 6 & 4 & 5 \\ 4 & 6 & 7 & 8 & 5 & 3 & 0 & 2 & 1 \\ 5 & 8 & 6 & 7 & 2 & 4 & 1 & 3 & 0 \\ 6 & 7 & 5 & 4 & 3 & 1 & 8 & 0 & 2 \\ 7 & 4 & 8 & 5 & 1 & 0 & 2 & 6 & 3 \\ 8 & 5 & 4 & 6 & 0 & 2 & 3 & 1 & 7 \end{pmatrix}$$

$$\begin{pmatrix} 0 & 1 & 2 & 3 & 7 & 6 & 5 & 8 & 4 \\ 1 & 0 & 3 & 2 & 6 & 5 & 4 & 7 & 8 \\ 2 & 3 & 0 & 1 & 4 & 8 & 7 & 5 & 6 \\ 3 & 2 & 1 & 0 & 8 & 7 & 6 & 4 & 5 \\ 4 & 6 & 7 & 8 & 5 & 3 & 0 & 2 & 1 \\ 5 & 8 & 6 & 7 & 2 & 4 & 1 & 3 & 0 \\ 6 & 7 & 5 & 4 & 3 & 1 & 8 & 0 & 2 \\ 7 & 4 & 8 & 5 & 1 & 0 & 2 & 6 & 3 \\ 8 & 5 & 4 & 6 & 0 & 2 & 3 & 1 & 7 \end{pmatrix}$$

$$(4.31)$$

The subsquare of order 4 in L of (4.31) is the species of $\mathbb{Z}_2 \oplus \mathbb{Z}_2$ so it has 8 distinct

transversals. We can form a transversal, $T = T_A \cup D$, where T_A is a transversal in $A = \mathbb{Z}_2 \oplus \mathbb{Z}_2$, and D is the set marked by the bold entries of L. Then, $L \setminus T$ contains no odd plex by (4.21). We discovered that each species of Ω has similar structure.

Lemma 4.32. If L is a latin square of order 9, then the following conditions are equivalent.

- 1. L has a transversal that has no parallel transversal;
- 2. L has a subsquare that is paratopic to $\mathbb{Z}_2 \oplus \mathbb{Z}_2$;
- 3. L has exactly 8 transversals that have no parallel transversal and each of them is the union of a transversal of a subsquare A, of order 4, and its corresponding set D, as defined in Section 4.4;
- 4. L has an 8-plex that contains no odd plex;
- 5. L belongs to one of the 36007 species in Ω .

Proof. Direct computation showed that $5 \implies 3$. There is only one species of order four that has transversals, so $3 \implies 2$. The implication $2 \implies 4$ follows by (4.21). The fact that $4 \implies 1$ is immediate. Likewise $1 \implies 5$, by the definition of Ω . \square

As an independent check of our computations, we built all species of order 9 with a subsquare paratopic to $\mathbb{Z}_2 \oplus \mathbb{Z}_2$ and confirmed that there are 36007 such species. It would be interesting to know whether the behaviour seen in Lemma 4.32 is a one-off, or whether analogous statements hold for some larger $n \equiv 1 \mod 4$.

Data notes: 6 species of interest of order n = 9. For a small selection of species of order 9, as next listed, we present in Table 4.13 the number of plexes and indivisible plexes, and note the existence or otherwise of indivisible partitions. The indivisible partitions with two parts are counted.

Species Source: interest

```
\mathbb{Z}_9 : species of group table.

\mathbb{Z}_3 \oplus \mathbb{Z}_3 : species of group table.

A (4.31) : the single species thus far identified with an indivisible 6-plex.

B [118, p281] : unique species with the minimum number of transversals.

C [118, p281] : unique species with the maximum number of intercalates.

D [155] : of an infinite family \mathcal{W}_n which has at most n-3 disjoint transversals.
```

None of the six species possess an indivisible 7-plex or an indivisible partition of type (3,6), $(1^3,6)$ or $(1^4,5)$. The types (9), (1,8) and $(1^2,7)$ were ruled out by the earlier mentioned search of all species of order 9. All of the six species possess an indivisible partition of type (2,3,4), (3^3) , $(1^2,3,4)$, $(1,2^2,4)$, $(1,2,3^2)$, $(2^3,3)$, $(1^3,2,4)$, $(1^3,3^2)$ $(1^2,2^2,3)$, $(1,2^4)$, $(1^4,2,3)$, $(1^3,2^3)$ and $(1^5,2^2)$. We omit from Table 4.13 such partition types for which data is common to all six species.

| Plex/indivisible | | | Spec | ies | | |
|------------------|----------------|----------------------------------|--------------|--------------|--------------|--------------|
| partition type | \mathbb{Z}_9 | $\mathbb{Z}_3\oplus\mathbb{Z}_3$ | A | B | C | D |
| 1-plex | 2025 | 2241 | 176 | 68 | 801 | 415 |
| 2-plex | 11312217 | 11299068 | 1361837 | 1266808 | 1328760 | 983984 |
| 3-plex | 3805235823 | 3805353948 | 416827284 | 422663812 | 419501388 | 383230707 |
| 4-plex | 67338677862 | 67339012014 | 7490644777 | 7482295830 | 7492688982 | 7920057672 |
| indiv 2-plex | 10517040 | 10331712 | 1356971 | 1266052 | 1212120 | 959909 |
| indiv 3-plex | 1572440364 | 1439897904 | 383502764 | 411021388 | 300228264 | 336935606 |
| indiv 4-plex | 12636 | 23328 | 922080208 | 1320329954 | 314944938 | 1944551389 |
| indiv 5-plex | × | × | 1862 | × | × | 14829 |
| indiv 6-plex | × | × | 2 | × | × | х |
| 45 | × | × | 69 | × | × | 3558 |
| 126 | х | х | 4 | х | × | х |
| 135 | х | х | \checkmark | х | × | \checkmark |
| 14^{2} | х | х | \checkmark | \checkmark | \checkmark | \checkmark |
| $2^{2}5$ | х | х | \checkmark | х | × | \checkmark |
| $1^{2}25$ | х | х | \checkmark | х | × | \checkmark |
| $1^{5}4$ | х | х | \checkmark | \checkmark | \checkmark | \checkmark |
| $1^{6}3$ | \checkmark | \checkmark | × | × | \checkmark | \checkmark |
| $1^{7}2$ | \checkmark | \checkmark | × | × | \checkmark | × |
| 1^9 | \checkmark | \checkmark | × | × | \checkmark | × |

Table 4.13: Plexes and indivisible partitions in six species of order n = 9.

Table 4.13 shows that the six species are of different categories and identifies 10 types of distinctive partitions. In Section 5.4 of the next chapter we identify species of order 9 possessing at most 3 disjoint transversals. Thus, partition types $(1^5, 2^2)$ and $(1^4, 2, 3)$ common to the species of Table 4.13, are distinctive. Type $(1^4, 5)$ is also distinctive as we found examples among the species in Ω . In summary, of the 30 types of indivisible partitions to consider, at least 24 types are known to occur and we confirm that at least 13 of those are distinctive types. At least 3 types never occur: (9), (1,8) and $(1^2,7)$.

We leave open the possibility of a latin square of order 9 with any one of the following properties:

- 1. has an indivisible 7-plex, thus an indivisible (2,7)-partition,
- 2. has an indivisible partition of type (3,6) or $(1^3,6)$,
- 3. has no indivisible 4-plex,
- 4. has no indivisible partition of any of the following 11 types: $(2,3,4), (3^3), (1^2,3,4), (1,2^2,4), (1,2,3^2), (2^3,3), (1^3,2,4), (1^3,3^2), (1^2,2^2,3), (1,2^4) \text{ or } (1^3,2^3).$

In (4.32) we show an example of the two species of Table 4.13 that possess an indi-

visible (4,5)-partition. On the left is species A and on the right is species D.

$$\begin{pmatrix} 0 & 1 & 2 & 3 & 7 & 6 & 5 & 8 & 4 \\ 1 & 0 & 3 & 2 & 6 & 5 & 4 & 7 & 8 \\ 2 & 3 & 0 & 1 & 4 & 8 & 7 & 5 & 6 \\ 3 & 2 & 1 & 0 & 8 & 7 & 6 & 4 & 5 \\ 4 & 6 & 7 & 8 & 5 & 3 & 0 & 2 & 1 \\ 5 & 8 & 6 & 7 & 2 & 4 & 1 & 3 & 0 \\ 6 & 7 & 5 & 4 & 3 & 1 & 8 & 0 & 2 \\ 7 & 4 & 8 & 5 & 1 & 0 & 2 & 6 & 3 \\ 8 & 5 & 4 & 6 & 0 & 2 & 3 & 1 & 7 \end{pmatrix}$$

$$\begin{pmatrix} 1 & 0 & 2 & 3 & 4 & 5 & 6 & 7 & 8 \\ 3 & 2 & 1 & 4 & 5 & 6 & 7 & 8 & 0 \\ 2 & 3 & 4 & 5 & 6 & 7 & 8 & 0 & 1 \\ 0 & 4 & 3 & 6 & 7 & 8 & 5 & 1 & 2 \\ 4 & 5 & 6 & 7 & 8 & 0 & 1 & 2 & 3 & 4 \\ 6 & 7 & 8 & 0 & 1 & 2 & 3 & 4 & 5 \\ 7 & 8 & 0 & 1 & 2 & 3 & 4 & 5 & 6 \\ 8 & 1 & 5 & 2 & 3 & 4 & 0 & 6 & 7 \end{pmatrix}$$

$$(4.32)$$

In Table 4.14 we summarise our results for $\kappa(n)$ for the small orders $n \leq 11$. The values $\kappa(10)$ and $\kappa(11)$ are justified by Theorem 4.2 and (4.19), respectively.

Table 4.14: $\kappa(n)$ for $n \leq 11$

| Order n | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | 10 | 11 |
|-------------|---|---|---|---|---|---|---|---|--------|-----|-----|
| $\kappa(n)$ | 1 | 2 | 1 | 2 | 3 | 4 | 5 | 5 | 6 or 7 | ≥ 5 | ≥ 6 |

4.6 Concluding remarks

The data of the last section offers a solid resource for testing and further development of general results about existence of plexes and partitions. As mentioned in Section 1.14, very little theory exists on the number of plexes in latin squares. We mentioned in that section the known general results for k = 1. For k > 1, we know of no general results other than Lemma 4.31 about the number of k-plexes; indivisible or otherwise.

A frustrating open problem is whether or not every latin square of order n>2 contains a non-trivial plex. We emphasise this important open problem, alternatively expressed:

Problem 4.33. Is $\kappa(n) < n$, for all n > 2?

A proof of Conjecture 1.2 would resolve Problem 4.33.

In the next chapter we will report on computations proving that Conjecture 1.3 is true for order $n \leq 9$. The data on indivisible partitions favours the next, slightly stronger, statement.

Conjecture 4.34. Every latin square of order n > 3 has $\lfloor n/2 \rfloor$ parallel indivisible 2-plexes.

Our computations show that few species of order $n \leq 8$ fail to possess an indivisible k-plex or an indivisible (k, n - k)-partition for $k \in \{\lfloor \frac{n}{2} \rfloor, \lceil \frac{n}{2} \rceil\}$. However, for any strictly larger k the incidence of indivisible k-plexes is much rarer. For order n = 9, we have not tested enough species to address the likelihood of an indivisible $\lceil \frac{n}{2} \rceil$ -plex. We leave open whether or not the family \mathcal{H}_n , of Section 4.3.2, possesses an indivisible $\lceil \frac{n}{2} \rceil$ -plex for all odd n > 3. We conjecture on a problem first raised in [65] and which remains open for odd $n \geq 13$. Theorem 4.3 answers the case for even $n \geq 4$.

Conjecture 4.35. For each $n \ge 4$, there exists a latin square of order n with an indivisible $(\lfloor \frac{n}{2} \rfloor, \lceil \frac{n}{2} \rceil)$ -partition.

The next statement is similar in expression to Problem 3.24.

Problem 4.36. For integers a < b < c, does there exist a latin square L with an indivisible a-plex, an indivisible c-plex and a divisible b-plex, but L has no indivisible b-plex?

In contrast to Problem 3.24, Conjecture 1.3 does not immediately imply a parity condition, on either the order of L or on the integers $\{a,b,c\}$, of an example meeting the conditions of Problem 4.36. Our data shows that no example exists among latin squares of order $n \leq 8$.

The techniques of Section 4.2.2 yield many different types of indivisible partitions in larger latin squares. The results of that section leave open the following special case of the problem of Conjecture 1.39.

Problem 4.37. For 4k < n < 5k, can a k-protoplex of order 2k always be extended to a k-plex of order n?

Avoidance of odd plexes in large parts was explained in Section 4.4. Thus, we understand the absence of odd plexes in the indivisible 6-plexes (4.31). We leave open explanation of why they contain no even plex.

Further study of indivisible plexes in \mathbb{Z}_n is due. Our small theoretic contribution in Section 4.3.3 does not explain larger indivisible plexes found in \mathbb{Z}_n for $n \leq 9$. Our data confirms an observation in [150] that the commutative groups tend to have many plexes. Therefore, the absence of indivisible k-plexes, for k > n/2, in those Cayley tables is not so surprising. As a benchmark, if indivisible k-plexes occur in \mathbb{Z}_n then we might reasonably expect other latin squares of order n to also contain indivisible k-plexes.

Whenever k properly divides n, Theorem 4.1 confirms the existence of an indivisible k-partition in some latin square of order n. Few latin squares of small composite order fail to contain such a partition so it would be of interest to identify conditions under which such failure occurs, particularly for odd orders. By Theorem 1.25, examples exist if k is odd and n is even. Also, in Chapter 3 we saw how the number of

odd plexes in a partition can be constrained. A search, of a small sample (100000) of randomly generated latin squares of order 9, found none that fail to contain an indivisible 3-partition. It would be useful to survey the types of indivisible partitions in a larger sample of order 9 and in latin squares of larger order.

In the next chapter, on latin squares with restricted transversals, we will see further behaviour that is not evident among latin squares of small order and their partitions with parts of size 1.

For a given latin square L, define K(L) to be the maximum k such that L has an indivisible k-plex. For large n, can we expect that almost all latin squares of order n will possess almost all types of indivisible partitions other than those types with part sizes near to, or exceeding, K(L)? Among the species of small order, we have not seen a huge variation in K(L). This suggests the following problem.

Problem 4.38. Consider, for fixed but arbitrary n,

 $\min \{K(L) : L \text{ is a latin square of order } n\}.$

How does this minimum value of K(L) behave for larger n?

Chapter 5

Latin squares with restricted transversals

5.1 Introduction

In this chapter we consider several ways in which transversals in latin squares can be restricted.

For a given latin square L of order n, we define:

$$\lambda(L) = \max \{ m : L \text{ possesses a } (1^m, n-m) \text{-partition} \}.$$

Lemma 4.29 showed the following.

Lemma 5.1. For each even $n \ge 6$ and each integer $j \equiv 0 \mod 4$, there exists a latin square $L = \mathcal{V}_{n,j}$ such that $\lambda(L) = j$.

In Section 5.3 we prove our main result which is Theorem 5.2. It proves the existence of arbitrarily large latin squares with $\lambda(L) = 1$.

Theorem 5.2. For all even $n \ge 10$, except perhaps if n is a power of 2, there exists a latin square of order n that has a transversal but every transversal coincides on a single entry.

The possibility of an infinite family of latin squares of odd order with a constant λ value remains an open question.

As mentioned in Section 1.8, a confirmed bachelor square is a latin square with at least one transversal-free entry. A corollary of Theorem 5.2 is a new construction of confirmed bachelor latin squares for orders $n \equiv 3 \mod 4$ when $n \geqslant 11$. Our construction of latin squares with at least one transversal-free entry is a general method. It is described in Section 5.3.

We define for all positive integers n:

$$\mu(n) = \min\{ \lambda(L) : L \text{ is a latin square of order } n \}.$$

Clearly $\mu(n) = 0$ for all even n, thus we are concerned with the case that n is odd. If Conjecture 1.1 is true, then $\mu(n) \ge 1$ for all odd n. A trivial upper bound, $\mu(n) \le n - 2$, follows immediately from the existence of a bachelor latin square of order n. However, a better bound can be derived by considering the number of disjoint transversals in two such constructions in the literature.

Similar to earlier discussion in Section 4.4, the next theorem follows from the reasoning of Mann [117]. Let t be a positive integer.

Theorem 5.3. Assume that L is a latin square of order 4t + 1 such that L contains a subsquare of order 2t. Then $\lambda(L) \leq 2t + 1$.

Theorem 5.3 is useful in Section 5.4 when we consider our data for order 9. For orders n = 4t + 3, the reasoning of Evans [73] also gives the result that $\lambda \leq 2t + 1$ for his construction. Together, these results by Mann and Evans justify the following.

Theorem 5.4. If n is odd and
$$n > 3$$
, then $\mu(n) \leq \frac{1}{2}(n+1)$.

We know of no general argument that improves on the bound given by Theorem 5.4, although it is not tight in the cases where μ is known. In Section 5.4 we report on computations proving that $\mu(9) = 3$. We found that almost all species of order 9 possess a λ value of at least 5. These computations, as we will explain, also prove Conjecture 1.3 for the latin squares of order $n \leq 9$. For order $n \in \{5,7\}$ there is a latin square whose transversals coincide on one entry, hence $\mu(5) = \mu(7) = 1$. For $n \in \{1,3\}$ we have $\mu(n) = n$.

In Section 5.4 we also report on computations concerning the number of transversal-free entries in the species of order $n \leq 9$. Our data supports the view that almost all large latin squares have no transversal-free entries.

For odd n > 3, the existence of a confirmed bachelor latin square of order n as stated in Theorem 1.23 was proved in [155] using the family \mathcal{W}_n . They did not name the family; it is a convenience introduced here. The definition of \mathcal{W}_n is included in Section 2.5.2. As observed in [155], the 3 elements of $\Delta_* \cap r_{n-1} \subset \mathcal{W}_n$ are transversal-free.

We note that for $n \geq 9$, as displayed in (2.28), there are 7 elements in $\Delta_* \setminus (\Delta_2 \cup \Delta_{-2}) \subset \mathcal{W}_n$. It follows from Lemma 2.1 that, for $n \geq 9$, all 7 of these elements are transversal-free. This proves the following.

Theorem 5.5. For all odd $n \ge 9$, there exists a latin square of order n with at least 7 transversal-free entries.

We recorded in Theorem 4.23 that the family \mathcal{H}_n has at least two transversal-free entries.

In Section 5.2 we prove the following theorem using an infinite family in which the proportion of transversal-free entries is asymptotic to one ninth. Theorem 5.6 is due to Ian Wanless.

Theorem 5.6. For all odd $m \ge 3$, there exists a latin square of order 3m that contains an $(m-1) \times m$ latin subrectangle consisting of transversal-free entries.

5.2 Transversals in \mathcal{D}_{3m}

In this section we prove Theorem 5.6 using the family \mathcal{D}_{3m} which is defined in Chapter 2 for all odd $m \geq 3$.

All results in this section, including the construction of the family \mathcal{D}_{3m} , are the work of Ian Wanless. The family \mathcal{D}_{3m} is a generalisation of an interesting example of order 9 which was obtained by the data project reported in the next section.

Suppose that T is a transversal of \mathcal{D}_{3m} . We define x_{ij} to be the number of elements of T of the form ((i,b),(j,d),(e,f)) where b,d and f are arbitrary and e=i+j in \mathbb{Z}_3 . We define y_{ij} to be the number of elements of T of the same form, but where $e \neq i+j$ in \mathbb{Z}_3 . Finally, we let z be the number of elements of T of the form ((0,1),(0,d),(0,f)) where d and f are arbitrary. A number of constraints are immediate from the definition of $\mathcal{D}_{3,m}$. We will make repeated implicit use of the bounds $0 \leq x_{ij} \leq m$, $0 \leq y_{ij} \leq 1$, $0 \leq z \leq 1$ and the fact that $y_{02} = y_{11} = y_{20} = 0$. Moreover, the construction of \mathcal{D}_{3m} forces

$$y_{00} + y_{01} \leqslant 1$$
 $y_{10} + y_{12} \leqslant 1$ $y_{21} + y_{22} \leqslant 1$, (5.1)

$$0 \leqslant x_{00} - z \leqslant m - 2. \tag{5.2}$$

Also, the need for T to include one representative from each row, column and symbol of \mathcal{D}_{3m} implies

$$x_{00} + x_{01} + x_{02} + y_{00} + y_{01} = m, (5.3)$$

$$x_{20} + x_{21} + x_{22} + y_{21} + y_{22} = m, (5.4)$$

$$x_{00} + x_{10} + x_{20} + y_{00} + y_{10} = m, (5.5)$$

$$x_{02} + x_{12} + x_{22} + y_{12} + y_{22} = m, (5.6)$$

$$x_{00} + x_{12} + x_{21} + y_{01} + y_{10} + y_{22} = m, (5.7)$$

$$x_{01} + x_{10} + x_{22} + y_{00} + y_{12} + y_{21} = m. (5.8)$$

Adding (5.3), (5.5) and (5.7), then subtracting (5.4), (5.6) and (5.8), gives

$$3x_{00} - 3x_{22} + y_{00} + 2y_{01} + 2y_{10} - 2y_{12} - 2y_{21} - y_{22} = 0. (5.9)$$

Moreover, Lemma 2.1 necessitates that

$$3 \mid y_{00} + 2y_{01} + 2y_{10} + y_{12} + y_{21} + 2y_{22}, \tag{5.10}$$

$$m \mid x_{22} + y_{10} + y_{12} - z. (5.11)$$

The above restrictions are enough to show that certain entries in \mathcal{D}_{3m} are not in any transversal.

Lemma 5.7. No transversal of \mathcal{D}_{3m} includes an entry in row (1,0) and column (0,d), where d is arbitrary.

Proof. We are required to show $y_{10} = 0$, so assume for the sake of contradiction that $y_{10} = 1$. By (5.1) it follows that $y_{12} = 0$.

First suppose that z = 1.

From (5.11), $x_{22} = 0$. To satisfy (5.10) and (5.1) the only possibilities are

- (i) $y_{01} = y_{22} = 1$, $y_{00} = y_{21} = 0$,
- (ii) $y_{00} = 1$, $y_{01} = y_{21} = y_{22} = 0$,
- (iii) $y_{21} = 1$, $y_{00} = y_{01} = y_{22} = 0$.

In all three cases x_{00} may be calculated from (5.9) but its value violates (5.2).

We conclude that z = 0, and hence $x_{22} = m - 1$ from (5.11). Again to satisfy (5.10) and (5.1) we must have (i), (ii) or (iii). However, (i) and (5.9) imply $x_{00} = m - 2$, which together with (5.7) gives the contradiction $x_{12} + x_{21} < 0$.

Similarly, (ii) and (5.9) imply $x_{00} = m - 2$, which with (5.5) leads to $x_{20} = 0$. However, this is impossible since having $x_{20} = y_{21} = y_{22} = 0$ prevents the transversal from including any entry in row (2,0).

Finally, (iii) and (5.9) imply $x_{00} = m - 1$ which breaches (5.2). We have exhausted all possibilities, and are forced to conclude that $y_{10} = 0$ as required.

Lemma 5.8. No transversal of \mathcal{D}_{3m} includes an entry in row (0, b) and column (0, d), where $b \in \{2, 3, ..., m-1\}$ and d is arbitrary.

Proof. We are required to show that $x_{00} - z = 0$ and may assume, given Lemma 5.7, that $y_{10} = 0$. From (5.9) we have

$$0 = 3(x_{00} - z) - 3(x_{22} + y_{12} - z) + y_{00} + 2y_{01} + y_{12} - 2y_{21} - y_{22}.$$
 (5.12)

As m > 1, we see from (5.11) that the only possible values for $x_{22} + y_{12} - z$ are 0 and m. In the latter case, (5.12) and (5.2) yield the immediate contradiction

$$0 \le 3(m-2) - 3m + y_{00} + 2y_{01} + y_{12} \le -2.$$

That leaves the case when $x_{22} + y_{12} - z = 0$. Here, (5.12) and (5.1) yield

$$0 \geqslant 3(x_{00} - z) - y_{21} - (y_{21} + y_{22}) \geqslant 3(x_{00} - z) - 2.$$

Given that $x_{00} - z$ is a non-negative integer, it must be zero and we are done. \Box

Lemmas 5.7 and 5.8 combine to prove Theorem 5.6. Note that in both lemmas the symbols occurring in the transversal-free entries have the form (0, f) for arbitrary f. Thus the transversal-free entries do form a latin subrectangle as claimed.

Computational results summarised in Section 5.4 show that there is no latin square of order 9 whose set of transversal-free entries contains a subrectangle of larger dimensions than 2×3 , although three species of order 9 have more than 6 transversal-free entries. We confirmed that Lemmas 5.7 and 5.8 identify the only transversal-free entries in \mathcal{D}_9 . Also by computer, $\lambda(\mathcal{D}_9) = 5$ which is slightly less than the value implied by the lemmas.

Certainly, \mathcal{D}_{3m} can never provide a counterexample to Conjecture 1.1. For example a transversal of \mathcal{D}_{3m} is given by

$$\operatorname{col}((a,b)) = \begin{cases} \{(a,0)\} & \text{if } b = 0, \\ \{(a-1,b)\} & \text{otherwise.} \end{cases}$$

5.3 Transversals in \mathcal{U}_n and \mathcal{B}_n

In this section we prove Theorem 5.2. We use the latin squares \mathcal{U}_n and \mathcal{B}_n defined in Chapter 2.

We remind the reader that \mathcal{B}_n is defined for n=2h, where $h \geq 5$ and h is odd.

Lemma 5.9. The latin square \mathcal{B}_n has a transversal. Every transversal in \mathcal{B}_n coincides on a single element e, where e = (4, 4, 4) if n = 10 and otherwise e = (1, n - 3, 1).

Proof. Assume that $T \subset \mathcal{B}_n$ is a transversal. First we prove the second statement of the lemma. Lemma 2.1 requires that $\sum_{e \in T} \Delta(e) \equiv h \mod n$, an odd value. The possible choices for odd Δ values are displayed (2.22). Respecting that T has at most one element in each row and each column, the odd sums obtained by using odd Δ values alone are ± 1 , ± 3 . Hence a sum of $\pm (h-1)$ or $\pm (h-3)$ must be met by elements of Δ_a where a is even. These sets Δ_a are listed in (2.23).

Considering the rows, T has at most one element in each of the sets $(\Delta_{-4} \cap r_4) \cup \Delta_{-3}$, $(\Delta_4 \cap r_0) \cup \Delta_1$, $\Delta_{h-7} \cap r_6$ and $\Delta_{h+7} \cap r_{h-1}$. It follows that, if n = 10, T contains the element $(4, 4, 4) \in \Delta_{-4}$ and an element of Δ_{-1} . If n > 10, then T must contain the element $(1, n - 3, 1) \in \Delta_3$, some element from $\Delta_4 \cap r_0$ and, for n > 14, we also need an element of $\Delta_{h-7} \cap r_6$.

Next we show that a transversal T exists in \mathcal{B}_n .

If n = 10, then T is given by

$$\{(0,7,7), (1,9,9), (2,0,2), (3,2,5), (4,4,4), (5,3,8), (6,5,1), (7,6,3), (8,8,6), (9,1,0)\}.$$

If n = 14, then T is given by;

$$\{(0,0,4), (1,11,1), (2,12,0), (3,13,2), (4,2,6), (5,3,8), (6,5,11), (7,6,13), (8,4,12), (9,1,10), (10,7,3), (11,8,5), (12,9,7), (13,10,9)\}.$$

For n > 14, we specify T in two cases.

Case 1: $n \equiv 2 \mod 8$

$$\operatorname{col}(x) = \begin{cases} \{0\} & \text{if } x = 0, \\ \{n-3\} & \text{if } x = 1, \\ \{h+3\} & \text{if } x = 6, \\ \{h-1\} & \text{if } x = h-3 \text{ and } n \equiv 10 \text{ mod } 16, \\ \{h-5\} & \text{if } x = h+1 \text{ and } n \equiv 2 \text{ mod } 16, \\ \{1\} & \text{if } x = n-1, \\ \{x+1\} & \text{if } x > 0 \text{ and } x \equiv 0 \text{ mod } 4, \\ \{x-2\} & \text{if } 1 < x < n-1 \text{ and } x \equiv 1 \text{ mod } 4, \\ \{x-1\} & \text{if } x \equiv 3 \text{ mod } 4, \\ \{x-10\} & \text{if } 14 \leqslant x < h+5 \text{ and } x \equiv 6 \text{ mod } 8, \\ \{x+2\} & \text{if } x \geqslant h+5 \text{ and } x \equiv 2 \text{ mod } 4, \\ \{x+6\} & \text{otherwise.} \end{cases}$$

Case 2: $n \equiv 6 \mod 8$

$$\operatorname{col}(x) = \begin{cases} \{h - 3\} & \text{if } x = 6, \\ \{h - 4\} & \text{if } x = h + 2, \\ \{x - 1\} & \text{if } x \in \{h, h - 1\}, \\ \{x - 2\} & \text{if } 4 \leqslant x \leqslant 8 \text{ and } x \notin \{6, h, h + 1\}, \\ \{x - 4\} & \text{if } (1 \leqslant x \leqslant 3) \text{ or } (11 \leqslant x \leqslant h - 2 \text{ and } x \in F), \\ \{x - 8\} & \text{if } x = 9 \text{ or } (12 \leqslant x \leqslant h + 1 \text{ and } x \equiv 0 \text{ mod } 4), \\ \{x\} & \text{if } x = 0 \text{ or } (10 \leqslant x \leqslant h - 5 \text{ and } x \equiv 2 \text{ mod } 4), \\ \{x - 3\} & \text{otherwise.} \end{cases}$$

We remind the reader that \mathcal{U}_n is defined for $n = 2mq \ge 6$, where $q \ge 3$ and q is odd, and $m = 2^t$ for $t \ge 1$. In other words, for all $n \ge 12$ such that $n \equiv 0 \mod 4$ and n is not a power of 2.

Lemma 5.10. The latin square \mathcal{U}_n has a transversal. Every transversal in \mathcal{U}_n contains the element (0, n-m, 0).

Proof. Assume that $T \subset \mathcal{U}_n$ is a transversal. Lemma 2.1 requires that $\sum_{e \in T} \Delta(e) = qm \equiv 0 \mod m$. Hence, T must contain an even number of elements from $\Delta_{-m/2} \cup \Delta_{m/2}$. However, these elements form an intercalate so T contains at most one, hence zero, of them. The only way to satisfy Lemma 2.1 is for T to include one element from each of Δ_m and Δ_{2m} and, if $q \neq 3$, also one element from $\Delta_{(q-3)m}$. Thus, T includes two elements of $\Delta_m \cup \Delta_{2m}$ and one of those must be from outside r_{n-m} . Hence, the element (0, n-m, 0) is required in T.

We now identify a transversal $T \subset \mathcal{U}_n$.

Case 1: $n \equiv 4 \mod 8$

We have n = 2mq with m = 2. Then T is given by

$$\operatorname{col}(x) = \begin{cases} \{n-m\} & \text{if } x = 0, \\ \{mq-2m\} & \text{if } x = m \text{ and } q > 3, \\ \{m\} & \text{if } x = n-m, \\ \{0\} & \text{if } x = n-2m, \\ \{x+2m+1\} & \text{if } x \in E \text{ and } mq-2m \leqslant x < n-2m, \\ \{x-2m+1\} & \text{if } x \in F \text{ and } x > mq, \\ \{x\} & \text{otherwise.} \end{cases}$$

Case 2: $n \equiv 0 \mod 8$

We have n = 2mq with $m \ge 4$. Then T is given by

$$\operatorname{col}(x) = \begin{cases} \{n-m\} & \text{if } x = 0, \\ \{mq-2m\} & \text{if } x = m \text{ and } q > 3, \\ \{m\} & \text{if } x = n-m, \\ \{mq+1\} & \text{if } x = mq-2m, \\ \{x+3\} & \text{if } x \in E \text{ and } mq \leqslant x < n-m, \\ \{x-1\} & \text{if } x \in F \text{ and } mq < x < n-m, \\ \{x+1\} & \text{if } n-m < x < n, \\ \{x\} & \text{otherwise.} \end{cases}$$

Together, Lemmas 5.9 and 5.10 cover all cases to prove Theorem 5.2.

The proof of Lemma 5.10 depends on the existence of a proper odd divisor q. It is possible to modify the definition for \mathcal{U}_n to permit q=1 but in that case each of the 3 elements of Δ_m alone satisfy Lemma 2.1, so the most we can say is that $\lambda(\mathcal{U}_n) \leq 3$. For the case that n is a power of 2, a promising stronger restriction occurs in the family \mathcal{A}_n .

The family A_n is defined for n = 16d, where $d \ge 1$.

Lemma 5.11. A transversal in A_{16d} intersects the elements (10d, 0, 4d) and (4d, 13d, 14d).

Proof. Assume that T is a transversal in \mathcal{A}_n . Lemma 2.1 requires that $\sum_{e \in T} \Delta(e) \equiv 8d \mod n$. The possible choices are shown in (2.20). Ensuring at most one element from each row and column means that T contains elements $(10d, 0, 4d) \in \Delta_{-6d}$, $(4d, 13d, 14d) \in \Delta_{-3d}$ and one element of Δ_d .

By computer we found that for $n \leq 64$, A_n does possess transversals. A general pattern for finding one is not known.

Conjecture 5.12. The latin square A_n has a transversal. Hence, Theorem 5.2 holds for all even orders $n \ge 10$.

We conclude this section by noting that, by the next statement, our latin squares of even order n with $\lambda = 1$ yield a confirmed bachelor latin square of order (n + 1).

Theorem 5.13. Suppose that L is a latin square of order n and that T is a transversal in L such that $L \setminus T$ contains no transversal. Then there exists a latin square L' of order (n+1) such that L' has at least one transversal-free entry.

Proof. Let $T \subset L$ be a transversal and let $\mathcal{I}(L) = \mathbb{Z}_n$. We construct L' by prolongation of the transversal T. That is, the latin square L' of order (n+1) is given by

$$L'_{n+1}[x,y] = \begin{cases} n & \text{if } x = y = n \text{ or } (x,y,z) \in T, \\ z & \text{if } x = n \text{ and } (x',y,z) \in T, \\ z & \text{if } y = n \text{ and } (x,y',z) \in T, \\ L[x,y] & \text{if otherwise } (x,y,z) \in L. \end{cases}$$
(5.13)

Consider the element $(n, n, n) \in L'$. This element must be transversal-free as otherwise there is a transversal contained in $L \setminus T$.

Applying Theorem 5.13 with $L = \mathcal{B}_n$ justifies our claim in Section 5.1 of a new construction for a confirmed bachelor latin square for all orders $n \equiv 3 \mod 4$, when $n \geqslant 11$.

5.4 Latin squares of small order

The computational results in this section have been independently confirmed by Ian Wanless.

For order 9, we used a program to generate isotopy class representatives. The program, which we also used in Section 4.5.3, is described in [119]. We report on two projects involving restricted transversals.

Search 1: for transversal-free entries. This search of latin squares of order 9 produced a list of isotopy classes which possess transversal-free entries. The list was then reduced to species representatives. We recorded that 20011 species of order 9 possess at least one transversal-free entry. For orders $n \leq 8$ we ran our programs directly on a list of species representatives. In Table 5.1 we summarise the latin squares of order $4 \leq n \leq 9$ according to the number of transversal-free entries they contain. The single species for each order 1, 2 and 3 have, respectively, 0,4 and 0 transversal-free entries.

Search 2: for partitions with transversals. In this search of the latin squares of order 9 we kept a list of each isotopy class representative of order 9 which failed to possess a $(1^5, 2^2)$ -partition. Reducing this list to species representatives we obtained a list Γ of 182 species representatives. A further search of the species in Γ showed that each of them has a $(1^3, 2^3)$ -partition. Thus, all species of order 9 possess a $(1, 2^4)$ -partition. Combined with known results for $n \leq 8$, first shown by Wanless [150], and confirmed by our data in Section 4.5.2, we proved the following.

Theorem 5.14. Conjecture 1.3 is true for the latin squares of order $n \leq 9$.

By further tests on species in Γ we found that the species of order 9 satisfy the following.

- 1. Exactly 156 species fail to possess a $(1^5, 2^2)$ -partition or a $(1^6, 3)$ -partition, but do have a $(1^5, 4)$ -partition.
- 2. Hence, in total 19270853515 species have $\lambda \geq 5$.
- 3. Exactly 26 species fail to possess a $(1^5, 4)$ -partition. Of these, 23 species have a $(1^4, 5)$ -partition, hence have $\lambda = 4$. The remaining 3 species each have $\lambda = 3$.
- 4. Exactly 7 species in Γ have an order 4 subsquare, and that subsquare is paratopic to the Cayley table of $\mathbb{Z}_2 \oplus \mathbb{Z}_2$. Hence, Lemma 4.32 applies to these 7 species. All 7 of these species have $\lambda = 4$.
- 5. It follows, from 4. and Theorem 5.3, that the other 36000 species in Ω (the species of Lemma 4.32) must have $\lambda = 5$. Similarly, any latin square of order 9 containing a subsquare isotopic to the Cayley table of \mathbb{Z}_4 must have $\lambda = 5$, however we did not calculate the number of species that this involves.

To the extent that we achieved for n=9, we summarise in Table 5.2 the species of order $n \leq 9$ according to λ . Data for $n \leq 8$ in Table 5.2 was first summarised in [150] building upon the studies for $n \leq 7$ mentioned in Section 1.14.1. The mark κ indicates that zero species possess the λ value, and \checkmark means that at least one species has the stated λ value.

We next illustrate a few examples of interest from the two projects.

The unique species of order 9 achieving the maximum number of transversal-free entries is one well known to us. The latin square L_1 , shown below in semi-symmetric form, is isotopic to W_9 . It has 415 transversals. Shading shows its 10 transversal-free entries which correspond to all of the elements $\Delta_* \subset W_9$. Other data on the indivisible plexes and partitions of W_9 is recorded in Table 4.13 (Species D). Also, by computation, we note that $\lambda(W_9) = 6$.

The next two latin squares, L_2 and L_3 shown below, represent the only other species of order 9 with more than 6 transversal-free entries. All three squares L_1 , L_2 and L_3 have at least one subsquare of order 3.

 L_2 has 287 transversals, 7 transversal-free entries (as shaded) and $\lambda(L_2) = 6$.

$$L_2 = \begin{pmatrix} 0 & 1 & 2 & 3 & 4 & 5 & 6 & 7 & 8 \\ 1 & 2 & 0 & 4 & 5 & 3 & 7 & 8 & 6 \\ 2 & 0 & 1 & 5 & 3 & 4 & 8 & 6 & 7 \\ \hline 3 & 4 & 5 & 6 & 7 & 8 & 1 & 2 & 0 \\ 4 & 7 & 3 & 2 & 8 & 6 & 0 & 5 & 1 \\ \hline 7 & 3 & 4 & 8 & 6 & 0 & 5 & 1 & 2 \\ \hline 6 & 5 & 8 & 1 & 2 & 7 & 4 & 0 & 3 \\ 5 & 8 & 6 & 7 & 0 & 1 & 2 & 3 & 4 & 5 \end{pmatrix} \quad L_3 = \begin{pmatrix} 0 & 1 & 2 & 4 & 5 & 3 & 6 & 7 & 8 \\ 1 & 2 & 0 & 5 & 3 & 4 & 7 & 8 & 6 \\ 4 & 0 & 1 & 3 & 2 & 5 & 8 & 6 & 7 \\ \hline 2 & 8 & 6 & 0 & 1 & 7 & 4 & 5 & 3 \\ 8 & 6 & 4 & 1 & 7 & 0 & 5 & 3 & 2 \\ 6 & 4 & 8 & 7 & 0 & 1 & 3 & 2 & 5 \\ \hline 3 & 7 & 5 & 2 & 8 & 6 & 0 & 1 & 4 \\ 7 & 5 & 3 & 8 & 6 & 2 & 1 & 4 & 0 \\ 5 & 3 & 7 & 6 & 4 & 8 & 2 & 0 & 1 \end{pmatrix}$$

 L_3 is isotopic to its transpose by the symbol permutation (38)(57). It has 8 transversal-free entries (shaded) and 92 transversals in total. This latin square exhibits both types of restrictions on its transversals. It is also one of three species recording the least λ value for order 9, that is $\lambda(L_3) = 3$.

The other two species which achieve $\mu(9) = 3$ are illustrated below by L_4 and L_5 . Note that L_3 differs from L_4 in only 6 entries, a 3-cycle between symbols 2 and 4.

 L_4 has 84 transversals and zero transversal-free entries. The square is nearly symmetric, just transpose, and then apply the symbol permutation (36)(58). A case analysis using Lemma 2.1 can be used to show that a transversal in L_4 must include 2 entries from the centre subsquare (with symbols $\{0, 1, 7\}$) and it follows from this that it has at most 3 disjoint transversals.

$$L_{4} = \begin{pmatrix} 0 & 1 & 2 & 4 & 5 & 3 & 6 & 7 & 8 \\ 1 & 2 & 0 & 5 & 3 & 4 & 7 & 8 & 6 \\ 2 & 0 & 1 & 3 & 4 & 5 & 8 & 6 & 7 \\ \hline 4 & 8 & 6 & 0 & 1 & 7 & 2 & 5 & 3 \\ 8 & 6 & 4 & 1 & 7 & 0 & 5 & 3 & 2 \\ 6 & 4 & 8 & 7 & 0 & 1 & 3 & 2 & 5 \\ \hline 3 & 7 & 5 & 2 & 8 & 6 & 0 & 1 & 4 \\ 7 & 5 & 3 & 8 & 6 & 2 & 1 & 4 & 0 \\ 5 & 3 & 7 & 6 & 2 & 8 & 4 & 0 & 1 \end{pmatrix} \qquad L_{5} = \begin{pmatrix} 2 & 1 & 0 & 4 & 5 & 3 & 7 & 8 & 6 \\ 1 & 0 & 2 & 7 & 8 & 6 & 4 & 5 & 3 \\ 0 & 2 & 1 & 8 & 6 & 7 & 5 & 3 & 4 \\ \hline 5 & 8 & 7 & 3 & 0 & 4 & 6 & 1 & 2 \\ 3 & 6 & 8 & 5 & 4 & 0 & 2 & 7 & 1 \\ 4 & 7 & 6 & 0 & 3 & 5 & 1 & 2 & 8 \\ 8 & 5 & 4 & 6 & 1 & 2 & 3 & 0 & 7 \\ 6 & 3 & 5 & 2 & 7 & 1 & 8 & 4 & 0 \\ 7 & 4 & 3 & 1 & 2 & 8 & 0 & 6 & 5 \end{pmatrix}$$

The latin square L_5 is isotopic to its transpose. It has 168 transversals, including at least one through every entry, one subsquare of order 3 and 48 intercalates. A parity argument shows that any odd plex must intersect the order 3 subsquare. In fact, every one of the transversals in L_3 contains exactly one of the three elements marked by the bold entries of the subsquare but an explanation of this remains open.

For n=9, members of our other latin families of odd order show no special significance in terms of the types of restrictions of transversals we have described. The family \mathcal{H}_n yielded large indivisible plexes in Chapter 4. By computation, $\lambda(\mathcal{H}_9)=7$ and its only transversal-free entries are the two specified by Theorem 4.23. Also $\mathcal{O}_{3,3}$, from another family featured in Chapter 4, has zero transversal-free entries and $\lambda(\mathcal{O}_{3,3})=9$. Species A of Table 4.13, the latin square that we identified with an indivisible 6-plex (and a member of Ω) has 176 transversals, $\lambda=5$ and zero transversal-free entries.

The number of transversals in latin squares is studied in [118]. The authors found there that the species of order n exhibiting the least number of transversals for order $n \in \{5,7,9\}$ were all examples of semi-symmetric latin squares. For n=9, the minimum number of transversals is 68 (Species B of Table 4.13), and the average number is 214 [118]. Species B has $\lambda=5$ and zero transversal-free entries. Species C of Table 4.13 has, for order 9, the maximum number of intercalates [118]. It has 801 transversals, $\lambda=9$ and zero transversal-free entries.

Table 5.1: Species of order $4\leqslant n\leqslant 9$ according to number of transversal-free entries.

| Number of | Order n | | | | | | | |
|--------------|-----------|---|---|----|--------|-------------|--|--|
| transversal | | | | | | | | |
| free entries | 4 | 5 | 6 | 7 | 8 | 9 | | |
| 0 | 1 | 1 | 2 | 54 | 267932 | 19270833530 | | |
| 1 | | | | 11 | 13165 | 18066 | | |
| 2 | | | | 26 | 1427 | 1853 | | |
| 3 | | | | 12 | 253 | 54 | | |
| 4 | | | | 12 | 508 | 21 | | |
| 5 | | | | 6 | 89 | 7 | | |
| 6 | | | 1 | 8 | 65 | 7 | | |
| 7 | | | | 3 | 33 | 1 | | |
| 8 | | | | 4 | 48 | 1 | | |
| 9 | | | | | 25 | | | |
| 10 | | | | 1 | 27 | 1 | | |
| 11 | | | | 1 | 9 | | | |
| 12 | | 1 | 2 | 6 | 9 | | | |
| 13 | | | | 1 | 2 | | | |
| 14 | | | | | 2 | | | |
| 16 | 1 | | 1 | 1 | 27 | | | |
| 18 | | | | | 1 | | | |
| 20 | | | | | 1 | | | |
| 28 | | | | | 1 | | | |
| 36 | | | 6 | 1 | | | | |
| 64 | | | | | 33 | | | |

Table 5.2: Species of order $n \leqslant 9$ according to $\lambda.$

| | Order n | | | | | | | | |
|-----------|---------|---|---|----|--------|--------------|--|--|--|
| λ | 4 | 5 | 6 | 7 | 8 | 9 | | | |
| 0 | 1 | × | 6 | х | 33 | × | | | |
| 1 | × | 1 | × | 1 | × | × | | | |
| 2 | × | × | 2 | 5 | 7 | × | | | |
| 3 | × | × | × | 24 | 46 | 3 | | | |
| 4 | 1 | × | 4 | 68 | 712 | 23 | | | |
| 5 | | 1 | × | 43 | 71330 | \checkmark | | | |
| 6 | | | × | × | 209505 | \checkmark | | | |
| 7 | | | | 6 | х | \checkmark | | | |
| 8 | | | | | 2024 | × | | | |
| 9 | | | | | | \checkmark | | | |

Another type of restriction on the transversals of a latin square is the following, weaker, restriction within a partition. For a given latin square L, we define:

```
\alpha(L) = \min \{ a : L \text{ has a } (1^a, n-a) \text{-partition with no 1-plex in the } (n-a) \text{-plex} \}.
```

By definition, for a given L, $\alpha(L) \leq \lambda(L)$. Our study of $\alpha(L)$ is quite limited as our computational projects targeted different questions. However for order 9, we found all of the species with $\alpha = 1$ and $\alpha = 2$. These are, respectively, the 36007 species of Lemma 4.32, and 8177 of the species in Ψ that we identified in Section 4.5.3 during the process of ruling out the existence of an indivisible $(1^2, 7)$ -partition.

For all positive integers n we define:

$$\beta(n) = \min \{ \alpha(L) : L \text{ is a latin square of order } n \}.$$

Conjecture 1.1 implies $\beta(n) \ge 1$ for all odd n, and we know by \mathbb{Z}_n that $\beta(n) = 0$ for all even n. In contrast to what we know of $\mu(n)$, Corollary 4.30 showed that for $n \ge 5$, $\beta(n) \le 1$ if $n \equiv 1 \mod 4$, and $\beta(n) \le 3$ if $n \equiv 3 \mod 4$. Some improvement, if indeed possible, of the upper bound of $\beta(n)$ remains open for $n \equiv 3 \mod 4$, where $n \ge 11$. The values, $\beta(5) = \beta(7) = 1$ and $\beta(3) = 3$, are not specifically reported in our data tables.

Unlike the restriction $\lambda(L) < n$, a small value $\alpha(L)$ does not in general immediately imply that the given latin square L has no 1-partition. As we have reported, if L is of order 9 then $\alpha(L) = 1$ implies that $\lambda(L) \leq 5$. We record as follows, $\alpha(L)$ for the examples illustrated in this section: $\alpha(W_9) = \alpha(L_2) = 3$, $\alpha(L_3) = \alpha(L_4) = 2$, $\alpha(L_5) = \lambda(L_5) = 3$. For our family members of order 9 we found that: $\alpha(\mathcal{D}_9) = 2$, $\alpha(\mathcal{H}_9) = 4$ and $\alpha(\mathcal{O}_{3,3}) = 3$.

5.5 Concluding remarks

The computational results of this chapter suggest that almost all large latin squares have no transversal-free entries. The various types of restrictions on the transversals that we have considered show that the existence of one type of restriction does not, in general, imply another. For example, L_4 has a small value of λ but no transversal-free entries.

Our data suggests that Theorems 5.2 and 5.6 identify what may be considered exotic behaviour for larger n. Indeed, all of the bachelor families mentioned here and in other chapters may be considered rare for larger n if the more recent views regarding Conjecture 1.24 hold true. Those more recent views [118, 155] suggested that, asymptotically, the proportion of bachelor latin squares is equal to zero.

Theorem 5.6 shows that the family \mathcal{D}_{3m} has quadratically many transversal-free en-

tries and is thus much stronger than Theorem 5.5. It would be interesting to know if a result similar to Theorem 5.6 applies to orders that are not divisible by 3.

An extension of Theorem 5.2 to all even $n \ge 10$ remains open. This would be resolved by a proof of Conjecture 5.12 that \mathcal{A}_n has a transversal. Data in Table 5.2 shows that if L is of even order $n \le 8$ then $\lambda(L) \ne 1$. Thus, Theorem 5.2 cannot be extended to smaller even n.

Collectively, Theorem 5.2, Lemma 5.1, the data in Table 5.2 and that on indivisible partitions in Section 4.5.2 support the following.

Conjecture 5.15. For all even $n \ge 10$ and each $m \in \{0, 1, 2, ..., n-2\} \cup \{n\}$, there exists a latin square of order n such that $\lambda(L) = m$.

Far less is known about λ for the latin squares of odd order. Our finding that $\mu(9)=3$ offers more support for Conjecture 1.1. There appears room to improve on Theorem 5.4 which was drawn from work in [73, 117]. In fact, for odd orders n and $\lambda < n$, it remains to be shown in general that any particular value of λ is actually achieved. We note that for order n=11 the construction by Evans [73] has an indivisible (1⁶, 5)-partition. Indivisibility of the 5-plex follows from the same argument stated in [73] and which justifies Theorem 5.4. If it is true that $\lambda = (n+1)/2$ for the construction in [73], then it would follow that there is an indivisible $(1^{(n+1)/2}, (n-1)/2)$ -partition for $n \ge 7$ when $n \equiv 3 \mod 4$.

It would be interesting to determine whether or not $\mu(n) < n/2$ for all odd n > 3. In particular:

Problem 5.16. Is $\mu(n)$ unbounded $n \to \infty$?

Afterword

Two articles based on results from this thesis are [19] and [67]. Results of Chapter 3 are published in [67]. The results in Section 4.2 are in [19]. These two articles were already mentioned.

Shortly after submission of this thesis, we received the news that a further article based on results from Chapter 4 has been accepted for publication. The article, [68], contains Theorem 4.1 and the results in Sections 4.3.1 and 4.4. Also reported in [68] are the computations and data on indivisible partitions in Section 4.5, and the computational proof of Theorem 5.14.

Also, just after submission of the thesis, the author proved two of the stated conjectures.

Conjecture 4.35 was proved by showing an indivisible $(\lfloor \frac{n}{2} \rfloor, \lceil \frac{n}{2} \rceil)$ -partition in \mathcal{H}_n . This gives a slightly improved lower bound for $\kappa(n)$. The new result, along with Theorems 4.2 and 4.23, and related work in Section 4.3.2, is submitted for publication [66].

Conjecture 5.12 was proved by showing a transversal in \mathcal{A}_n . This gives the anticipated extension of Theorem 5.2 to all even orders $n \geq 10$. The extended result, along with Theorem 5.6, Table 5.1 and related work in Chapter 5, is submitted for publication [69].

References

- [1] R. J. R. Abel, D. Combe, G. Price, and W. D. Palmer. Existence of generalized Bhaskar Rao designs with block size 3. *Discrete Math.*, **309**:4069–4078, 2009.
- [2] R. Aharoni and E. Berger. Rainbow matchings in r-partite r-graphs. *Electron. J. Combin.*, **16**:R119, 2009.
- [3] S. Akbari and A. Alipour. Transversals and multicoloured matchings. *J. Combin. Des.*, **12**:325–332, 2004.
- [4] B. Alspach and K. Heinrich. Matching designs. Australas. J. Combin., 2:39–55, 1990.
- [5] A. Altshuler. Construction and enumeration of regular maps on the torus. Discrete Math., 4:201–217, 1973.
- [6] L. D. Andersen and A. J. W. Hilton. Thank Evans! Proc. London Math Soc., 47:507–522, 1983.
- [7] R. A. Bailey. Orthogonal partitions in designed experiments. *Des. Codes Cryptogr.*, 8:45–77, 1996.
- [8] K. Balasubramanian. On transversals in latin squares. *Linear Algebra Appl.*, **131**:125–129, 1990.
- [9] J. A. Bate and G. H. J. van Rees. Minimal and near-minimal critical sets in back-circulant latin squares. *Australas. J. Combin.*, **27**:47–61, 2003.
- [10] R. Bean, H. Bidkhori, M. Khosravi, and E. S. Mahmoodian. k-homogeneous latin trades. Bayreuth. Math. Schr., 74:7–18, 2005.
- [11] D. Bedford and R. M. Whitaker. Enumeration of transversals in the Cayley tables of the non-cyclic groups of order 8. *Discrete Math.*, **197/198**:77–81, 1999.
- [12] Bagheri Gh. Behrooz and E.S. Mahmoodian. On the existence of k-homogeneous latin bitrades. *Util. Math*, to appear.
- [13] R. C. Bose. On the application of the properties of Galois fields to the problem of construction of hyper-Graeco-Latin squares. Sankhyā, 3:323–338, 1938.
- [14] R. C. Bose and S. S. Shrikhande. On the falsity of Euler's conjecture about the non-existence of two orthogonal latin squares of order 4t + 2. *Proc. Nat. Acad. Sci. U.S.A.*, **45**:734–737, 1959.
- [15] R. C. Bose and S. S. Shrikhande. On the construction of sets of mutually orthogonal latin squares and the falsity of a conjecture of Euler. *Trans. Amer. Math. Soc.*, **95**:191–209, 1960.
- [16] R. C. Bose, S. S. Shrikhande, and E. T. Parker. Further results on the construction of mutually orthogonal latin squares and the falsity of Euler's conjecture. *Canad. J. Math.*, **12**:189–203, 1960.

[17] A. E. Brouwer, A. J. de Vries, and R. M. A. Wieringa. A lower bound for the length of partial transversals in a latin square. *Nieuw Arch. Wisk.*, **26**:330–332, 1978.

- [18] R. H. Bruck and H.J. Ryser. The non-existence of certain finite projective planes. *Canad. J. Math.*, **1**:88–93, 1949.
- [19] D. Bryant, J. Egan, B. Maenhaut, and I.M. Wanless. Indivisible plexes in latin squares. *Des. Codes Cryptogr.*, **52**:93–105, 2009.
- [20] B. Burton. Completion of partial latin squares. Honours thesis, Department of Mathematics, The University of Queensland, Brisbane, Australia, 1996.
- [21] P. J. Cameron. Permutation codes. European. J. Combin., 31:482–490, 2010.
- [22] P. J. Cameron and C. Y. Ku. Intersecting families of permutations. *European J. Combin.*, **24**:881–890, 2003.
- [23] P. J. Cameron and I. M. Wanless. Covering radius for sets of permutations. *Discrete Math.*, **293**:91–109, 2005.
- [24] N. Cavenagh, D. Donovan, and A. Drápal. 3-homogeneous latin trades. *Discrete Math.*, **300**:57–70, 2005.
- [25] N. Cavenagh, D. Donovan, and A. Drápal. 4-homogeneous latin trades. Australas. J. Combin., 32:285–303, 2005.
- [26] N. Cavenagh and P. Lisonek. Planar Eulerian triangulations are equivalent to spherical latin bitrades. J. Combin. Theory Ser. A, 115:193–197, 2008.
- [27] N. J. Cavenagh. Latin trade algorithms and the smallest critical set in a latin square. J. Autom. Lang. Comb., 8:567–578, 2003.
- [28] N. J. Cavenagh. Embedding 3-homogeneous latin trades into abelian 2-groups. Comment. Math. Univ. Carolin., 45:191–212, 2004.
- [29] N. J. Cavenagh. A uniqueness result for 3-homogeneous latin trades. *Comment. Math. Univ. Carolin.*, **47**:337–358, 2006.
- [30] N. J. Cavenagh. The theory and application of latin bitrades: a survey. *Math. Slovaca*, **58**:691–718, 2008.
- [31] N. J. Cavenagh, D. M. Donovan, and E. S. Yazici. Minimal homogeneous latin trades. *Discrete Math.*, **306**:2047–2055, 2006.
- [32] N. J. Cavenagh, A. Drápal, and C. Hämäläinen. Latin bitrades derived from groups. *Discrete Math*, **308**:6189–6202, 2008.
- [33] N. J. Cavenagh, C. Greenhill, and I. M. Wanless. The cycle structure of two rows in a random latin square. *Random Structures Algorithms*, **33**:286–309, 2008.
- [34] N. J. Cavenagh, C. Hämäläinen, and A. M. Nelson. On completing three cyclically generated transversals to a latin square. *Finite Fields Appl.*, **15**:294–303, 2009.
- [35] N. J. Cavenagh and I. M. Wanless. Latin trades in groups defined on planar triangulations. *J. Algebraic Combin.*, **30**:323–347, 2009.
- [36] N. J. Cavenagh and I. M. Wanless. On the number of transversals in Cayley tables of cyclic groups. *Discrete Appl. Math.*, **158**:136–146, 2010.
- [37] D. Clark and J. T. Lewis. Transversals of cyclic latin squares. *Congr. Numer.*, 128:113–120, 1997.
- [38] C. J. Colbourn. The complexity of completing partial latin squares. *Discrete Appl. Math.*, 8:25–30, 1984.

[39] C. J. Colbourn and J. H. Dinitz (eds). *Handbook of Combinatorial Designs*, Second Edition. Chapman & Hall/CRC, Boca Raton, 2007.

- [40] C. J. Colbourn and A. Rosa. Triple Systems. Clarendon Press, Oxford, 1999.
- [41] C. Cooper. A lower bound for the number of good permutations. *Data Recording, Storage and Processing (Nat. Acad. Sci. Ukraine)*, **2.3**:15–25, 2000.
- [42] C. Cooper, R. Gilchrist, I. N. Kovalenko, and D. Novakovic. Estimation of the number of "good" permutations with applications to cryptography. *Cybernet. Systems Anal.*, **35**:688–693, 1999.
- [43] C. Cooper and I. N. Kovalenko. The upper bound for the number of complete mappings. *Theory Probab. Math. Statist.*, **53**:77–83, 1996.
- [44] P. Danziger, I.M. Wanless, and B.S. Webb. Monogamous latin squares. preprint.
- [45] D. E. Daykin and R. Häggkvist. Completion of sparse partial latin squares. Graph theory and combinatorics (Cambridge, 1983), pages 127–132, Academic Press, London, 1984.
- [46] J. Dénes and A. D. Keedwell. *Latin squares and their applications*. Akadémiai Kiadó, Budapest, 1974.
- [47] J. Dénes and A.D. Keedwell. A new conjecture concerning admissibility of groups. *European J. Combin.*, **10**:171–174, 1989.
- [48] J. Dénes and A. D. Keedwell. *Latin Squares: New Developments in the Theory and Applications*. Annals Discrete Mathematics **46**, North-Holland, Amsterdam, 1991.
- [49] T. Denley. On a conjecture of Häggkvist on partial latin squares. Congr. Numer., 150:73–78, 2001.
- [50] T. Denley and R. Häggkvist. Completing some partial latin squares. *Europ. J. Combin.*, **21**:877–880, 2000.
- [51] I. I. Derienko. On the Brualdi hypothesis (Russian). Mat. Issled., 102:53–65, 1988.
- [52] R. Diestel. Graph theory. Third edition. Springer-Verlag, Berlin, 2005.
- [53] D. Donovan. The completion of partial latin squares. Australas. J. Combin., 22:247–264, 2000.
- [54] S. T. Dougherty. Latin squares. http://academic.uofs.edu/faculty/Doughertys1/latin.htm.
- [55] S. T. Dougherty. Planes, nets and codes. Math. J. Okayama Univ., 38:123–143, 1996.
- [56] D. A. Drake. Maximal sets of latin squares and partial transversals. *J. Statist. Plann. Inference*, 1:143–149, 1977.
- [57] D. A. Drake, G. H. J. van Rees, and W. D. Wallis. Maximal sets of mutually orthogonal latin squares. *Discrete Math.*, **194**:87–94, 1999.
- [58] A. Drápal. On a planar construction of quasigroups. Czechoslovak Math. J., 41:538–548, 1991.
- [59] A. Drápal. Hamming distances of groups and quasi-groups. *Discrete Math.*, **235**:189–197, 2001.
- [60] A. Drápal. Geometrical structure and construction of latin trades. Adv. Geom., 9:311–348, 2009.
- [61] A. Drápal, C. Hämäläinen, and V. Kala. Latin bitrades, dissections of equilateral triangles and abelian groups. *J. Combin. Design*, **18**:1–24, 2010.

[62] A. Drápal, C. Hämäläinen, and D. Rosendorf. An enumeration of spherical latin bitrades. *Australas. J. Combin.*, to appear.

- [63] A. Drápal and T. Kepka. Exchangeable partial groupoids I. Acta Univ. Carolin. Math. Phys., 24:57–72, 1983.
- [64] A. Drápal and T. Kepka. Exchangeable partial groupoids. II. *Acta Univ. Carolin. Math. Phys.*, **26**:3–9, 1985.
- [65] J. Egan. Generalised transversals of latin squares. Honours thesis, Faculty of Technology, Charles Darwin University, Darwin, Australia, 2005.
- [66] J. Egan. Bachelor latin squares with large indivisible plexes. preprint.
- [67] J. Egan and I. M. Wanless. Latin squares with no small odd plexes. J. Combin. Des., 16:477–492, 2008.
- [68] J. Egan and I. M. Wanless. Indivisible partitions of latin squares. *J. Statist. Plann. Inference*, **141**:402–417, 2011.
- [69] J. Egan and I. M. Wanless. Latin squares with restricted transversals. preprint.
- [70] P. Erdős and J. Spencer. Lopsided Lovász local lemma and latin transversals. Discrete Appl. Math., **30**:151–154, 1991.
- [71] L. Euler. Recherches sur une nouvelle espèce de quarrés magiques. Verh. v. h. Zeeuwsch Genoottsch. der Wetensch., Vlissingen, 9:85–239, 1782.
- [72] A. B. Evans. The existence of complete mappings of finite groups. *Congr. Numer.*, **90**:65–75, 1992.
- [73] A. B. Evans. Latin squares without orthogonal mates. Des. Codes Cryptogr., 40:121–130, 2006.
- [74] A. B. Evans. The admissibility of sporadic simple groups. *J. Algebra*, **321**:105–116, 2009.
- [75] T. Evans. Embedding incomplete latin squares. Amer. Math. Monthly, 67:958–961, 1960.
- [76] D. J. Finney. Some orthogonal properties of the 4×4 and 6×6 latin squares. Ann. Eugenics., 12:213–219, 1945.
- [77] D. J. Finney. Orthogonal partitions of the 5×5 latin squares. Ann. Eugenics., 13:1-3, 1946.
- [78] D. J. Finney. Orthogonal partitions of the 6×6 latin squares. Ann. Eugenics., 13:184–196, 1946.
- [79] D. J. Finney. Some enumerations for the 6×6 latin squares. *Utilitas Math.*, **21A**:137–153, 1982.
- [80] R. A. Fisher and F. Yates. The 6×6 latin squares. *Proc. Camb. Phil. Soc.*, 30:492-507, 1934.
- [81] R. A. Fisher and F. Yates. Statistical Tables for Biological, Agricultural and Medical Research, 2nd Edition. Oliver and Boyd, London, 1943.
- [82] G.H. Freeman. The addition of further treatments to latin square designs. *Biometrics*, **20**:713–729, 1964.
- [83] G. H. Freeman. Some non-orthogonal partitions of 4×4 , 5×5 and 6×6 latin squares. Ann. Math. Statist., **37**:666–681, 1966.
- [84] G. H. Freeman. Duplexes of 4×4 , 5×5 and 6×6 latin squares. *Utilitas Math.*, **27**:5–24, 1985.

[85] G. H. Freeman. Orthogonal partitions of 6×6 latin squares of set X. *Utilitas Math.*, **41**:139–149, 1992.

- [86] G. H. Freeman. Triplex properties of 6×6 latin squares. *Utilitas Math.*, **41**:11–32, 1992.
- [87] H. L. Fu, S. C. Lin, and C. M. Fu. The length of a partial transversal in a latin square. *J. Combin. Math. Combin. Comput.*, **43**:57–64, 2002.
- [88] D. C. Gilliland. A note on orthogonal partitions and some well known structures in design of experiments. *Ann. Statist.*, **5**:565–570, 1977.
- [89] M. J. Grannell, T. S. Griggs, and M. Knor. Biembeddings of symmetric configurations and 3-homogeneous latin trades. *Comment. Math. Univ. Carolin.*, 49:411–420, 2008.
- [90] M. Grüttmüller. Completing partial latin squares with two cyclically generated prescribed diagonals. J. Combin. Theory Ser A, 103:349–362, 2003.
- [91] M. Grüttmüller. Completing partial latin squares with prescribed diagonals. Discrete Appl. Math., 138:89–97, 2004.
- [92] R. Häggkvist. A solution of the Evans conjecture for latin squares of large size. Colloq. Math. Soc. János Bolyai, 18:495–514, 1978.
- [93] J. W. Hall. Designs for the addition of six further treatments to latin squares of order six. *Biometrics*, **40**:215–218, 1984.
- [94] M. Hall. An existence theorem for latin squares. Bull. Amer. Math. Soc., 51:387–388, 1945.
- [95] M. Hall and L. J. Paige. Complete mappings of finite groups. *Pacific J. Math.*, **5**:541–549, 1955.
- [96] P. Hall. On representatives of subsets. J. London Math. Soc., 10:26–30, 1935.
- [97] C. Hämäläinen. Partitioning 3-homogeneous latin bitrades. *Geom. Dedicata*, **133**:181–193, 2008.
- [98] P. Hatami and P. W. Shor. A lower bound for the length of a partial transversal in a latin square. J. Combin Theory Ser. A, 115:1103–1113, 2008.
- [99] D. M. Johnson, A. L. Dulmage, and N. S. Mendelsohn. Orthomorphisms of groups and orthogonal latin squares I. *Canad. J. Math.*, **13**:356–372, 1961.
- [100] B. Johnston and T. Fullerton. Construction of the 3^2 orthogonal partitions of 6×6 latin squares of set X. Congr. Numer., **42**:199–212, 1984.
- [101] P. Kaski and P. R. J. Östergård. Classification algorithms for codes and designs. Springer-Verlag, Berlin, 2006.
- [102] P. Katerinis. Some conditions for the existence of f-factors. J. Graph Theory, 9.513-521, 1985.
- [103] A. D. Keedwell. Critical sets in latin squares and related matters: an update. *Util. Math.*, **65**:97–131, 2004.
- [104] P. Keevash and C. Y. Ku. A random construction for permutation codes and the covering radius. *Des. Codes Cryptogr.*, **41**:79–86, 2006.
- [105] A. E. Kézdy and H. S. Snevily. Unpublished manuscript cited by [23], [130] and http://www.math.uiuc.edu/west/regs/permcov.html. 2001.
- [106] R. Killgrove, C. Roberts, R. Sternfeld, R. Tamez, R. Derby, and D. Kiel. Latin squares and other configurations. *Congr. Numer.*, **117**:161–174, 1996.

[107] K. K. Koksma. A lower bound for the order of a partial transversal in a latin square. J. Combinatorial Theory, 7:94–95, 1969.

- [108] I. N. Kovalenko. Upper bound on the number of complete maps. *Cybernet. Systems Anal.*, **32**:65–68, 1996.
- [109] J. S. Kuhl and T. Denley. On a generalization of the Evans conjecture. *Discrete Math.*, **308**:4763–4767, 2008.
- [110] N. Y. Kuznetsov. Applying fast simulation to find the number of good permutations. *Cybernet. Systems Anal.*, **43**:830–837, 2007.
- [111] N. Y. Kuznetsov. Estimating the number of good permutations by a modified fast simulation method. *Cybernet. Systems Anal.*, **44**:547–554, 2008.
- [112] C. W. H. Lam, L. Thiel, and S. Swiercz. The nonexistence of finite projective planes of order 10. *Canad. J. Math.*, **41**:1117–1123, 1989.
- [113] C. F. Laywine and G. L. Mullen. Discrete Mathematics Using Latin Squares. Wiley, New York, 1998.
- [114] J. Lefevre, D. Donovan, N. Cavenagh, and A. Drápal. Minimal and minimum size latin bitrades of each genus. *Comment. Math. Univ. Carolin.*, **48**:189–203, 2007.
- [115] J. G. Lefevre, D. Donovan, and A. Drápal. Permutation representation of 3 and 4-homogeneous latin bitrades. *Fund. Inform.*, **84**:99–110, 2008.
- [116] H. B. Mann. The construction of orthogonal latin squares. *Ann. Math. Statist.*, 13:418–423, 1942.
- [117] H. B. Mann. On orthogonal latin squares. Bull. Amer. Math. Soc., **50**:249–257, 1944
- [118] B. D. McKay, J. C. McLeod, and I. M. Wanless. The number of transversals in a latin square. *Des. Codes Cryptogr.*, **40**:269–284, 2006.
- [119] B. D. McKay, A. Meynert, and W. Myrvold. Small latin squares, quasigroups, and loops. J. Combin. Des., 15:98–119, 2007.
- [120] E.H. Moore. Tactical memoranda I-III. Amer. J. Math., 18:264–303, 1896.
- [121] S. Negami. Uniqueness and faithfulness of embedding of toroidal graphs. *Discrete Math.*, **44**:161–180, 1983.
- [122] L. J. Paige. Complete mappings of finite groups. *Pacific J. Math.*, 1:111–116, 1951.
- [123] E. T. Parker. Construction of some sets of mutually orthogonal latin squares. *Proc. Amer. Math. Soc.*, **10**:946–949, 1959.
- [124] E. T. Parker. Orthogonal latin squares. *Proc. Nat. Acad. Sci. U.S.A.*, **45**:859–862, 1959.
- [125] H. O. Pflugfelder. Quasigroups and Loops: Introduction. Heldermann Verlag, Berlin, 1990.
- [126] M. D. Plummer. Graph factors and factorization: 1985-2003: A survey. *Discrete Math.*, **307**:791–821, 2007.
- [127] D. A. Preece. Orthogonality and designs: a terminological muddle. *Utilitas Math.*, **12**:201–223, 1977.
- [128] K. Pula. Products of all elements in a loop and a framework for non-associative analogues of the Hall-Paige conjecture. *Electron. J. Combin.*, **16**:R57, 2009.
- [129] K. Pula. A generalization of plexes of latin squares. preprint.

[130] J. Quistorff. A survey on packing and covering problems in the Hamming permutation space. *Electron. J. Combin.*, **13**:A1, 2006.

- [131] I. Rivin, I. Vardi, and P. Zimmermann. The *n*-queens problem. *Amer. Math. Monthly*, **101**:629–639, 1994.
- [132] P. Rodney. Private communication at Vermont Summer Conference cited by [55]. 1994.
- [133] H. J. Ryser. A combinatorial theorem with an application to latin rectangles. *Proc. Amer. Math. Soc.*, **2**:550–552, 1951.
- [134] H. J. Ryser. Neuere probleme der kombinatorik. Vortrage über Kombinatorik Oberwolfach, 24–29 Juli:69–91, 1967.
- [135] S. B. Seidman and B.L.Foster. A graph-theoretic generalization of the clique concept. J. Math. Sociol., 6:69–91, 1978/79.
- [136] P. W. Shor. A lower bound for the length of a partial transversal in a latin square. J. Combin. Theory Ser. A, 33:1–8, 1982.
- [137] S. S. Shrikhande. A note on mutually orthogonal latin squares. Sankhyā Ser. A, 23:115-116, 1961.
- [138] B. Smetaniuk. A new construction on latin squares I. A proof of the Evans conjecture. Ars Combin., 11:155–172, 1981.
- [139] S. K. Stein. Transversals of latin squares and their generalizations. *Pacific J. Math.*, **59**:567–575, 1975.
- [140] D. S. Stones and I. M. Wanless. Compound orthomorphisms of the cyclic group. *Finite Fields Appl.*, **16**:277–289, 2010.
- [141] R. Stong. Orthogonal matchings. Discrete Math., 256:513–518, 2002.
- [142] G. H. J. van Rees. Subsquares and transversals in latin squares. Ars Combin., **29**:193–204, 1990.
- [143] I. Vardi. Computational Recreations in Mathematica. Addison-Wesley, Redwood City, 1991.
- [144] M. Vaughan-Lee and I. M. Wanless. Latin squares and the Hall-Paige conjecture. *Bull. London Math. Soc.*, **35**:191–195, 2003.
- [145] F. Dalla Volta and N. Gavioli. Complete mappings in some linear and projective groups. Arch. Math. (Basel), **61**:111–118, 1993.
- [146] F. Dalla Volta and N. Gavioli. On the admissibility of some linear and projective groups in odd characteristic. *Geom. Dedicata*, **66**:245–254, 1997.
- [147] F. Dalla Volta and N. Gavioli. Minimal counterexamples to a conjecture of Hall and Paige. Arch. Math. (Basel), 77:209–214, 2001.
- [148] W. D. Wallis. *One-Factorizations*. Kluwer Acadamedic Publishers, Dordrecht, 1997.
- [149] I. M. Wanless. Latin trades. http://users.monash.edu.au/iwanless/data/trades/.
- [150] I. M. Wanless. A generalisation of transversals for latin squares. *Electron. J. Combin.*, $\mathbf{9}$:R12, 2002.
- [151] I. M. Wanless. Cycle switches in latin squares. *Graphs Combin.*, **20**:545–570, 2004.
- [152] I. M. Wanless. Diagonally cyclic latin squares. European J. Combin., 25:393–413, 2004.

[153] I. M. Wanless. A computer enumeration of small latin trades. *Australas. J. Combin.*, **39**:247–258, 2007.

- [154] I. M. Wanless. Transversals in latin squares. Quasigroups Related Systems, 15:169–190, 2007.
- [155] I. M. Wanless and B. S. Webb. The existence of latin squares without orthogonal mates. *Des. Codes Cryptogr.*, **40**:131–135, 2006.
- [156] S. Wilcox. Reduction of the Hall-Paige conjecture to sporadic simple groups. J. Algebra, **321**:1407–1428, 2009.
- [157] D. E. Woolbright. An $n \times n$ latin square has a transversal with at least $n \sqrt{n}$ distinct elements. J. Combin. Theory Ser. A, 24:235–237, 1978.
- [158] D. E. Woolbright and H. Fu. On the existence of rainbows in 1-factorizations of K_{2n} . J. Combin. Des., **6**:1–20, 1998.
- [159] M. Zaker. Maximum transversal in partial latin squares and rainbow matchings. *Discrete Appl. Math.*, **155**:558–565, 2007.